UNIVERSIDAD DE GRANADA



Departamento de Ciencias de la Computación e Inteligencia Artificial

Programa de Doctorado en Tecnologías de la Información y la Comunicación

Supervised Multi-person Multi-criteria Decision-Making Methodologies under Risk and Uncertainty: Engineering Applications

Tesis Doctoral

Arian Hafezalkotob

Granada, febrero de 2023

Editor: Universidad de Granada. Tesis Doctorales Autor: Arian Hafezalkotob ISBN: 978-84-1117-898-3 URI: <u>https://hdl.handle.net/10481/82559</u>

UNIVERSIDAD DE GRANADA



Supervised Multi-person Multi-criteria Decision-Making Methodologies under Risk and Uncertainty: Engineering Applications

MEMORIA PRESENTADA POR

Arian Hafezalkotob

Para optar al grado de doctor por la Universidad de Granada dentro del

Programa de Doctorado Tecnologías de la Información y la Comunicación

febrero de 2023

DIRECTOR

Francisco Herrera Triguero

Departamento de Ciencias de la Computación e Inteligencia Artificial

This doctoral thesis is supported by the project PP2019.PRI..I.05 funded by the Plan Propio de Investigación y Transferencia of the University of Granada (Research Intensification Program) – started on 05/10/2020 and ends on 05/10/2023.

Acknowledgements

My PhD study at the University of Granada (UGR) from late-2017, supported by the Research Intensification Program of the UGR, has been a great research experience for me which paves the way for a future academic career. It affords me an opportunity to go in depth in my favorite research line, i.e., Decision-Making and Soft Computing, previously started in 2014.

For this privilege, I would like to express my deep appreciation to my supervisor, Prof. Francisco Herrera Triguero, for such a great opportunity, insightful supervision, and continuous support throughout my study. Apart from what I learned academically from my supervisor, I learned ethical issues, research methods, and philosophy of science from him. These issues will be sure necessary for a solid research career in future. Besides, I am also go grateful to all my co-authors for their great collaborations.

Table of Contents

Page

Summary	11
Resumen	12
Introduction	14
Context	14
Contexto	14
Motivation and hypothesis	15
Objectives	16
Methodologies and models	16
Structure of the thesis	17
I. Preliminaries on MpMcDM, uncertainty, and risk	
1. MpMcDM methods	
1.1.MULTIMOORA method for MpMcDM	
1.2.BWM for MpMcDM	
2. Uncertainty theories	19
2.1.Interval numbers	19
2.2. Trapezoidal fuzzy numbers	
3. Risk theory based on prospect theory	23
3.1.Principles of prospect theory	
3.2. Proposed Fuzzy-Rationality-Based Prospect (FRFP) theory	24
II. Overview on MULTIMOORA for risky uncertain MpMcDM and its	
applications	
1. Introduction	27
2. Theory of MULTIMOORA method	
2.1.Fundamentals of MULTIMOORA	
2.2.MpMcDM with MULTIMOORA method	33
2.3. Developments of MULTIMOORA method under risk and uncertainty	
3. Application of MULTIMOORA method	
III. I_S-MpMcDM methodology: Industrial application	
1. Introduction	
2. Theory of I S-MpMcDM methodology	
2.1.BWM-based model for weighting process	39
2.2.MULTIMOORA-based model for ranking process	42
3. Application of I_S-MpMcDM methodology in industrial sector	44

IV. RF_S-MpMcDM_FRFP methodology: Biomedical application	49
1. Introduction	50
2. Theory of RF_S-MpMcDM_FRFP methodology	51
2.1.BWM-based model for weighting process	53
2.2.MULTIMOORA-based model for ranking process	
3. Application of RF_S-MpMicDM_FRFP methodology in biomedical sector	
V. DI_S-MpMcDM_SoESs methodology: Energy application	65
1. Introduction	66
2. Theory of DI_S-MpMcDM_SoES methodology	67
2.1.BWM-based models for weighting process	68
2.2.MULTIMOORA-based model for ranking process	
3. Application of DI_S-MpMcDM_SoES methodology in energy sector	74
VI. Conclusions, publications, and future works	78
1. Conclusions	78
2. Publications	79
3. Future works	79
References	81

Summary

In a realistic decision-making process, there are often multiple subjects in a problem. Each subject can include various criteria. In traditional decision-making, multiple experts usually tackle the problems; however, they are supposed to have all-inclusive competence in each subject. Moreover, human decisions are subjected to risk of cognitive imperfections. In other problems, rationality of decision-makers is vague in real life. Besides, the data of decision-making problems are associated with uncertainty. Therefore, for multidisciplinary decision-making problems, the most important challenges are:

- Segmentation is needed for experts as they may not have competence in all subjects of problem.
- Risk of imperfect decisions should be considered as rationality of experts is under question.
- Uncertainty of decision-making problems should be tackled with robust approaches to avoid loss of uncertain information.

We attempt to efficiently tackle the above difficulties, as follows:

- Multi-person structures supervised by a director are considered for our methodologies. Subject-oriented expert segments structures are employed when multiple subjects exist in decision-making problem.
- We use the concept of fuzzy rationality as a cognitive aid to represent the degree of rationality of expert decisions in risky situations.
- New uncertain measures are introduced to reach complete uncertain computation and preserve all data.

Therefore, the main objectives of this doctoral thesis are studying: (1) supervised multi-person structures with subject-oriented expert segments, (2) risk in uncertain environment, and (3) uncertainty to preserve all of data in decision-making problems with multiple criteria and alternatives.

In this regard, three methodologies are proposed, as follows:

- An interval supervised multi-person multi-criteria decision-making methodology is presented. This methodology is without subject-oriented expert segments and has a non-risky decision-making process. The computation is based on interval distances of interval numbers and interval preference matrix.
- A risky fuzzy supervised multi-person multi-criteria decision-making methodology is developed supported on fuzzy-rationality-based fuzzy prospect theory. This methodology is without subject-oriented expert segments and formulated based on proposed fuzzy-rationality-based fuzzy prospect theory. The computation is based on fuzzy distances of trapezoidal fuzzy numbers and fuzzy distance matrix for extremum and ranking.

• A dynamic interval supervised multi-person multi-criteria decision-making methodology with subjectoriented expert segments is introduced. This methodology includes the proposed Subject-oriented Expert Segments and has a non-risky decision-making process. The parametric representation is employed for interval preference in interval optimization model.

The aforementioned methodologies are derived by introducing the following decision-making models:

- BWM-based models for weighting process.
- MULTIMOORA-based models for ranking process.

We use the aforementioned methodologies and models to tackle engineering problems in the area of industrial, biomedical, and energy sectors.

Resumen

En un proceso de toma de decisiones realista, a menudo hay múltiples temas en un problema. Cada tema puede involucrar diferentes criterios. En la toma de decisiones tradicional, se dispone de múltiples expertos que suelen abordar los problemas; sin embargo, se parte del supuesto de que tienen competencias omnicomprensivas en cada tema. Además, las decisiones humanas están sujetas al riesgo de imperfecciones cognitivas. En otras palabras, la racionalidad de los decisores es imprecisa en la vida real. Asimismo, los datos de los problemas de toma de decisiones están asociados a la incertidumbre. Por lo tanto, para los problemas de toma de decisiones multidisciplinares, los retos importantes más destacados son:

- Se requiere de una segmentación de los expertos, ya que pueden no tener competencia en todos los temas del problema.
- Hay que tener en cuenta el riesgo de decisiones imperfectas, ya que se cuestiona la racionalidad de los expertos.
- La incertidumbre de los problemas de toma de decisiones debe abordarse con enfoques sólidos para evitar la pérdida de información incierta.

Intentamos abordar eficazmente las dificultades anteriores de la siguiente manera:

 Para nuestras metodologías se consideran estructuras multipersonales supervisadas por un director. Se emplean estructuras de segmentos expertos orientadas a sujetos cuando existen múltiples sujetos en el problema de toma de decisiones.

- Utilizamos el concepto de racionalidad difusa como ayuda cognitiva para representar el grado de racionalidad de las decisiones de los expertos en situaciones de riesgo.
- Se introducen nuevas medidas de incertidumbre para alcanzar un cálculo incierto completo y preservar todos los datos.

Por lo tanto, los principales objetivos de esta tesis doctoral son estudiar: (1) las estructuras supervisadas multipersona con segmentos de expertos orientados por temas, (2) el riesgo en un entorno incierto, y (3) la incertidumbre para preservar todos los datos en problemas de toma de decisiones con múltiples criterios y alternativas.

A este respecto, se proponen las tres metodologías siguientes:

- Se presenta una metodología de toma de decisiones multicriterio multipersona supervisada por intervalos. Esta metodología carece de segmentos de expertos orientados por temas y dispone de un proceso de toma de decisiones no arriesgado. El cálculo se basa en distancias de intervalo de números de intervalo y matriz de preferencias de intervalo.
- Se desarrolla una metodología de toma de decisiones multicriterio multipersona supervisada de riesgo difuso basada en la Teoría de la Perspectiva Difusa basada en la Racionalidad Difusa. Esta metodología carece de segmentos de expertos orientados por temas y se formula a partir de la Teoría difusa de las perspectivas basada en la racionalidad difusa propuesta. El cálculo se basa en distancias difusas de números trapezoidales difusos y matrices de distancias difusas para el extremo y la clasificación.
- Se introduce una metodología dinámica de toma de decisiones multicriterio supervisada por intervalos con segmentos expertos orientados a temas. Esta metodología incluye los segmentos expertos orientados a temas propuestos y tiene un proceso de toma de decisiones no arriesgado. La representación paramétrica se emplea para la preferencia de intervalo en el modelo de optimización de intervalo.

Las metodologías mencionadas se derivan introduciendo los siguientes modelos de toma de decisiones:

- Modelos basados en BWM para el proceso de ponderación.
- Modelos basados en MULTIMOORA para el proceso de clasificación.

Utilizamos las metodologías y modelos mencionados para abordar problemas de ingeniería en el ámbito de los sectores industrial, biomédico y energético.

Introduction

In this section, we present the context of Supervised Multi-person Multi-criteria Decision-Making (S-MpMcDM) considering risk and various concepts of uncertainty, motivation and hypothesis, objectives, methodologies and models, and structure of the thesis.

Context

The context of Supervised Multi-person Multi-criteria Decision-Making (S-MpMcDM) under risk and uncertainty deals with the significance of supervision of multiple experts and the impact of various theories of vagueness in decision-making area. In such problems, multiple experts supervised by a director are responsible for evaluating alternatives according to multiple criteria in a vague situation subjected to risk and uncetainty. The ideas of "supervised multi-person structures", "risk", and "uncertainty" in S-MpMcDM and its engineering applications are clarified as follows.

In real-life, many socio-economic and industrial companies deal with multidisciplinary S-MpMcDM problems. In this regard, multiple groups of experts with special competence in each subject are often required [1]. Such decision-making process can be reinforced by the supervision of a director. The director can have multiple roles to evaluate competence of experts and also participates in assessment of criteria and alternatives of the problem [2].

In the models of S-MpMcDM, the term of risk can be utilized and interpreted differently. In some cases, risk may be the indicator of the degree of optimism about the information available in the problem. Additionally, risk can be interpreted as the intensity and probability of unfavorable effects in other cases [3]. The uncertain conditions can incur risk to the problem. Risk can also be interpreted as vagueness of rationality of decision-makers [4].

In many practical S-MpMcDM problems, the available data are associated with some degrees of vagueness [5], [6]. In this regard, different kinds of uncertain sets are applied to tackle the vagueness of data, including interval numbers, fuzzy sets, and their extensions [7].

In industrial, biomedical, and energy applications of S-MpMcDM, supervised multi-person structure can result in robust decision-making process. Moreover, in many practical engineering problems, the input data are reported as uncertain values which may also entail some risks [8].

Contexto

El contexto de la Toma de Decisiones Multicriterio Supervisada por Múltiples Personas (S-MpMcDM) bajo riesgo e incertidumbre aborda la importancia de la supervisión de múltiples expertos y el impacto de diversas teorías de vaguedad en el ámbito de la toma de decisiones. En este tipo de problemas, múltiples expertos supervisados por un director se encargan de evaluar alternativas según múltiples criterios en una situación de vaguedad sometida a riesgo e incertidumbre. Las ideas de "estructuras multipersonales supervisadas", "riesgo" e "incertidumbre" en el proceso S-MpMcDM se aclaran como sigue.

En la vida real, muchas empresas socioeconómicas e industriales se enfrentan a problemas multidisciplinares de S-MpMcDM. En este sentido, a menudo se requieren múltiples grupos de expertos con competencias especiales en cada materia [1]. Este proceso de toma de decisiones puede verse reforzado por la supervisión de un director. El director puede desempeñar múltiples funciones para evaluar la competencia de los expertos y también participa en la evaluación de criterios y alternativas del problema [2].

En los modelos para abordar el S-MpMcDM, el término de riesgo puede utilizarse e interpretarse de forma diferente. En algunos casos, el riesgo puede ser el indicador del grado de optimismo sobre la información disponible en el problema. Además, en otros casos, el riesgo puede interpretarse como la intensidad y probabilidad de efectos desfavorables [3]. Las condiciones de incertidumbre pueden suponer un riesgo para el problema. El riesgo también puede interpretarse como la vaguedad de la racionalidad de los decisores [4].

En muchos problemas prácticos de S-MpMcDM, los datos disponibles están asociados con algunos grados de vaguedad [5], [6]. En este sentido, se aplican diferentes tipos de conjuntos inciertos para abordar la vaguedad de los datos, incluidos los números de intervalo, los conjuntos difusos y sus extensiones [7].

En las aplicaciones industriales, biomédicas y energéticas de S-MpMcDM, la estructura multipersonal supervisada puede dar lugar a un proceso de toma de decisiones robusto. Además, en muchos problemas prácticos de ingeniería, los datos de entrada se presentan como valores inciertos que también pueden conllevar algunos riesgos [8].

Motivation and hypothesis

Although many researches have been conducted on S-MpMcDM under risk and uncertainty, serious research gaps still exist in this regard, as follows:

- Supervision analysis in S-MpMcDM: In traditional S-MpMcDM, decision-making with multi-person structure is analyzed in very limited studies [2]. Director can also participate in downstream decisions of the problem. If there exist multiple subjects in decision-making problem, experts can be segmented based on their competence in the subjects. Director may additionally evaluate the relative significance of the problem subjects.
- **Risk analysis in S-MpMcDM:** The concept of risk factors are usually presented as independent criteria in the context of MpMcDM methods [9]; however, there are a few studies that include risk factors directly into the formulation [10]. As rationality of human decisions is imperfect in real-life, considering the risk of the degrees of rationality is an interesting topic in the S-MpMcDM context. In some real cases, decisions are subject to the risks posed from the lack of or limited knowledge of problem conditions.
- Uncertainty analysis in S-MpMcDM: There are many studies in the field of uncertain decision-making problems
 with multiple person structure; however, some research gaps still exist in this regard. For considering uncertainty,
 the thesis focuses on interval numbers and fuzzy sets for suggesting new developments. Interval S-MpMcDM
 approaches have got rather lower attention comparing the fuzzy methods despite their privileges especially in
 practical industrial application [11]. Research gaps exist regarding complete uncertain computation based on
 interval and fuzzy data.

This study attempts to plug the aforementioned research gaps, as follows:

Supervision analysis in S-MpMcDM: We employ two supervised multi-person structures for S-MpMcDM
process. The first structure consists of one director and multiple experts. For the second structure, multiple experts
are segmented based on problem subjects. That is, the second structure comprises one director and multiple expert

segments. For the second structure, we coin the expression "Subject-oriented Expert Segments (SoESs)" and introduce it into the S-MpMcDM context. The SoESs structure is a multi-person decision-making framework in which experts are segmented based on their competence in problem subjects.

- **Risk analysis in S-MpMcDM:** We introduce the "Fuzzy-Rationality-based Fuzzy Prospect (FRFP)" theory to evaluate the risk of experts decisions. We use the concept of fuzzy rationality as an cognitive aid to represent the degree of rationality of expert decisions in risky S-MpMcDM. That is, fuzzy rationality is used to indicate risk of expert decision in realistic problems.
- Uncertainty analysis in S-MpMcDM: We tackle uncertainty for S-MpMcDM based on theories of interval and trapezoidal fuzzy numbers. In the theory of interval numbers, we propose interval distance and interval preference matrix to make pairwise comparison and obtain extremum and ranking, respectively. In the theory of trapezoidal fuzzy numbers, we develop fuzzy distance to make pairwise comparison. To compute extremum and ranking of trapezoidal fuzzy numbers, we first calculate α-cuts to reach the corresponding interval numbers and then employ the proposed interval preference matrix.

Objectives

The main objectives of this thesis is to develop Supervised Multi-person Multi-criteria Decision-Making (S-MpMcDM) methodologies under risk and uncertainty. Besides, the related studies in the literature are reviewed comprehensively and the proposed methodologies are utilized in practical engineering problems.

We present the objectives of this thesis, as follows:

- *Objective 1:* Overview of the MpMcDM methodologies with specific interest in uncertainty, risk, and their applications.
- *Objective 2:* Study of "supervised multi-person structures" in S-MpMcDM process with subject-oriented expert segments.
- Objective 3: Study of "risk" in S-MpMcDM process based on the prospect theory in uncertain environment.
- *Objective 4:* Study of "uncertainty" in S-MpMcDM process based on the theories of interval and fuzzy numbers.
- *Objective 5:* Analysis of engineering applications of S-MpMcDM problems in industrial, biomedical, and energy sectors.

Methodologies and models

As it is aforementioned in Objectives Section, the focus of theory of this thesis is on the study of "supervised multi-person structures", "risk", and "uncertainty" in S-MpMcDM process. In this regard, three methodologies are proposed, as follows:

- *Methodology 1*: We present the "*Interval Supervised Multi-person Multi-criteria Decision-Making (I_S-MpMcDM)*" methodology. This methodology is without subject-oriented expert segments and has a non-risky decision-making process. The computation is based on interval distances of interval numbers and interval preference matrix.
- *Methodology 2*: We develop the "*Risky Fuzzy Supervised Multi-person Multi-criteria Decision-Making methodology supported on Fuzzy-Rationality-based Fuzzy Prospect Theory (RF_S-MpMcDM_FRFP)*". This methodology is without subject-oriented expert segments and formulated based on "Proposed Fuzzy-Rationality-

based Fuzzy Prospect Theory (FRFP)". The computation is based on fuzzy distances of trapezoidal fuzzy numbers and and fuzzy distance matrix for extremum and ranking.

• *Methodology 3*: We introduce the "Dynamic Interval Supervised Multi-person Multi-criteria Decision-Making methodology with Subject-oriented Expert Segments (DI_S-MpMcDM_SoESs)". This methodology includes the proposed "Subject-oriented Expert Segments (SoESs)" and has a non-risky decision-making process. The parametric representation is employed for interval preference in interval optimization model.

The abovementioned methodologies will be explained and clarified with details in Methodologies Section. The proposed methodologies are derived by introducing the following decision-making models:

- 1) BWM-based models for weighting process: We derive models based on Best-Worst Method (BWM) to compute weights needed for the proposed methodologies. For the I_S-MpMcDM methodology, two interval BWM-based model is generated to weight experts and criteria. For the RF_S-MpMcDM_FRFP methodology, two risky fuzzy BWM-based model supported on the FRFP theory is generated to weight experts and criteria. For the DI_S-MpMcDM_SoESs methodology, three interval BWM-based models supported on the SoESs structure are generated to calculate dynamic weights of subjects, experts, and alternatives.
- 2) MULTIMOORA-based models for ranking process: We develop models based on Multi-Objective Optimization by Ratio Analysis plus the full MULTIPlicative form (MULTIMOORA) to calculate ranking of alternatives in the eventual step of the proposed methodologies. The obtained weights in the BWM-models are used in derivation of the MULTIMOORA-based models. For the I_S-MpMcDM methodology, the final rankings of alternatives are obtained utilizing the interval MULTIMOORA-Borda model. For the RF_S-MpMcDM_FRFP methodology, subordinate rankings are generated based on the fuzzy MULTIMOORA model and the final ranking is obtained employing the fuzzy distance matrix. For the DI_S-MpMcDM_SoESs methodology, the dynamic assessment values of alternatives are aggregated based on the interval MULTIMOORA-Borda model.

Structure of the thesis

The remainder of this thesis is arranged into six chapters as follows:

- Chapter I gives preliminaries on MpMcDM, uncertainty, and risk.
- Chapter II is allocated to the overview on MULTIMOORA for risky uncertain MpMcDM and its applications.
- Chapter III presents the I_S-MpMcDM methodology and its industrial application.
- Chapter IV discusses the RF_S-MpMcDM_FRFP methodology and its biomedical application.
- Chapter V presents the DI_S-MpMcDM_SoESs methodology and its energy application.
- Chapter VI gives the concluding remarks, research papers published regarding the overview and methodologies, as well as the directions for future works.

Chapter I

Preliminaries on MpMcDM, uncertainty, and risk

In this chapter, we present the preliminaries required for derivation of methodologies. Sections 1, 3, and 2 give fundamentals on MpMcDM, uncertainty, and risk, respectively.

1. MpMcDM methods

We discuss two important decision-making methods in the context of MpMcDM methods, MULTIMOORA and BWM in Sections 1.1 and 1.2, respectively.

1.1. MULTIMOORA method for MpMcDM

MULTIMOORA is a robust decision-making approach employing three subordinate methods and a rank aggregation function. The subordinate techniques are Ratio System, Reference Point Approach, and Full Multiplicative Form. Ratio System and Full Multiplicative Form effectively solve problems with independent and dependent criteria, respectively. Reference Point Approach leads to a conservative ranking of alternatives [12].

Earlier studies have exploited the approach of MULTIMOORA in many MpMcDM problems. Wang et al. [13] put forward a trust model for preference of experts based on the multi-person MULTIMOORA. Zolfaghari and Mousavi [14] presented a risky multi-person MULTIMOORA extension employing failure mode and effect analysis. Deng et al. [15] introduced an optimization-based consensus model supported on the multi-person MULTIMOORA.

A review of the researches on MULTIMOORA, including its developments and applications, has been conducted by Hafezalkotob et al. [16].

1.2. BWM for MpMcDM

BWM is an important decision-making technique for obtaining relative preferences of decision items as: (i) it needs less number of comparisons in contrast to Analytic Hierarchy Process (AHP); (ii) BWM has higher consistency than AHP; (iii) consistency of BWM also represents the confidence level; and (iv) BWM is straightforward for decision-makers from the viewpoint of mathematical computation [17].

Many researchers developed MpMcDM approaches based on the BWM. Hafezalkotob *et al.* [18] suggested a multiperson BWM model in which a director supervises an expert panel. Yazdi *et al.* [19] developed a risky multi-person BWM approach with a democratic-autocratic decision-making style. Omrani *et al.* [20] employed the concept of data envelopment analysis for a multi-person BWM approach.

A review of the studies on BWM comprising its extensions and applications has been presented by Mi et al. [17].

2. Uncertainty theories

We discuss the important theories and our novelties related to interval numbers nand trapezoidal fuzzy numbers in Sections 2.1 and 2.2, respectively.

2.1. Interval numbers

Mathematical theories of interval numbers are given in this section including basic definitions, interval distance, and preference matrix used for extremum and rankings.

• Basic definitions of interval numbers

The midpoint of \overline{A} is defined as [21]:

$$pm\left(\overline{A}\right) = A^{M} = \frac{A^{L} + A^{U}}{2} \tag{I.1}$$

For two non-negative real interval numbers $\overline{A} = [A^L, A^U]$ and $\overline{B} = [B^L, B^U]$ and a non-negative real number *r*, the

following arithmetic operations were defined [22]:

$$\overline{A} + \overline{B} = \begin{bmatrix} A^{L} + B^{L}, A^{U} + B^{U} \end{bmatrix}, \overline{A} - \overline{B} = \begin{bmatrix} A^{L} - B^{U}, A^{U} - B^{L} \end{bmatrix}$$

$$\overline{AB} = \begin{bmatrix} A^{L}B^{L}, A^{U}B^{U} \end{bmatrix}$$

$$\overline{A}/\overline{B} = \begin{bmatrix} A^{L}/B^{U}, A^{U}/B^{L} \end{bmatrix}, \text{ with } B^{L} \text{ and } B^{U} \neq 0$$

$$r\overline{A} = \begin{bmatrix} rA^{L}, rA^{U} \end{bmatrix}, \ \overline{A}^{r} = \begin{bmatrix} \left(A^{L}\right)^{r}, \left(A^{U}\right)^{r} \end{bmatrix}$$
(I.2)

• Interval distance between two interval numbers

The interval distance between interval numbers \overline{A} and \overline{B} was defined by Trindade et al. [21]:

$$\overline{d}(\overline{A},\overline{B}) = \begin{cases} \left[\min\left\{ \left| A^{L} - B^{U} \right|, \left| A^{U} - B^{L} \right| \right\} \right] \\ \max\left\{ \left| A^{L} - B^{U} \right|, \left| A^{U} - B^{L} \right| \right\} \right], & \text{if } \overline{A} \cap \overline{B} = \emptyset \\ \left[0, \max\left\{ \left| A^{L} - B^{U} \right|, \left| A^{U} - B^{L} \right| \right\} \right], & \text{if } \overline{A} \cap \overline{B} \neq \emptyset \end{cases}$$
(I.3)

Eq. (I.3) has some shortcomings and thus Hafezalkotob and Hafezalkotob [23] improved it to the following form:

$$\overline{d}^{*}(\overline{A},\overline{B}) = \begin{cases} \left[\min\left\{ \left| A^{L} - B^{U} \right|, \left| A^{U} - B^{L} \right| \right\}, \left| A^{M} - B^{M} \right| \right], \text{ if } \overline{A} \cap \overline{B} = \emptyset \\ \left[\left[0, \left| A^{M} - B^{M} \right| \right], & \text{ if } \overline{A} \cap \overline{B} \neq \emptyset \end{cases}$$
(I.4)

To show the differences between \overline{d} and \overline{d}^* , we present some numerical examples as listed in Table I.1. The general point which can be conceived from Table I.1 is that \overline{d}^* leads to smaller interval distance comparing with \overline{d} . Considering the second to fifth rows of Table I.1, it is obvious that \overline{d} may not be sensitive to the degrees of intersection and inclusion of two interval numbers. However, \overline{d}^* changes with the degrees of intersection and inclusion. The last row of Table I.1

shows that the interval distance \overline{d} between two equal interval numbers is a given interval number; however, it may not be rational. In contrast to \overline{d} , the interval distance \overline{d}^* between two equal interval numbers is exactly zero.

Ā	\overline{B}	\overline{A} vs. \overline{B}	Relation	$\overline{d}(\overline{A},\overline{B})$	$\overline{d}^*(\overline{A},\overline{B})$
[1, 2]	[3, 7]	$\overline{A} < \overline{B} \Big _{\overline{A} \cap \overline{B} = \emptyset}$	Non-intersection	[1, 6]	[1, 3.5]
[1, 4]	[3, 7]	$\overline{A} < \overline{\mathbf{B}} \Big _{\overline{A} \frown \overline{B} \neq \emptyset}$	Intersection	[0, 6]	[0, 2.5]
[1, 6]	[3, 7]	$\overline{A} < \overline{\mathbf{B}} \Big _{\overline{A} \frown \overline{B} \neq \emptyset}$	Intersection	[0, 6]	[0, 1.5]
[3.5, 5.5]	[3, 7]	$\overline{A} \subset \overline{B}$	Inclusion	[0, 3.5]	[0, 0.5]
[3.5, 6]	[3, 7]	$\overline{A} \subset \overline{B}$	Inclusion	[0, 3.5]	[0, 0.25]
[3, 7]	[3, 7]	$\overline{A} = \overline{B}$	Equality	[0, 4]	0

Table I.1. The differences between the interval distance \overline{d} and \overline{d}^* .

• Preference matrix for finding the extremum and rankings of interval numbers

The preference degree of $\overline{A} = [A^L, A^U]$ over $\overline{B} = [B^L, B^U]$ is defined as [24]:

$$P(\bar{A} > \bar{B}) = \frac{\max\{0, A^{U} - B^{L}\} - \max\{0, A^{L} - B^{U}\}}{A^{U} - A^{L} + B^{U} - B^{L}}$$
(I.5)

The following properties hold [25]:

- $P(\overline{B} > \overline{A}) = 1 P(\overline{A} > \overline{B})$. If $\overline{A} = \overline{B}$ then $P(\overline{A} > \overline{B}) = P(\overline{B} > \overline{A}) = 0.5$.
- If $P(\overline{A} > \overline{B}) > P(\overline{B} > \overline{A})$, then \overline{A} is said to be superior to \overline{B} to the degree of $P(\overline{A} > \overline{B})$, represented as $\overline{A} \xrightarrow{P(\overline{A} > \overline{B})} \overline{B}$; If $P(\overline{A} > \overline{B}) = P(\overline{B} > \overline{A}) = 0.5$, then \overline{A} is said to be indifferent to \overline{B} , indicated as $\overline{A} \sim \overline{B}$; If $P(\overline{B} > \overline{A}) \xrightarrow{P(\overline{B} > \overline{A})} = 0.5$, then \overline{A} is said to be indifferent to \overline{B} , indicated as $\overline{A} \sim \overline{B}$; If
 - $P(\overline{B} > \overline{A}) > P(\overline{A} > \overline{B})$, then \overline{A} is said to be inferior to \overline{B} to the degree of $P(\overline{B} > \overline{A})$, displayed as $\overline{A} \stackrel{P(\overline{B} > \overline{A})}{\prec} \overline{B}$
- If $\overline{A} \leq \overline{B}$ and $\overline{A} \cap \overline{B} = \emptyset$, then $P(\overline{A} \geq \overline{B}) = 0$. If $\overline{A} \leq \overline{B}$ and $\overline{A} \cap \overline{B} \neq \emptyset$, then $0 \leq P(\overline{A} \geq \overline{B}) \leq 0.5$.
- If $\overline{A} \ge \overline{B}$ and $\overline{A} \cap \overline{B} = \emptyset$, then $P(\overline{A} \ge \overline{B}) = 1$. If $\overline{A} \ge \overline{B}$ and $AP(\phi)$, then $0.5 \le P(\overline{A} \ge \overline{B}) \le 1$.

To rank a set of interval numbers $\{\overline{A}, \overline{B}, \dots, \overline{N}\}$, a preference degree matrix is defined as Table I.2.

Table I.2. Preference degree matrix.

	\overline{A}	\overline{B}	 \overline{N}
\overline{A}	$P(\overline{A} > \overline{A})$	$P(\overline{A} > \overline{B})$	 $P(\overline{A} > \overline{N})$
\overline{B}	$P(\overline{B} > \overline{A})$	$P(\overline{B} > \overline{B})$	 $P(\overline{B} > \overline{N})$
:	:	:	 :
\overline{N}	$P(\overline{N} > \overline{A})$	$P(\overline{N} > \overline{B})$	 $P(\overline{N} > \overline{N})$

In the next step, the relative preference $P_{\varphi\theta}$ is generated as:

$$P_{\varphi\theta} = \begin{cases} 1, & \text{if } P(\varphi > \theta) > 0.5, \\ 0, & \text{if } P(\varphi > \theta) \le 0.5, \end{cases} \qquad \varphi, \ \theta = \overline{A}, \overline{B}, \ \dots, \overline{N}$$
(I.6)

Table I.3 shows the preference matrix for calculating extremum and ranking of interval numbers.

Table I.3. Preference matrix.

Relative p	references		Aggregated preferences	
	\overline{A}	\overline{B}	 \overline{N}	
\overline{A}	$P_{\overline{A}\overline{A}}$	$P_{\overline{A}\overline{B}}$	 $P_{\overline{A}\overline{N}}$	$AP(\overline{A})$
\overline{B}	$P_{\overline{BA}}$	$P_{\overline{B}\overline{B}}$	 $P_{\overline{B}\overline{N}}$	$AP(\overline{B})$
:	÷	÷	÷	÷
\overline{N}	$P_{\overline{NA}}$	$P_{\overline{NB}}$	 $P_{\overline{N}\overline{N}}$	$AP(\overline{N})$

The aggregated preference $AP(\phi)$ is the sum of the relative preferences $P_{\phi\theta}$ in each row of Table I.3:

$$AP(\varphi) = \sum_{\theta=\bar{A}}^{\bar{N}} P_{\varphi\theta} \tag{I.7}$$

The maximum and minimum values of the set of interval values $\{\overline{A}, \overline{B}, \dots, \overline{N}\}$ as well as the descending and ascending rankings of each interval value are obtained as:

 $\max\left\{\overline{A}, \overline{B}, \dots, \overline{N}\right\} = \underset{\varphi}{\arg\max} AP(\varphi)$ (I.8)

$$\min\left\{\overline{A}, \overline{B}, \dots, \overline{N}\right\} = \arg\min_{\varphi} AP(\varphi) \tag{I.9}$$

$$\begin{aligned} r_{d}(\varphi) &= r\left(\varphi, \left\{\bar{A}, \bar{B}, \dots, \bar{N}\right\}, \left[\text{decending order}\right]\right) = \\ r\left(AP(\varphi), \left\{AP(\bar{A}), AP(\bar{B}), \dots, AP(\bar{N})\right\}, \left[\text{decending order}\right]\right) \end{aligned}$$
(I.10)
$$\begin{aligned} r_{a}(\varphi) &= r\left(\varphi, \left\{\bar{A}, \bar{B}, \dots, \bar{N}\right\}, \left[\text{ascending order}\right]\right) = \\ r\left(AP(\varphi), \left\{AP(\bar{A}), AP(\bar{B}), \dots, AP(\bar{N})\right\}, \left[\text{ascending order}\right]\right) \end{aligned}$$
(I.11)

2.2. Trapezoidal fuzzy numbers

A trapezoidal fuzzy number is a prevalent form of representing fuzzy numbers [5]. Every fuzzy number is defined through a membership function. A trapezoidal fuzzy number $\tilde{A} = (A_1, A_2, A_3, A_4)$ is defined by a membership function as introduced in Eq. (I.12):

$$\mu_{\tilde{A}}(x) = \begin{cases} 0, & x < A_{1}, \\ \frac{x - A_{1}}{A_{2} - A_{1}}, & A_{1} \le x \le A_{2}, \\ 1, & A_{2} \le x \le A_{3}, \\ \frac{x - A_{4}}{A_{3} - A_{4}}, & A_{3} \le x \le A_{4}, \\ 0, & x > A_{4}. \end{cases}$$
(I.12)

The important mathematical concepts of trapezoidal fuzzy numbers include (i) basic mathematics, (ii) distance measure, and (iii) extremum and rankings. The following points presents the three items:

• Basic mathematics of trapezoidal fuzzy numbers

A trapezoidal fuzzy number $\tilde{A} = (A_1, A_2, A_3, A_4)$ is defuzzified based on the concept of centroid obtains as [26]:

$$\overline{A} = \frac{1}{3} \left[A_1 + A_2 + A_3 + A_4 - \frac{A_4 A_3 - A_1 A_2}{(A_4 + A_3) - (A_1 + A_2)} \right].$$
(I.13)

The arithmetic operations for two positive trapezoidal fuzzy numbers $\tilde{A} = (A_1, A_2, A_3, A_4)$ and $\tilde{B} = (B_1, B_2, B_3, B_4)$ as well as a positive real number τ are [26]:

$$\widetilde{A} \oplus \widetilde{B} = (A_1 + B_1, A_2 + B_2, A_3 + B_3, A_4 + B_4),
\widetilde{A} \odot \widetilde{B} = (A_1 - B_4, A_2 - B_3, A_3 - B_2, A_4 - B_1),
\widetilde{A} \otimes \widetilde{B} \cong (A_1 B_1, A_2 B_2, A_3 B_3, A_4 B_4),
\widetilde{A} \oslash \widetilde{B} \cong (A_1 / B_4, A_2 / B_3, A_3 / B_2, A_4 / B_1),
\widetilde{A} \odot \tau = (A_1 \tau, A_2 \tau, A_3 \tau, A_4 \tau).$$
(I.14)

• Distance measures of trapezoidal fuzzy numbers

Based on the vertex method, the traditional crisp distance between positive trapezoidal fuzzy numbers is represented as follows [26]:

$$d_{\rm CR}(\tilde{A},\tilde{B}) = \sqrt{\frac{1}{6}} \left[\left(A_1 - B_1 \right)^2 + 2\left(A_2 - B_2 \right)^2 + 2\left(A_3 - B_3 \right)^2 + \left(A_4 - B_4 \right)^2 \right].$$
(I.15)

Interval distance measure for two positive trapezoidal fuzzy numbers is defined as [27]:

$$\overline{d}_{\rm IN}(\tilde{A},\tilde{B}) = \left[d_{\rm IN}^{\rm L}, d_{\rm IN}^{\rm U}\right],\tag{I.16}$$

where d_{IN}^{L} and d_{IN}^{U} are the lower and upper bounds given in Appendix in the supplementary file. \overline{d}_{IN} is computed based on α -cut with a weighting factor for different values of α . d_{IN}^{L} is obtained by assigning more weight to the intervals with higher values of α , while d_{IN}^{U} indicates the equal weights for the intervals at various levels of α .

In the literature of fuzzy computation, different types of distance measures of trapezoidal fuzzy numbers exist, but the introduced distances are crisp or parametric (e.g., based on α -cut)[27]–[29]. In contrast, we propose a novel fuzzy distance measure of positive trapezoidal fuzzy numbers as defined in Eq. (I.17):

$$\tilde{d}(\tilde{A}, \tilde{B}) = \{ (\varphi_1, \varphi_2, \varphi_3, \varphi_4), \text{ for Non-Intersection or Partly-Intersection,} \\ (0, \varphi_1, \varphi_2, \varphi_5), \text{ for Fully-Intersection (Inclusion or Equality Modes)} \},$$

$$(I.17)$$

 $\tilde{d}(\tilde{A}, \tilde{B})$ is also a trapezoidal fuzzy number where $\varphi_1 = \min \theta$, $\varphi_2 = \min \{\theta \setminus \varphi_1\}$, $\varphi_3 = \max \theta$, $\varphi_4 = \max \{\theta \setminus \varphi_3\}$, and $\varphi_5 = \min \{\theta \setminus \{\varphi_1, \varphi_2\}\}$ (operator "\" denotes the extraction of its right side term from set ℓ). Set ℓ is defined as follows:

$$\theta = \left\{ z_0, \left\{ z_{11}, \dots, z_{qf}, \dots, z_{44} \right\} \right\}.$$
(I.18)

where z_0 and z_{ef} are defined as:

$$z_{0} = \left|\overline{A} - \overline{B}\right|, \ z_{ef} = \left|A_{e} - B_{f}\right|, \ e = 1, 2, 3, \text{and}, 4, \ f = 1, 2, 3, \text{and}, 4.$$
(I.19)

In Table I.4, we have compared the results of the proposed distance measure of trapezoidal fuzzy numbers with those of two measures previously introduced in the literature by providing some numerical examples. Four relative conditions of trapezoidal fuzzy numbers, i.e., Non-Intersection, Partly-Intersection, Fully-Intersection (Inclusion Mode), and Fully-

Intersection (Equality Mode), are considered in Table I.4. $d_{CR}(\tilde{A}, \tilde{B})$ and $\overline{d}_{IN}(\tilde{A}, \tilde{B})$ provide crisp and interval distances obtained based on Eqs. (I.15) and (I.16), respectively. In Table I.4, the two crisp and interval distance measures are followed by the proposed fuzzy distance (obtained based on Eq. (I.17)) and the last column shows the defuzzified values of the fuzzy distance. Table I.4 indicates that $\overline{d}_{IN}(\tilde{A}, \tilde{B})$ has a shortcoming in Equality Mode as it finds a value for the distance between a trapezoidal fuzzy number and itself. However, in Equality Mode, our measure, i.e., $\tilde{d}(\tilde{A}, \tilde{B})$, obtains zero which is reasonable. The defuzzified values of the fuzzy distances are consistent with the values of the crisp distances (i.e., $d_{CR}(\tilde{A}, \tilde{B})$). This issue implies the validity of our proposed fuzzy distance measure.

Fable I.4. Comparison between the provided of the pro	roposed fuzzy distance	and other distance measures	of trapezoidal fuzzy numbers.
--	------------------------	-----------------------------	-------------------------------

Ã	Ĩ	à vs. B	Relation	$d_{CR}(\widetilde{A},\widetilde{B})[26]$	$\overline{d}_{IN}(\widetilde{A},\widetilde{B})[27]$	$\widetilde{d}(\widetilde{A},\widetilde{B})$ (our proposed measure)	Defuzzified value of $\widetilde{d}(\widetilde{A},\widetilde{B})$
(1, 2, 3, 4)	(5, 6, 7, 8)	$\bar{A} < \bar{B} \Big _{\bar{A} \frown \bar{B} = \emptyset}$	Non-Intersection	4	[4.08, 4.09]	(1, 2, 6, 7)	4
(1, 2, 3, 4)	(3.5, 4.5, 5.5, 6.5)	$\overline{A} < \overline{\mathbf{B}} \Big _{\overline{A} \cap \overline{B} \neq \emptyset}$	Partly- Intersection	2.5	[2.63, 2.64]	(0.5, 0.5, 3.5, 4.5)	2.26
(1, 2, 3, 4)	(1.5, 2, 2.5, 3)	$\bar{A} \subset \bar{B}$	Fully-Intersection (Inclusion Mode)	0.54	[0.69, 0.72]	(0, 0, 1, 1)	0.5
(1, 2, 3, 4)	(1, 2, 3, 4)	$\bar{A} = \bar{B}$	Fully-Intersection (Equality Mode)	0	[0.82, 0.85]	(0, 0, 0, 0)	0

• Extremum and rankings of trapezoidal fuzzy sets

Several previous studies have presented some techniques for computing extremum and rankings of trapezoidal fuzzy numbers [27], [30]. In this paper, we use a preference matrix for pairwise comparison of trapezoidal fuzzy numbers. First, the trapezoidal fuzzy numbers are converted into interval numbers based on the α -cut ($\alpha = 0.5$); then, the maximum and minimum values of the set of resultant intervals besides the descending and ascending rankings of the set are determined based on the preference matrix. For briefness, we avoid mentioning the details of the related computation which can be found in [18].

3. Risk theory based on prospect theory

In this part, the principles of prospect theory are discussed, initially. Then, the proposed Fuzzy-Rationality-Based Prospect (FRFP) theory is described.

3.1. Principles of prospect theory

Kahneman and Tversky [31] proposed the prospect theory. Daniel Kahneman won a "Nobel Memorial Prize in Economics" for his outstanding work concerning the development of prospect theory. Risky decision-making cannot be tackled by the expected utility theory; however, the prospect theory considers the risk of decisions to provide a more realistic modeling of human behaviors. In prospect theory, some pervasive effects of risky decisions are contemplated which are inconsistent with the expected utility theory. The main pervasive effects in prospect theory are the certainty and isolation effects. The certainty effect relates to the tendency of people to underestimate the probable outcomes compared to the certain ones. That is, the certainty effect matches the tendency of people to be risk-averse toward the selections with sure gains while to be risk-seeking as for the selections associated with sure losses. To streamline the selection process from a set of alternatives, people may neglect the components which the alternatives share while pay attention to the components that differentiate them. This tendency of people is entitled the isolation effect which may result in inconsistent preferences[31].

In prospect theory, a value function denotes the gain or loss of a decision (relative to the reference point) represented as follows[4]:

$$v(x) = \begin{cases} x^{\gamma}, & x \ge 0, \\ -\delta(-x)^{\beta}, & x < 0, \end{cases}$$
(I.20)

where x represents the deviation of a decision from the reference point dependent on the subjective viewpoint of the decision-maker. x^{γ} and $-\delta(-x)^{\beta}$ are the terms for the gain or loss where γ and β ($0 \le \gamma, \beta \le 1$) denote the concave and convex degree of the terms, respectively[32]. That is, γ and β represent the sensitive degree of the decision-maker on gain or loss, respectively. δ ($\delta > 1$) indicates the loss aversion degree of the decision-maker. In prospect theory, the utility of decision-makers on losses is considered to be higher than the utility on gains[33]. Based on the study of Kahneman and Tversky[31], the values $\gamma = \beta = 0.88$ and $\beta = 2.25$ could show the real behavior of a decision-maker.

Eq. (I.20) could be reformulated based on the distance to the reference points. Eq. (I.21) shows the distance-based formulation where x is the decision as a positive value, and R^+ and R^- stand for positive and negative reference points, respectively:

$$v(x) = \begin{cases} |x - R^{-}|^{\gamma}, & x \ge R^{-}, \\ -\delta(|x - R^{+}|)^{\beta}, & x < R^{+}. \end{cases}$$
(I.21)

Dai et al.[4] introduced a fuzzy prospect theory based on crisp distance of triangular fuzzy numbers. In a similar way, fuzzy prospect theory based on crisp distance of trapezoidal fuzzy numbers can be obtained as:

$$v_{\rm CR}\left(\tilde{x}\right) = \begin{cases} \left[d_{\rm CR}\left(\tilde{x},\,\tilde{R}^{-}\right) \right]^{\gamma}, & \tilde{x} \ge \tilde{R}^{-}, \\ -\delta \left[d_{\rm CR}\left(\tilde{x},\,\tilde{R}^{+}\right) \right]^{\beta}, & \tilde{x} < \tilde{R}^{+}, \end{cases}$$
(I.22)

where d_{CR} represents crisp distance of trapezoidal fuzzy numbers defined in Eq. (I.15).

3.2. Proposed Fuzzy-Rationality-Based Prospect (FRFP) theory

In this section, we introduce the Fuzzy-Rationality-Based Prospect (FRFP) theory. The theory is based on the cognitive theory of decision-choice as illustrated in Fig. I.1.



Fig. I.1. The relation of the proposed FRFP theory and the cognitive theory of decision-choice (the flowchart in the figure extracted form Ref. [34]).

For the problems based on the FRFP theory, a fuzzy form of the value function of prospect theory is needed. Some researchers introduced such fuzzy extensions of the value function of prospect theory[4],[35]. They have extended the traditional value function of prospect theory (i.e., Eq. (I.20)) into fuzzy representation or presented fuzzy form of the distance-based value function (i.e., Eq. (I.21)) supported on the crisp distance of fuzzy numbers. However, we suggest a novel derivation by developing the distance-based value function of the prospect theory (i.e., Eq. (I.21)) employing the proposed measure of fuzzy distance (i.e., Eq. (I.17)), as:

$$\tilde{v}(\tilde{x}) = \begin{cases} \left[\tilde{d}\left(\tilde{x}, \tilde{R}^{-}\right)\right]^{\gamma}, & \tilde{x} \ge \tilde{R}^{-}, \\ -\delta \left[\tilde{d}\left(\tilde{x}, \tilde{R}^{+}\right)\right]^{\beta}, & \tilde{x} < \tilde{R}^{+}, \end{cases}$$
(I.23)

where \tilde{x} is the fuzzy decision of a decision-maker as a positive value and γ , β , and δ are defined similar to Eq. (I.20). If we consider \tilde{x} , \tilde{R}^+ , and \tilde{R}^- as trapezoidal fuzzy values, the resultant value function would be also a trapezoidal fuzzy rating (the distance measure defined in Eq. (I.17) provides trapezoidal fuzzy outputs).

Chapter II

Overview on MULTIMOORA for risky uncertain MpMcDM and its applications

MULTIMOORA is a useful multi-criteria decision-making technique. The output of the MULTIMOORA is a ranking obtained by aggregating the results of the ternary ranking methods: Ratio System, Reference Point Approach, and Full Multiplicative Form. In the literature of MULTIMOORA, there is not a comprehensive review study. In this chapter, we conduct an overview of MULTIMOORA by categorizing and analyzing main researches, theoretically and practically.

First, we go through an theoretical survey of MULTIMOORA in terms of the subordinate ranking methods, and the robustness of the method. We analyze MpMcDM with MULTIMOORA. We scrutinize the developments of MULTIMOORA based on uncertainty theories.

Practical problems of MULTIMOORA are discussed mainly sectors concerning industries and healthcare management.

1. Introduction

MpMcDM methods tackle the problem of finding the best solution from a set of candidate alternatives in respect of multiple criteria. Often, there is no alternative which dominates the others on all criteria; thus, decision-makers usually look for the satisfactory solution [36]. The decision-making pproaches can be categorized into three groups: ① Value Measurement Methods, like SAW (Simple Additive Weighting) [37] and WASPAS (Weighted Aggregated Sum Product Assessment) [38]; ② Goal or Reference Level Models, such as TOPSIS (Technique for Order Performance by Similarity to Ideal Solution) [39] and VIKOR (VIse Kriterijumska Optimizacija kompromisno Resenje, in Serbian, Multiple Criteria Optimization Compromise Solution) [40]; and ③ Outranking Techniques, like PROMETHEE (Preference Ranking Organization METHod for Enrichment of Evaluations) [41], ELECTRE (ELimination Et Choix Traduisant la REalité, in French, ELimination and Choice Expressing the Reality) [42], ORESTE (Organisation, Rangement Et SynThèse de donnéEs relarionnelles, in French, Organization, Arrangement and Synthesis of Relational Data) [43], and GLDS (Gained and Lost Dominance Score) method [44].

In 2006, Brauers and Zavadskas [45] introduced MOORA (Multi-Objective Optimization on the basis of a Ratio Analysis) combining Ratio System and Reference Point Approach. In 2010, Brauers and Zavadskas [46] improved MOORA to MULTIMOORA (Multi-Objective Optimization on the basis of a Ratio Analysis plus the full MULTIPlicative form) by adding Full Multiplicative Form and employing Dominance Theory to obtain a final integrative ranking based on the results of these triple subordinate methods. Ratio System and Full Multiplicative Form belong to the first group of decision-making approaches (i.e., Value Measurement Methods) while Reference Point Approach falls in the second group of decision-making approaches (i.e., Goal or Reference Level Models).

As Ratio system employs arithmetic weighted aggregation operator, it is useful in applications like student selection in which "independent" criteria exist in the problem. Suppose, we compare two students based on their exam marks. As the exams are independent on each other, arithmetic operator works fine for the case. That is, it is not important that in which exams the student has better performance. Thus, the overall performance in all exams (which are independent) is significant. However, Ratio system has defects in the cases where "dependent" criteria appear in a decision-making problem. Suppose, we compare two investment companies based on their portfolios in different years. The performance of an investment company in each year is "dependent" on the other years, that is, investment in a particular year influences the status of the following years. For example, if the portfolio return in one year were very poor, in reality, this issue should affect the overall performance dramatically. Geometric operator could consider the dependency of performances of each year while arithmetic operator neglects the issue. In the cases where "dependent" criteria exist in decision-making problems, Full Multiplicative Form can be helpful as it applies geometric weighted aggregation operator. Reference Point Approach which utilizes Min-Max Metric is a "conservative" method useful for the cases where the optimal choice for decision-makers is the alternative that does not have a very bad performance on none of the criteria.

To integrate the outcomes of the three subordinate parts, a variety of ranking aggregation techniques can be deployed. In this regard, the most common ranking aggregation tool in the literature of MULTIMOORA is Dominance Theory which is also the concept adopted in the original MULTIMOORA suggested by Brauers and Zavadskas [46]. Other ranking aggregation tools such as Dominance-Directed Graph, Rank Position Method, Technique of Precise Order Preference, Borda Rule, Improved Borda Rule, ORESTE Method, and Optimization Model have also been applied to generate the final ranking of the MULTIMOORA approach. Only one survey study was previously conducted on MULTIMOORA, by Baležentis and Baležentis [36] in 2014. The work is limited to a few models regarding Group Decision-Making, Fuzzy Set Theory, and practical applications. In the current overview, we discuss MULTIMOORA models not only based on Group Decision-Making, Fuzzy Set Theory and applications, but also evaluate multiple theoretical features, various uncertainty theories, and applications in different fields besides provide bibliometric analysis, and identify significant theoretical and practical challenges. In this regard, this overview can be presented as the following itemized list:

- We highlight the features of MULTIMOORA by discussing the ternary subordinate utilities and clarifying the robustness of the decision-making method. Besides, we analyze multi-person decision-making structures, and the models used for combination.
- 2) We present the developments of MULTIMOORA based on uncertainty theories including interval numbers, and fuzzy set theories as well as their combinations. The formulations of the significant uncertain extensions are also provided and all developments are evaluated statistically.
- We present the applications of MULTIMOORA in the sectors of industries and healthcare management. Also, the important applications are evaluated statistically.

This overview is organized as follows. Section 2 focuses on Theory of MULTIMOORA method. We present applications of MULTIMOORA method in Section 3.

2. Theory of MULTIMOORA method

Fundamentals of MULTIMOORA, MpMcDM with MULTIMOORA, and its uncertain developments are discussed in Sections 2.1, 2.2, and 2.3, respectively.

2.1. Fundamentals of MULTIMOORA

MULTIMOORA exploits the vector normalization technique for generating comparable ratings and three subordinate ranking methods entitled Ratio System, Reference Point Approach, and Full Multiplicative Form. Each of the three ranking methods has some privileges but suffers from shortcomings; thus, MULTIMOORA uses more than one approach. In this subsection, we make a description about these three subordinate ranking method to facilitate the understanding of the MULTIMOORA method.

The first step in an decision-making problem is constructing a decision matrix and weight vector. Thus, for MULTIMOORA, decision matrix composed of the ratings x_{ij} of *m* candidate alternatives of the problem with respect to *n* criteria is first constructed, as follows [47]:

	c_1	•••	c_{j}	•••	C_n
	$\int x_{11}$	•••	x_{1j}		$x_{1n} \mid A_1$
	1		÷		: :
$\mathbf{X} =$	x_{i1}	•••	x_{ij}		$x_{in} \mid A_i$
	÷		:		: :
	x_{m1}		x_{mj}		$x_{mn} \int A_m$
W =	w_1		W_{j}		W_n

Because the ratings of alternatives on the multiple criteria of the problem may have different dimensions, the ratings should be normalized before utilization in a decision-making model. Different normalization schemes have been employed in MpMcDM methods [45], [48]. Liao et al. [49] made a comparison over different normalization schemes. Brauers et al.

$$x_{ij}^* = x_{ij} / \sqrt{\sum_{i=1}^m (x_{ij})^2}$$
 (II.2)

Ratio System, as a fully compensatory model, is useful when "independent" criteria exist in the problem. For cases with the existence of "dependent" criteria, Full Multiplicative Form, as an incompletely-compensatory model, is a beneficial tool. Reference Point Approach, as a non-compensatory model, is a "conservative" method comparing Ratio System and Full Multiplicative Form. Ratio System and Full Multiplicative Form both provide the opportunity to compensate the poor performance of an alternative on one criterion by the performances on other criteria (the degree of compensation related to the two techniques is not equal); however, Reference Point Approach does not allow such an opportunity. As "dependent" and "independent" criteria may exist simultaneously in the problem and for the sake of having a "conservative" result, MULTIMOORA integrates the triple methods to exploit the advantages of each of them and reach a final outcome that is more robust than the individual results [51]. We discuss the derivation of the triple subordinate ranking methods besides the connection of the methods with other decision-making approaches, as follows:

Ratio System

Ratio System which uses the arithmetic weighted aggregation operator is a fully compensatory model. It means that small normalized values of an alternative could be completely compensated by the same degree of large values. In other words, an alternative with poor performance in respect to some criteria and fine performance in respect to the remained criteria can be substituted by an alternative with moderate performance in respect to all criteria [52]. To compute the utility of Ratio System, the weighted normalized ratings are added for beneficial criteria and deducted for non-beneficial criteria as follows [53]:

$$y_i = \sum_{j=1}^{g} w_j x_{ij}^* - \sum_{j=g+1}^{n} w_j x_{ij}^*$$
(II.3)

where g is the number of beneficial criteria and (n-g) is the number of non-beneficial criteria. The best alternative based on Ratio System has the maximum utility y_i and the ranking of this method is obtained in descending order as:

$$\mathbf{R}_{\rm RS} = \left\{ A_i | \max_i y_i \succ \cdots \succ A_i | \min_i y_i \right\}.$$
(II.4)

Ratio System, is inspired by SAW. In SAW, same as Ratio System, the utility is obtained by aggregation of the weighted normalized alternatives ratings; however, there is only one term for sum (i.e., no term exists for subtraction) because SAW's normalization is based on a linear ratio. For beneficial criteria, each alternative rating is divided by the maximum value of ratings per criterion and for non-beneficial criteria, minimum value of ratings per criterion is divided by each alternative rating. The concept of Ratio System can be also found in other decision-making methods like WASPAS and (Multi-Objective Optimization by Simple Ratio Analysis) (MOOSRA). The first term of WASPAS utility is inspired by Ratio System. In MOOSRA, the beneficial sum is divided by the non-beneficial sum while in Ratio System, the non-beneficial sum is subtracted from the beneficial sum.

• Reference Point Approach

In Reference Point Approach, the best alternative is the one that its worst value in respect of all criteria is not very bad [52]. This approach, as a non-compensatory model, first finds the alternatives ratings with the worst performance with respect to each criterion and finally selects the overall best value (i.e., the minimum value) from these worst ratings. Reference Point Approach is based on Tchebycheff Min-Max Metric [45]. Tchebycheff Min-Max Metric is originated from the general theory of Murkowski Metric which is the source of several decision analysis approaches in literature such as Goal Programming. To obtain the utility, first, Maximal Objective Reference Point (MORP) Vector is defined as [45]:

$$r_{j} = \left\{ \max_{i} x_{ij}^{*}, \quad j \le g; \quad \min_{i} x_{ij}^{*}, \quad j > g \right\}.$$
(II.5)

The distance between the weighted value of each member of MORP Vector and the weighted alternative rating is obtained as [54]:

$$d_{ij} = \left| w_j r_j - w_j x_{ij}^* \right|_{.}$$
(II.6)

The utility of Reference Point Approach is obtained by maximizing the distance introduced in Eq. (II.6) as follows [54]:

$$z_i = \max_j d_{ij} \tag{II.7}$$

The best alternative based on Reference Point Approach has the minimum utility z_i and the ranking of the approach is produced in ascending order as:

$$\mathbf{R}_{\mathrm{RPA}} = \left\{ A_{i \mid \min_{i} z_{i}} \succ \cdots \succ A_{i \mid \max_{i} z_{i}} \right\}.$$
(II.8)

In Reference Point Approach, the distance of each alternative rating from MORP Vector is obtained. There are other forms of Reference Point Vectors in the literature, including:

- Utopian Objective Reference Point (UORP) Vector: In this vector, higher values are targeted not the maximum values, necessarily;
- Aspiration Objective Reference Point (AORP) Vector: This vector tries to moderate aspirations as finding the
 maximum distance from the target values; that is, finding the alternatives with the worst performance.

TOPSIS and VIKOR also fall into the group of "Goal or Reference Level Models". Both of them are based on L_P-

Metric. TOPSIS is supported on L_2 while VIKOR is formulated on the basis of L_1 and L_{∞} . In TOPSIS, there exist two Reference Points, including the Positive-Ideal Solution (PIS) inspired by MORP and the Negative-Ideal Solution (NIS) inspired by AORP. In Classical Reference Point Approach, only MORP Vector is considered without paying attention to AORP Vector, but in Extended Reference Point Approach suggested by Eghbali-Zarch et al. [55], AORP Vector is also taken into account. Reference Point Approach sometimes cannot differ on two or more alternatives; that is, the approach leads to same rankings [56]. Thus, Reference Point Approach is often integrated with other decision-making tools to remedy the defect.

• Full Multiplicative Form

Full Multiplicative Form, which uses the geometric weighted aggregation operator, is an incompletely-compensatory model. In this technique, small normalized values of an alternative could not be completely compensated by the same degree of large values. Thus, the issue leads to the perception that an alternative with moderate performance may be

superior to an alternative which has both good and bad performances with respect to different criteria [52]. To obtain the utility of Full Multiplicative Form, the product of weighted normalized alternatives ratings on beneficial criteria are divided by the product of weighted normalized alternatives ratings on non-beneficial criteria [53]:

$$u_{i} = \prod_{j=1}^{g} \left(x_{ij}^{*}\right)^{w_{j}} / \prod_{j=g+1}^{n} \left(x_{ij}^{*}\right)^{w_{j}}$$
(II.9)

In utility formula of Full Multiplicative Form, multiplying normalized ratings with weights leads to the same result as the situation in which no weights are considered. Thus, weights should be considered as exponent in utility equation of Full Multiplicative Form. The best alternative based on Full Multiplicative Form has the maximum utility u_i and the ranking of this technique is generated in descending order as:

$$\mathbf{R}_{\text{FMF}} = \left\{ A_i | \max_i u_i \succ \cdots \succ A_i | \min_i u_i \right\}.$$
(II.10)

The concept of Full Multiplicative Form can be observed in other decision-making techniques like WASPAS. That is, the second term of WASPAS utility index is similar to Full Multiplicative Form. However, WASPAS uses a linear ratio for normalization considering the maximum and minimum values of alternatives ratings.

• Dominance Theory

Dominance Theory was used in the original MULTIMOORA method. This theory is supported on some principles including Dominance (Absolute Dominance and Partial Dominance), Equality (Absolute Equality, Partial Equality, and Equality according to Circular Reasoning), and Transitiveness [51]. There are some drawbacks to utilizing Dominance Theory: (1) obtaining ranks of alternatives is hard as the theory is not yet automated [6]; (2) the theory only uses ordinal values by neglecting the relative importance of alternatives; and (3) circular reasoning happens in some cases which leads to identical ranks which is not satisfactory [52].

• Borda and Improved Borda Rules

Borda Rule, also named Borda Count, is an easy but effective technique from the group of single-winner election methods in which the number of votes equals to the number of alternatives [57]. In this method, if there are t alternatives, the firstranked alternative gets t votes and the second-ranked gets one vote less, and so on. The final score of Borda Rule is computed by the summation of the scores of the subordinate methods. The highest value of Borda Rule score shows the best alternative.

Improved Borda Rule is based on Borda Count [58]; however, it integrates both cardinal and ordinal values (i.e., utilities and rankings, respectively) of each subordinate methods of MULTIMOORA. In this sense, the Improved Borda Rule is superior to Dominance Theory. To employ the Improved Borda Rule, first, the subordinate utilities are normalized based on Vector Normalization to produce y_i^* , z_i^* , and u_i^* . The assessment value of Improved Borda Rule, i.e., $IMB(A_i)$, is obtained using the following equation [58]:

$$IMB(A_i) = y_i^* \frac{m - r(y_i) + 1}{m(m+1)/2} - z_i^* \frac{r(z_i)}{m(m+1)/2} + u_i^* \frac{m - r(u_i) + 1}{m(m+1)/2}$$
(II.11)

where $r(y_i)$, $r(z_i)$, and $r(u_i)$ are the rankings of Ratio System, Reference Point approach, and Full Multiplicative Form, respectively. The best alternative based on Improved Borda Rule has the maximum value of $IMB(A_i)$.

Remark: Dominance Theory is complicated due to pairwise comparisons and probable occurrence of circular reasoning. The case would be more confusing for decision-makers when the number of alternatives and criteria are large because Dominance Theory is based on manual comparison. Nevertheless, Improved Borda Rule neither needs any manual comparison, nor has special conditions.

• Robustness of MULTIMOORA

In Table II.1, the performance of MOORA which is a part of MULTIMOORA is compared with other decision-making methods. As we can find from Table II.1, MOORA is simple and reliable. Original MULTIMOORA combines MOORA with the full multiplicative form using the dominance theory. Brauers and Zavadskas [51] claimed that "use of two different methods of multi-objective optimization is more robust than the use of a single method; the use of three methods is more robust than the use of two, and so on;" thus, "MULTIMOORA is more robust than MOORA".

Table II.1. Performance of MOORA regarding other decision-making methods [51].

MCDM method	Computational time	Simplicity	Mathematical calculation	s Stability	Information type		
MOORA	Very less	Very simple	Minimum	Good	Quantitative		
AHP	Very less	Very critical	Maximum	Poor	Mixed		
TOPSIS	Moderate	Moderately critical	Moderate	Medium	Quantitative		
VIKOR	Less	Simple	Moderate	Medium	Quantitative		
ELECTRE	High	Moderately critical	Moderate	Medium	Mixed		
PROMETHEE	High	Moderately critical	Moderate	Medium	Mixed		

Generally, the advantages of MULTIMOORA include: (1) simple mathematics, (2) low computational time, (3) straightforwardness for decision-makers, (4) using three different methods for determining subordinate rankings, and (5) employing ranking aggregation tools for integrating the subordinate rankings. To clarify item (5), it is worthwhile to mention that many decision-making methods have only one utility function; however, MULTIMOORA produces an integrative outcome by combining three utility values employing a ranking aggregation tool.

The three subordinate parts of MULTIMOORA are based on the fully compensatory, non-compensatory, and incompletely-compensatory models. As discussed in Subsections 2.2, each of the approaches may have some shortcomings, in practice. Therefore, integration of their outcomes would lead to a more robust final result comparing to the individual outcomes by curing the existing defects.

• Graphical summary of MULTIMOORA theory

The concepts used in Subsections 2.2 and 2.3 to derive the model of MULTIMOORA can be summarized into five phases as illustrated in Fig. II.1. Decision matrix and weight vector are constructed in Phase 1. The decision matrix is normalized in Phase 2. The utilities of subordinate parts of MULTIMOORA, i.e., Ratio System, Reference Point Approach, and Full Multiplicative Form, are computed in Phase 3. Rankings of subordinate methods are produced in Phase 4. Eventually, the subordinate rankings are combined into final outcomes of MULTIMOORA in Phase 5.



Fig. II.1. Flowchart of MULTIMOORA phases.

2.2. MpMcDM with MULTIMOORA method

In real life, many significant decisions are made through a group of elites and experts rather than considering an individual decision-maker. In industries and factories, technical expert panel takes the crucial decisions on identifying plans and strategies, selecting staff, and exploiting available resources. In practical problems like legal systems, healthcare management, and social services, the significant decisions are usually made based on collective opinions of multiple advisors and experts. Sometimes, experts may have different fields and levels of expertise; therefore, in such cases, collective decisions are analyzed to handle the conflicts among various opinions [2].

Multiple-person decision-making can be done in cooperative or non-cooperative styles. Cooperative group decisions are significant in engineering, medical, and scientific fields while non-cooperative group decisions are common in economic and political areas. Even by considering cooperative group decisions, reaching a complete consensus among all members of the group on the eventual solution is nearly infeasible because decision-makers, who are supposed to have identical goals, may have some opinion conflicts, in practice [59].

Near a half of total studies on MULTIMOORA has a multiple-person decision-making structure, which shows the importance of multiple-person decision-making. In some studies on MULTIMOORA, multiple-person decision-making structure is employed to generate criteria weights and alternatives ratings on the criteria; however, in the others, multiple decision-makers only participate in criteria weighting procedure.

2.3. Developments of MULTIMOORA method under risk and uncertainty

• Uncertain developments based on interval number theory

Interval number theory is a simple but applicable concept of considering vagueness in decision-making problems. Interval numbers can be defined as: (1) an extension of a real number; (2) a degenerate flat fuzzy number without membership function; and (3) an α -cut of a fuzzy number. INs are important in decision-making problems, because: (1) INs require the minimum amount of data; (2) decision-makers could easily present the range of available data as interval numbers; and (3) INs are very practical as many data in real problems are essentially reported in the form of ranges.

There are four important MULTIMOORA extensions based on interval theory. Kracka et al. [60] proposed interval MULTIMOORA utilizing arithmetic of interval numbers (MOORE rule), the crisp distance of interval numbers, and comparison based on arithmetic average. Hafezalkotob and Hafezalkotob [11] suggested a new model of interval MULTIMOORA by using preference matrix without degeneration of interval numbers. Hafezalkotob and Hafezalkotob [23] presented interval target-based MULTIMOORA employing MOORE rule, interval distance of interval numbers, and the preference matrix. Hafezalkotob and Hafezalkotob [47] developed interval target-based MULTIMOORA by adding preference-based rankings of interval numbers.

• Uncertain developments based on fuzzy set theory

Fuzzy set theory, introduced by Zadeh [61] in 1965, is an important theory of uncertainty which models the vagueness or imprecision of the human cognitive process. A fuzzy set is generally introduced by a membership function that maps elements to degrees of membership in a certain interval [62]. The theory is very applicable in various fields such as decision making, artificial intelligence, expert systems, control theory, and neural networks. There are different types of fuzzy sets like interval-valued fuzzy number, intuitionistic fuzzy number, and interval type-2 fuzzy set [7].

As fuzzy theory is one of most important concepts of uncertainty, there are many extensions of MULTIMOORA based on this theory. Triangular fuzzy number is the simplest form of representing the fuzziness of data. Triangular fuzzy number with mathematical features such as Vertex method for crisp distance and centroid-based method for defuzzification has combined with MULTIMOORA in several studies [6], [55], [71]-[74], [63]-[70]. However, Tian et al. [62] employed graded mean integration as defuzzification technique to generate triangular fuzzy MULTIMOORA. Trapezoidal fuzzy number with concepts of Vertex method for crisp distance and centroid-based method for defuzzification is used for three developments [3], [5], [26]. Liu et al. [75] applied the integral of area for defuzzification to derive Trapezoidal Fuzzy MULTIMOORA. Stanujkic et al. [76] suggested interval-valued fuzzy MULTIMOORA based on the weighted averaging operator and the geometric averaging operator of interval-valued fuzzy numbers. Dorfesh et al. [77] suggested Interval type-2 fuzzy MULTIMOORA. Generalized interval-valued fuzzy number is a basis for four developments [78]-[81]. In these studies, centroid-based method is used for crisp distance of Generalized interval-valued fuzzy numbers and defuzzification is also based on the crisp distance. Baležentis and Baležentis [82] introduced intuitionistic fuzzy MULTIMOORA based on the power ordered weighted average operator and the power ordered weighted geometric operator as well as Euclidean distance and expected values of interval-valued fuzzy numbers. Baležentis et al. [83] presented another version of intuitionistic fuzzy MULTIMOORA using negation operator, the power ordered weighted average operator, the power ordered weighted geometric operator, comparison rule, and crisp distance of interval-valued fuzzy numbers. Interval-valued intuitionistic fuzzy MULTIMOORA has been developed considering the weighted average operator, the weighted geometric operator, and score function of interval-valued fuzzy numbers [84], [85]. Hesitant fuzzy set was exploited in three studies [86]–[88] for new developments.

3. Application of MULTIMOORA method

The applications of MULTIMOORA in the sector of Industries are divided into the following subsectors: Construction, Automotive, Agricultural, Mining, Entertainment, Logistics, Steel, Aviation, Beverage, Carpentry, Energy, Ship-Building, and Textile Industries, besides Manufacturing System. In Construction Industry subsector, there are several case studies related to Buildings Revitalization Appraisal [84], Project Management [77], [89], and Ranking Countries/Cities/Regions [90] besides the selection of Investment [84], Component [91], [92], Design [60], [93], [94], Material [92], Supplier [95], and Technology [96]. In Automotive Industry subsector, there are multiple case studies related to Battery Recycling Mode Selection [97] and Location Planning [98] as well as the selection of Material [26], [53], Robot [99], Supplier [56], and Vehicle [57], [100], [101]. In Agricultural Industry subsector, the case studies include Farming Efficiency Estimation [68] and the selection of Crop [73], Machine [102], and Supplier [75]. In Mining Industry subsector, there exist four case studies related to Design Selection [76], [103], Mining Technique Selection [104], and Personnel Management [105]. In Entertainment Industry subsector, two case studies have considered the problems regarding Partner Selection [6] and Transportation Efficiency Evaluation [107]. In Manufacturing System subsector, the practical cases are Enterprise Resource Planning [108] and the selection of Design [109], Machine [23], [80], and Material [11], [53]. In Steel Industry subsector, two researches exist in respect to Risk Evaluation [69], [85].

For other subsectors of Industries sector, there is only one case study. Dorfeshan et al. [77] evaluated a project management problem in the area aircraft component development planning. Çebi and Otay [74] tackled a supplier selection problem in a company operating in beverage industry. Stojić et al. [38] assessed selection process of supplier for a PVC carpentry manufacturing company. Hafezalkotob and Hafezalkotob [3] handled material selection process for the blades of industrial gas turbine. Qin and Liu [110] chose a suitable supplier for purchasing components of ship equipment. Brauers and Zavadskas [111] undertook a project management problem for Tunisian textile industry.

The applications of MULTIMOORA in the sector of Medical/Healthcare Management are divided into the following subsectors: Medical Service, Biomedical Service, and Health-Care Management. In Medical Service subsector, there is one case study related to pharmacological selection of type 2 diabetes [55]. In Biomedical Service subsector, two studies has conducted on the selection process of biomaterials for hip and knee surgical prostheses [47], [54]. In Health-Care Management subsector, three case studies have handled Risk Evaluation and Waste Management. Liu et al. [5] used the concept of failure mode and effects analysis to prevent infant abduction from hospitals. Two researched analyzed the treatment technologies regarding health-care waste management in Shanghai, China [63], [112].

Chapter III

I_S-MpMcDM methodology: Industrial application

In this chapter, we present the I_S-MpMcDM methodology. The methodology is based on interval numbers and modeled by the BWM-MULTIMOORA approach. The method with complete interval computation in which interval distance of interval numbers and preference matrix are used. In addition, we propose a multi-person interval best-worst method with interval preference degree. The multi-person interval best-worst method has two levels of experts.

We introduce the interval Borda rule as an aggregation function which does not have the defects of the dominance theory. We calculate objective interval weights of criteria based on interval entropy method, which are integrated by the subjective weights computed by the multi-person interval bestworst method.

The I_S-MpMcDM methodology is employed to tackle a real-world engineering selection problem of hybrid vehicle engine based on real data and opinions of engineering design experts of automotive industry of Iran.
1. Introduction

In many practical decision-making problems, the available data are associated with some degrees of vagueness. In this regard, different kinds of uncertain sets are applied to tackle the vagueness of data in the context of MpMcDM, including fuzzy sets and interval numbers and their extensions [7], [113]. Interval numbers are the simple form of embodying uncertainty in decision-making problems, which can be defined as: (1) an extension of a real number; (2) a degenerate flat fuzzy number without membership function; (3) an α -cut of a fuzzy number. The merits of interval numbers in MpMcDM problems can be denoted as [3]: (1) interval numbers need the minimum amount of data; (2) decision-maker can easily obtain the range of available data as interval numbers; (3) interval numbers can be very practical. In many real-life problems particularly industrial cases such as material selection [3] and structural systems selection [4], the input data is based on measurements reported as interval numbers. Many MpMcDM methods were extended using the theory of interval numbers such as ELECTRE [4], TOPSIS [115], and VIKOR [6].

The MULTIMOORA technique is a widely used MpMcDM method due to its simple mathematics, low computational time and straightforwardness for decision-makers [46]. It aggregates ranks obtained from three subordinate models, namely, the reference point model, the ratio model, and the full multiplicative model. The MULTIMOORA method has been developed based on various uncertainty theories, including interval numbers [3], fuzzy sets [8], interval-valued fuzzy numbers [9], double hierarchy hesitant fuzzy linguistic term set [117] and the probabilistic linguistic term set [58]. The method is utilized in some practical applications, including mining method selection [12], performance appraisal method assessment [13], and logistics service provider selection [6]. Hafezalkotob et al. [16] reviewed and discussed the important studies on the MULTIMOORA.

Determining the weights of criteria is a critical step in MpMcDM problems [119]. For subjective weighting methods based on the judgments of decision-makers, the best-worst method (BWM) [15] which compares each criterion with the best and worst criteria is more effective than analytic hierarchy process (AHP) which makes pairwise comparisons. In the BWM, the weights of criteria are produced by solving a max-min problem. The method has been developed using uncertainty theories, including intuitionistic fuzzy sets [120], intuitionistic fuzzy multiplicative numbers [121], and Z-numbers [122]. For objective weighting methods based on the decision matrix, Shannon entropy is a powerful tool to calculate weights according to the contrast of information [3]. Shannon entropy measures diversity of data in a decision matrix. There are some extensions of the entropy method based on interval and fuzzy numbers [3], [26]. The entropy-based weighting model has been integrated with the MULTIMOORA method in a few studies [26], [118].

The I_S-MpMcDM methodology is motivated by the interval target-based MULTIMOORA [47], the fuzzy BWM based on the combination of group and individual decisions [2] and the enhanced Borda rule [123]. To tackle some open problems detected in the literature analysis and compared with the three aforementioned studies, the main contributions of this work are discussed in the following five items:

We suggest an interval MULTIMOORA with entire interval computation. That is, no degradation of interval data is performed in the proposed interval MULTIMOORA model. For this purpose, interval distances between interval numbers and preference matrix are used. Hafezalkotob and Hafezalkotob [47] also proposed the interval distances between interval numbers and preference matrix; however, they only introduced a special form of interval MULTIMOORA with target-based criteria. In this chapter, we enhance the original MULTIMOORA [7], which is based on beneficial and non-beneficial criteria, by considering the multiple theories of interval numbers.

- We present a multi-person interval BWM. In Ref. [2], fuzzy triangular preference degree was considered in the multi-person fuzzy BWM model. By contrast, we use interval preference degree with a different multi-person decision-making structure.
- We develop the interval Borda rule to integrate ranks. We propose an interval form of the enhanced Borda rule [123] to integrate the assessment values and ranks of subordinate parts of interval MULTIMOORA to a final assessment values.
- We use interval entropy method to determine the objective weights of criteria. The interval objective weights are then integrated with the subjective weights obtained from the multi-person interval BWM.
- We apply the proposed model in a practical case of selecting hybrid vehicle engines. In many engineering problems like this case study, the ratings of alternatives on their criteria are intrinsically given as interval numbers. Thus, the developed interval decision-making method can be utilized in such significant real-world engineering problems.

The remainder of this chapter is arranged as follows. The derivation of the proposed methodology including the interval BWM, interval entropy, interval MULTIMOORA, and its subordinate parts are presented in Section 2. We present a real-world industrial decision-making problem in Section 3.

2. Theory of I_S-MpMcDM methodology

The decision matrix with interval ratings $\overline{x}_{ij} = [x_{ij}^L, x_{ij}^U]$ of alternatives A_i , i = 1, ..., m, on criteria C_j , j = 1, ..., n, can be represented as follows [124]:

$$\overline{\mathbf{X}} = \begin{bmatrix}
\overline{x}_{11} & \cdots & \overline{x}_{1j} & \cdots & \overline{x}_{1n} \\
\overline{x}_{11} & \cdots & \overline{x}_{1j} & \cdots & \overline{x}_{1n} \\
\vdots & \vdots & \vdots & \vdots \\
\overline{x}_{i1} & \cdots & \overline{x}_{ij} & \cdots & \overline{x}_{in} \\
\vdots & \vdots & \vdots & \vdots \\
\overline{x}_{m1} & \cdots & \overline{x}_{mj} & \cdots & \overline{x}_{mn}
\end{bmatrix} A_{m}$$
(III.1)

The flowchart of the proposed methodology is illustrated in Fig. III.1:



Fig. III.1. Flowchart of the I_S-MpMcDM methodology.

2.1. BWM-based model for weighting process

Subjective weights: Multi-person interval best-worst-method-based weighting model

Hafezalkotob and Hafezalkotob [2] suggested a group fuzzy BWM. They proposed that director evaluates both the expertise degrees of the panel members and the relative importance of criteria. We propose a group interval BWM. The decision-making structure of the proposed group interval BWM is different from the study of Hafezalkotob and Hafezalkotob [2]. The steps of group interval BWM are as follows:

Step 1. Consider a "director" and expert panel {Expert₁, Expert₁, ..., Expert_k, ..., Expert_k}. In the multi-person interval BWM, the director evaluates the relative importance of the members of the panel in interval numbers, and the expert panel determines the relative importance of the criteria $\{C_1, C_2, ..., C_j, ..., C_n\}$ in interval numbers as well (see Fig. III.2).

Step 2. Determine the best and worst experts by the director.

Step 3. Assess expertise degrees of the panel by the director:

Step 3.1. Specify the interval preference degree $\bar{p}_{Bk} = [p_{Bk}^L, p_{Bk}^U]$ of the best expert *B* who has the highest expertise and knowledge over each expert *k* in the panel. The interval best-to-others vector of the expertise degrees of the panel is: $\bar{P}_B = (\bar{p}_{B1}, \bar{p}_{B2}, ..., \bar{p}_{Bt}).$



Fig. III.2. Supervised multi-person structure in the I_S-MpMcDM methodology.

Step 3.2. Specify the interval preference degree $\bar{p}_{kW} = \left[p_{kW}^L, p_{kW}^U \right]$ of expert *k* over the worst expert *W* who has the lowest expertise and knowledge in the panel. The interval others-to-worst vector of the expertise degrees of the panel is: $\bar{P}_W = \left(\bar{p}_{1W}, \bar{p}_{2W}, \dots, \bar{p}_{tW} \right)^T$.

Step 3.3. Calculate the optimal weights vector $(\lambda_1^*, \lambda_2^*, ..., \lambda_t^*)$ for the expert panel using the following min-max model: min max $\{ |\lambda_B - \overline{p}_{Bk} \lambda_k|, |\lambda_k - \overline{p}_{kW} \lambda_W| \}$ s.t., $\sum_{k=1}^{t} \lambda_k = 1, \quad \lambda_k \ge 0$, for all k(III.2)

Model (III.2) can be converted into a linear programming form as follows:

min ε ,

s.t.,
$$\left|\lambda_{B} - \overline{p}_{Bk}\lambda_{k}\right| \leq \varepsilon$$
, for all k
 $\left|\lambda_{k} - \overline{p}_{Wk}\lambda_{W}\right| \leq \varepsilon$, for all k
 $\sum_{k=1}^{l}\lambda_{k} = 1, \ \lambda_{k} \geq 0$, for all k
(III.3)

In Model (III.3), symbol \leq denotes 'almost lesser than' constraint and leads to some degree of "smaller" condition considering the interval numbers \overline{p}_{Bk} and \overline{p}_{Wk} . Model (III.3) is equivalent to the following form based on the definition of absolute values of $|\lambda_B - \overline{p}_{Bk} \lambda_k|$ and $|\lambda_k - \overline{p}_{Wk} \lambda_W|$:

min ε

s.t.,
$$\lambda_B - \varepsilon \leq \overline{p}_{Bk} \lambda_k$$
, for all k
 $\lambda_B + \varepsilon \geq \overline{p}_{Bk} \lambda_k$, for all k
 $\lambda_k - \varepsilon \leq \overline{p}_{Wk} \lambda_W$, for all k
 $\lambda_k + \varepsilon \geq \overline{p}_{Wk} \lambda_W$, for all k
 $\sum_{k=1}^{t} \lambda_k = 1, \ \lambda_k \geq 0$, for all k
(III.4)

According to the crisp equivalents of the interval constraints in Model (III.4), the model can be transformed into the following form:

min ε

s.t.,
$$\lambda_{B} - \varepsilon \leq \left[p_{Bk}^{L} + \left(p_{Bk}^{U} - p_{Bk}^{L} \right) \alpha \right] \lambda_{k}$$
, for all k
 $\lambda_{B} + \varepsilon \geq \left[p_{Bk}^{U} - \left(p_{Bk}^{U} - p_{Bk}^{L} \right) \alpha \right] \lambda_{k}$, for all k
 $\lambda_{k} - \varepsilon \leq \left[p_{Wk}^{L} + \left(p_{Wk}^{U} - p_{Wk}^{L} \right) \alpha \right] \lambda_{W}$, for all k
 $\lambda_{k} + \varepsilon \geq \left[p_{Wk}^{U} - \left(p_{Wk}^{U} - p_{Wk}^{L} \right) \alpha \right] \lambda_{W}$, for all k
 $\sum_{k=1}^{L} \lambda_{k} = 1, \ \lambda_{k} \geq 0$, for all k
(III.5)

where α ($0 \le \alpha \le 1$) denotes a possibility level which can be defined by the experts. Model (III.5) is linear; thus, solving the problem for any given possibility level α leads to a unique optimal weight vector $(\lambda_1^*, \lambda_2^*, ..., \lambda_l^*)$ and a corresponding objective value ε^* . ε^* denotes the consistency level of the interval preference relation. If $\varepsilon^* = 0$ for all α , we have $\lambda_B/\lambda_j = \overline{p}_{Bk}$ and $\lambda_j/\lambda_W = \overline{p}_{Bk}$, and correspondingly $\overline{p}_{Bk} \times \overline{p}_{kW} = \overline{p}_{BW}$ for all k. Thus, the interval preference relation \overline{P}_B and \overline{P}_W are fully consistent.

Step 4. Multi-person decision-making process: Each expert specifies the best and worst criteria from his/her attitude, and the interval preference degree $\overline{q}_{Bj}^{[k]} = \left[q_{Bj}^{[k],L}, q_{Bj}^{[k],U}\right]$ of the best attribute *B* over each attribute *j*. The interval best-to-others vector of criteria determined by expert *k* is $\overline{Q}_{B}^{[k]} = (\overline{q}_{B1}^{[k]}, \overline{q}_{B2}^{[k]}, ..., \overline{q}_{Bn}^{[k]})$. In analogous, each expert specifies the

interval preference degree $\overline{q}_{jW}^{[k]} = \left[q_{jW}^{[k],L}, q_{jW}^{[k],U}\right]$ of each attribute j over the worst attribute W and establishes the interval others-to-worst vector of criteria as $\overline{Q}_{W}^{[k]} = \left(\overline{q}_{1W}^{[k]}, \overline{q}_{2W}^{[k]}, ..., \overline{q}_{nW}^{[k]}\right)^{T}$.

Step 5. Obtain the optimal weight vector of criteria $(W_1^*, W_2^*, ..., W_n^*)$ by the following model:

$$\min \sum_{k=1}^{l} \lambda_{k}^{*} \max_{j} \left\{ \left| w_{B} - \overline{q}_{Bj}^{[k]} w_{j} \right|, \left| w_{j} - \overline{q}_{jW}^{[k]} w_{W} \right| \right\}$$

$$\text{s.t.,} \quad \sum_{j=1}^{n} w_{j} = 1, \quad w_{j} \ge 0, \quad \text{for all } j$$

$$(III.6)$$

where λ_k^* is the optimal weight of expert k obtained from Step 3.3. Model (III.6) can be converted to the following form:

$$\min \Psi = \sum_{k=1}^{l} \lambda_{k}^{*} \varepsilon_{k}$$
s.t., $\begin{vmatrix} w_{B} - \overline{q}_{Bj}^{[k]} w_{j} \end{vmatrix} \leq \varepsilon_{k}$, for all j
 $\begin{vmatrix} w_{j} - \overline{q}_{jW}^{[k]} w_{W} \end{vmatrix} \leq \varepsilon_{k}$, for all j for all k
 $\sum_{j=1}^{n} w_{j} = 1, \ w_{j} \geq 0$, for all j

$$(III.7)$$

Similar to Model (III.5), a crisp equivalent can be formulated for Model (III.7); however, we ignore the detail explanation here for briefness. The crisp equivalent of Model (III.7) has a linear form; thus, its solution leads to a unique optimal weight vector $(w_1^*, w_2^*, ..., w_n^*)$ and the consistency levels $(\mathcal{E}_1^*, \mathcal{E}_1^*, ..., \mathcal{E}_t^*)$ for any given possibility level α . $\Psi = 0$ means the full consistency of interval preference relations of the experts panel, i.e., $\overline{Q}_B^{[k]}$ and $\overline{Q}_W^{[k]}$. However, the interval preference relation may not be so precise to reach the full consistency in practice. In $\overline{Q}_B^{[k]}$ and $\overline{Q}_W^{[k]}$, small deviations from the full consistency situation lead to small positive value of Ψ . Thus, the magnitude of Ψ is a measure of the consistency level of group decisions.

• Objective interval weights: Interval Shannon entropy weighting method

The following steps can be considered to calculate objective interval weights based on Shannon entropy [125] from the interval ratings.

Step 1. Normalize the interval rating \overline{x}_{ij} to obtain the interval project outcome $\overline{p}_{ij} = \left[p_{ij}^L, p_{ij}^U\right]$ based on linear normalization ratio:

$$p_{ij}^{L} = \frac{x_{ij}^{U}}{\sum_{i=1}^{m} x_{ij}^{U}}, \quad p_{ij}^{U} = \frac{x_{ij}^{U}}{\sum_{i=1}^{m} x_{ij}^{U}}$$
(III.8)

Step 2. Obtain interval Shannon entropy measure $\overline{E}_j = \left[E_j^L, E_j^U\right]$ based on the interval project outcomes:

$$E_{j}^{L} = \min\left\{-E_{0}\sum_{i=1}^{m} \left(p_{ij}^{L} \ln p_{ij}^{L}\right), -E_{0}\sum_{i=1}^{m} \left(p_{ij}^{U} \ln p_{ij}^{U}\right)\right\}$$

$$E_{j}^{U} = \max\left\{-E_{0}\sum_{i=1}^{m} \left(p_{ij}^{L} \ln p_{ij}^{L}\right), -E_{0}\sum_{i=1}^{m} \left(p_{ij}^{U} \ln p_{ij}^{U}\right)\right\}$$
(III.9)

where $E_0 = 1/\ln(m)$.

Step 3. Compute the interval diversification $\overline{\delta}_j = \left[\delta_j^L, \delta_j^U\right]$, where

$$\delta_j^L = 1 - E_j^U, \quad \delta_j^U = 1 - E_j^L \tag{III.10}$$

Step 4. Specify the objective interval weight $\overline{w}_{j}^{\rho} = \left[w_{j}^{\rho,L}, w_{j}^{\rho,U}\right]$, where

$$w_{j}^{o,L} = \frac{\delta_{j}^{L}}{\sum_{j=1}^{n} \delta_{j}^{U}}, \qquad w_{j}^{o,U} = \frac{\delta_{j}^{U}}{\sum_{j=1}^{n} \delta_{j}^{U}}$$
(III.11)

The higher Shannon entropy measure \overline{E}_j of an attribute is, the smaller variation degree of interval ratings on attribute j will be, and the existent information on the attribute is less. Then its resultant objective interval weight \overline{w}_j^{ρ} will be lower [3].

• Integrating interval weights of criteria

The subjective weight w_i^* determined from Model (III.7) and the objective interval weight \overline{w}_i^o determined from Eq. (III.11)

can be combined to produce the integrated interval weight $\overline{s}_j = \left[s_j^L, s_j^U\right]$, where

$$s_{j}^{L} = \frac{w_{j}^{*}w_{j}^{o,L}}{\sum_{j=1}^{n}w_{j}^{*}w_{j}^{o,U}}, \qquad s_{j}^{U} = \frac{w_{j}^{*}w_{j}^{o,U}}{\sum_{j=1}^{n}w_{j}^{*}w_{j}^{o,U}}$$
(III.12)

2.2. MULTIMOORA-based model for ranking process

Interval ratings of alternative \overline{x}_{ij} are converted into normalized interval value $\overline{x}_{ij}^* = \begin{bmatrix} x_{ij}^{*,L}, x_{ij}^{*,U} \end{bmatrix}$; thus, $x_{ij}^{*,L}$ and $x_{ij}^{*,U}$ are within the range [0, 1]. Stanujkic *et al.* [126] introduced a robust normalization formula for $\overline{x}_{ij}^* = \begin{bmatrix} x_{ij}^{*,L}, x_{ij}^{*,U} \end{bmatrix}$ as:

$$\begin{aligned} x_{ij}^{*,L} &= x_{ij}^{L} / \sqrt{\frac{1}{2} \sum_{i=1}^{m} \left[\left(x_{ij}^{L} \right)^{2} + \left(x_{ij}^{U} \right)^{2} \right]} \\ x_{ij}^{*,U} &= x_{ij}^{U} / \sqrt{\frac{1}{2} \sum_{i=1}^{m} \left[\left(x_{ij}^{L} \right)^{2} + \left(x_{ij}^{U} \right)^{2} \right]} \end{aligned}$$
(III.13)

• The interval ratio system

Based on arithmetic operations of interval numbers given in Eq. (I.2), and by considering the integrated interval weight \overline{s}_j determined by Eq. (III.13), the interval assessment value \overline{y}_i is calculated as:

$$\overline{y}_i = \sum_{j=1}^{g} \overline{s}_j \overline{x}_{ij}^* - \sum_{j=g+1}^{n} \overline{s}_j \overline{x}_{ij}^*$$
(III.14)

The best alternative A_{I-RS}^* has the maximum value of \overline{y}_i which can be found using Eq. (I.8):

$$A_{\text{LRS}}^* = \left\{ A_i | \max_i \bar{y}_i \right\}$$
(III.15)

• The interval reference point approach

Employing Eqs. (I.8) and (I.9), the interval reference point is:

$$\overline{r}_{j} = \left\{ \max_{i} \overline{x}_{ij}^{*}, \quad j \le g; \qquad \min_{i} \overline{x}_{ij}^{*}, \quad j > g \right\}$$
(III.16)

The deviation between the weighted normalized interval rating $s_j \overline{x}_{ij}^*$ and the interval weighted reference point $w_j \overline{r}_j$ is:

$$\overline{d}_{ij}^* = \overline{d}^* \left(\overline{s}_j \overline{r}_j, \overline{s}_j \overline{x}_{ij}^* \right)$$
(III.17)

where \overline{d}^* is an interval distance computed by Eq. (I.4). The interval assessment value of the interval reference point \overline{z}_i is determined as follows:

$$\overline{Z}_i = \max_j \, \overline{d}_{ij}^* \tag{III.18}$$

Based on Eq. (I.9), the optimal alternative A^*_{I-RP} has the minimum interval assessment value:

$$A_{\text{L-RP}}^* = \left\{ A_i | \min_i \bar{z}_i \right\}$$
(III.19)

• The interval full multiplicative form

Based on the arithmetic operations of interval numbers shown as Eq. (I.2) as well as the integrated interval weights \overline{s}_j , the interval assessment value \overline{u}_i of the interval full multiplicative form is calculated as follows:

$$\bar{u}_{i} = \frac{\prod_{j=1}^{s} (\bar{x}^{*})^{S_{j}}}{\prod_{j=g+1}^{n} (\bar{x}^{*})^{\bar{S}_{j}}}$$
(III.20)

Because all the elements \overline{x}^* and \overline{s}_j are between 0 and 1, it is clear that:

$$\left(\overline{x}^{*}\right)^{\overline{s}_{j}} = \left[\left(x_{ij}^{L}\right)^{s_{j}^{U}}, \left(x_{ij}^{U}\right)^{s_{j}^{L}}\right]$$
(III.21)

The best alternative A_{I-MF}^* has the maximum value of \overline{u}_i , which can be found using Eq. (I.8):

$$A_{\text{I-MF}}^* = \left\{ A_i \big| \max_i \bar{u}_i \right\}$$
(III.22)

• The rankings of the interval MULTIMOORA

Benshan [123] presented the enhanced Borda rule by consolidating the rankings and assessment values into aggregating assessment values. In this study, we propose the interval Borda rule to integrate the interval assessment values and rankings derived from the three models of interval MULTIMOORA. The interval Borda rule is superior to the dominance theory from the perspectives of mathematics and straightforwardness for decision-makers. Mathematically speaking, the

dominance theory only employs the ordinal rankings for aggregation; however, the interval Borda rule considers both cardinal interval assessment values and the ordinal rankings. From the aspect of application, the dominance theory is complicated because of pairwise comparisons and probable occurrence of circular reasoning in particular. The case would be confusing for decision-makers when the number of alternatives and criteria are large. In contrast, the interval Borda rule does not require any manual comparison and has special conditions.

To use the interval Borda rule, the first step is to normalize the subordinate interval assessment values, i.e., Eqs. (III.14), (III.18), and (III.20), to obtain $\overline{y}_i^* = \left[y_i^{*,L}, y_i^{*,U}\right], \overline{z}_i^* = \left[z_i^{*,L}, z_i^{*,U}\right], \text{ and } \overline{u}_i^* = \left[u_i^{*,L}, u_i^{*,U}\right].$ The method to compute $\overline{y}_i^*, \overline{z}_i^*$, and \overline{u}_i^* is identical to Eq. (III.13).

We define the interval Borda rule to obtain the interval assessment values of the interval MULTIMOORA as follows:

$$\overline{IMB_i} = \overline{y}_i^* \frac{m - r(\overline{y}_i) + 1}{m(m+1)/2} - \overline{z}_i^* \frac{r(\overline{z}_i)}{m(m+1)/2} + \overline{u}_i^* \frac{m - r(\overline{u}_i) + 1}{m(m+1)/2}$$
(III.23)

The best alternative based on the interval MULTIMOORA- Borda has the maximum value of IMB; that can be found by Eq. (I.8):

$$A_{\text{I-MULTIMOORA-Borda}}^* = \left\{ A_i | \max_i \overline{IMB_i} \right\}$$
(III.24)

3. Application of I S-MpMcDM methodology in industrial sector

This section discusses a practical problem in industrial sector. The case study concerns the selection of appropriate hybrid vehicle engine selection tackled using the proposed I S-MpMcDM methodology as shown in Fig. III.3. Fig. III.4 illustrates the supervised multi-person structure for the case study.



Fig. III.3. The case study of the I S-MpMcDM methodology: Hybrid vehicle engine selection.



Fig. III.4. Supervised multi-person structure for the case study of the I S-MpMcDM methodology.

Vehicles are producing a major part of carbon emissions in the world, which has resulted in climate change. The global warming issue and poor air quality are the results of traditional internal combustion engines (ICEs) which are based on fossil fuels, such as gasoline and diesel fuels [127]. In recent decades, however, the efforts for changing the fuel blend and technology of vehicle engines have been markedly increased [128]. Besides, the pressure from social communities have stimulated the development of cleaner and more efficient vehicles and the production of hybrid vehicles. Hybrid vehicles have much lower deleterious effects on environment.

There are different kinds of technologies for hybrid vehicle engines such as compact natural gas (CNG)-gasoline, plugin electric-gasoline, regenerative electric-gasoline, fuel cell-gasoline, and hydrogen-gasoline. Penetration rates of the category of hybrid electric cars are still low in big cities, mostly due to the high cost of batteries and the low number of existing charging stations [129]. Apart from the USA, Europe, Japan, and India also help in commercializing of hybrid electric cars [130]. In recent years, many leading automotive companies, such as Toyota, Ford, BMW, Honda, Chevrolet, Porsche, and McLaren, have been investing in hybrid vehicles. In addition to the firms, "Tesla" is a major company that both produces high-tech hybrid electric vehicles and batteries as well as solar panels. In late 2017, Tesla released its stateof-the-art electric truck named "Tesla Semi". The company claimed that "Semi has maximum power and acceleration and requires the lowest energy cost per mile (See tesla.com/semi)". The mentioned issues clarify that both heavy and light hybrid vehicles will get much more attention worldwide and by the improvements in the technology of batteries as well as optimizing the engine technology and electrical and electronic systems besides investing in charging stations, hybrid vehicles would definitely get a substantial market share of automotive industry in near future. Moreover, government policies including consumer incentives, taxation, investment in research and development, as well as incentives for automotive companies and consumers play a key role in the progress of growing industry of hybrid vehicles [131].

This case study is conducted on the basis of the collected information and comments of engineering design experts of Iran automotive industry (as shown Fig. III.4). Major automotive firms in Iran are researching into the possibility of improving the engine technology from traditional to hybrid engines. Six hybrid engines are taken into account as candidate alternatives in this case study, which include five items based on the type of CNG-gasoline (EF7 naturally aspirated, EF7 Turbo charged, TU5 naturally aspirated, XU7 naturally aspirated, and XUM naturally aspirated) besides one engine with the technology of electric-gasoline which is planned to be produced and assembled in near future. The prospective electric-gasoline (PEG) engine is a serial coupling of two engines, i.e., an internal combustion engine (ICE) based on gasoline fuel and an electric engine (motor/generator). Both ICE and electric engine (as motor) can drive the vehicle separately or simultaneously as coupled by clutch. Electric engine can also act as a generator and charges the batteries. Table III.1 shows the criteria of the hybrid engines and their units as well as the objective related to each criterion. The criteria of the hybrid engines are classified into three categories: technical, financial, and environmental criteria. Each category has items that may be beneficial or non-beneficial criteria.

Туре	Criterion name	Abb.*	Unit	Objective
	Combined ultimate power	CUP	hp	Maximum
Technical	Combined ultimate torque	CUT	N.m	Maximum
criteria	Fuel consumption (gasoline)	FCG	L/(100 km)	Minimum
	Index of durability and maintenance	IDM	_	Maximum
Einen sist	Relative price	RPC	-	Minimum
Financial	Relative needed foreign fund	RNF	-	Minimum
cinena	Job creation index	JCI	-	Maximum
Fnvironmental	Carbon monoxide emission	CME	g/km	Minimum
criteria	Nitrogen oxides emission	NOE	g/km g/km	Minimum

Table III.1. Criteria of candidate hybrid vehicle engines.

*Abb.: Abbreviation

The linguistic terms are obtained for IDM, RPC, RNF, and JCI based on the comments of design experts. The corresponding interval numbers for linguistic terms related to the four linguistic criteria are defined as shown in Table III.2. The decision matrix consists of the interval ratings of the hybrid engines on the ten criteria (i.e., the performances of the engines on each specifications) are given in Table III.3. The data of Table III.3 is collected based on the references of related engines.

Table III.2. Corresponding interval numbers for linguistic terms.

Linguistic terms	Corresponding interval number
Very high (VH)	[7,9]
High (H)	[6, 8]
Medium high (MH)	[5, 7]
Medium (M)	[4, 6]
Medium low (ML)	[3, 5]
Low (L)	[2, 4]
Very low (VL)	[1, 3]

Eng. ID ¹	Engine name	Technical criteria				Financial criteria			Environmental criteria		
		CUP	CUT	FCG	IDM	RPC	RNF	JCI	CME	THE	NOE
E_1	EF7 TC ²	[122, 150]	[171, 209]	[6.5, 7.9]	[6, 8]	[6, 8]	[6, 8]	[2, 4]	[0.51, 0.55]	[0.06, 0.07]	[0.08, 0.09]
E_2	EF7 NA ³	[96, 117]	[131, 160]	[6.8, 8.3]	[6, 8]	[6, 8]	[6, 8]	[3, 5]	[0.50, 0.53]	[0.07, 0.08]	[0.07, 0.08]
E_3	TU5 NA	[84, 102]	[109, 134]	[6.7, 8.1]	[5, 7]	[5, 7]	[4, 6]	[4, 6]	[0.54, 0.56]	[0.06, 0.07]	[0.10, 0.11]
E_4	XU7 NA	[79, 96]	[116, 142]	[8.3, 10.1]	[4, 6]	[4, 6]	[5, 7]	[6, 8]	[0.60, 0.63]	[0.08, 0.09]	[0.10, 0.11]
E_5	XUM NA	[84, 102]	[117, 143]	[7.9, 9.7]	[3, 5]	[5, 7]	[3, 5]	[6, 8]	[0.61, 0.64]	[0.08, 0.09]	[0.09, 0.10]
E_6	PEG ⁴	[110, 125]	[114, 174]	[4.7, 5.8]	[7, 9]	[7, 9]	[7, 9]	[4, 6]	[0.45, 0.48]	[0.05, 0.06]	[0.07, 0.08]

Table III.3. Decision matrix.

¹Eng. ID: Engine ID; ²TC: Turbo-charged; ³NA: Naturally-aspirated; ⁴PEG: Prospective electric-gasoline.

Table III.4 represents the linguistic preferences and the corresponding interval preferences. The design manager evaluates the expertise degrees of the designers by linguistic preferences. The corresponding interval preferences of the linguistic terms considered by the design manager are shown in Table III.5. Indeed, Table III.5 lists the interval best-to-others and others-to-worst vectors of the designers' expertise degrees.

Table III.4. Linguistic preferences versus the corresponding intervals.

Linguistic preferences	Corresponding interval preferences
Equally importance	[1,1]
Weakly importance	[1, 3]
Fairly importance	[3, 5]
Very importance	[5,7]
Absolutely importance	[7,9]

Table III.5. Expertise degrees of designers.

DM	Best & worst designers	Interv	Interval preferences obtained from linguistic terms					
		į	Designer 1	Designer ₂	Designer 3	Designer ₄		
Design manager	<i>Best</i> : Designer ₂	\overline{P}_B	[3, 5]	[1, 1]	[1, 3]	[7, 9]		
	Worst: Designer 4	\overline{P}_W	[1, 3]	[7, 9]	[1, 3]	[1, 1]		

Based on group interval BWM, each designer rates the preferences of the engines' criteria regarding to the best and worst criteria. The corresponding interval preferences of the linguistic terms considered by the designers are shown in Table III.6. Table III.6 presents the interval best-to-others and others-to-worst vectors of preference degrees of the engines criteria.

Table III.6. Preference degrees of the engines criteria.

		Interval	preference	es obtained	from lingu	istic terms							
DMs	Best criterion	Worst criterion		CUP	CUT	FCG	IDM	RPC	RNF	JCI	CME	THE	NOE
Designer 1 JCI	ICI	тне	$ar{Q}^{[1]}_{\scriptscriptstyle B}$	[3, 5]	[5, 7]	[1, 3]	[1, 3]	[5, 7]	[5, 7]	[1,1]	[5, 7]	[7, 9]	[1, 3]
	JC1	ITE	$ar{Q}_W^{[1]}$	[1, 3]	[5, 7]	[1, 3]	[3, 5]	[5, 7]	[1, 3]	[7, 9]	[1, 3]	[1, 1]	[5, 7]
	DNE	IDM	$\bar{Q}_{\scriptscriptstyle B}^{[2]}$	[5, 7]	[3, 5]	[5, 7]	[7, 9]	[1, 3]	[1, 1]	[1, 3]	[5, 7]	[5, 7]	[1, 3]
Designer 2	KINF		$ar{Q}^{[2]}_W$	[3, 5]	[5, 7]	[1, 3]	[1, 1]	[5, 7]	[7, 9]	[5, 7]	[1, 3]	[1, 3]	[5, 7]
Designer	ICI	IDM	$\bar{Q}_{B}^{[3]}$	[1, 3]	[5, 7]	[3, 5]	[7, 9]	[5, 7]	[1, 3]	[1, 1]	[5, 7]	[3, 5]	[1, 3]
Designer ₃ JC	JCI	IDW	$ar{Q}^{[3]}_W$	[3, 5]	[5, 7]	[1, 3]	[1, 1]	[3, 5]	[3, 5]	[7, 9]	[1, 3]	[5, 7]	[5, 7]
Designer 4	DDC	CUT	$\bar{Q}^{[4]}_{\scriptscriptstyle B}$	[3, 5]	[7, 9]	[3, 5]	[3, 5]	[1, 1]	[5, 7]	[1, 3]	[3, 5]	[5, 7]	[1, 3]
	KPU	CUT	$ar{Q}^{[4]}_W$	[3, 5]	[1, 1]	[3, 5]	[1, 3]	[7, 9]	[5, 7]	[5, 7]	[3, 5]	[1, 3]	[5, 7]

For the subordinate parts and the interval MULTIMOORA-Borda, the best alternative is obtained based on Eqs. (III.15), (III.22), (III.24), and (III.19), shown as follows: $A_{1-RS}^* = A_{1-MF}^* = A_{1-MULTIMOORA-Borda}^* = E_5$ and $A_{1-RP}^* = E_4$.

The normalized interval ratings of the engines for this problem are calculated using Eq. (III.13). Based on Eq. (III.17), the deviations of the weighted normalized interval ratings are generated. Table III.7 lists the assessment values for the interval ratio system, the interval reference point model, the interval full multiplicative model, and the interval MULTIMOORA-Borda model computed based on Eqs. (III.14), (III.18), (III.20), and (III.23), respectively, as well as the rankings for the interval ratio system, the interval reference point model, the interval full multiplicative model, and interval MULTIMOORA-Borda.

Table III.7. Assessment values and rankings of the interval MULTIMOORA.

Eng. ID	. ID I-RS		I-RP		I-MF		I-MULTIMOORA-Borda ¹	
	$\overline{\mathcal{Y}}_i$	$r(\overline{y}_i)$	\bar{z}_i	$r(\overline{z}_i)$	$ar{u}$	$r(\overline{u}_i)$	$\overline{IMB_i}$	$r\left(\overline{IMB_i}\right)$
E_1	[-0.205, 0.207]	6	[0.000, 0.047]	6	[0.485, 1.549]	6	[-0.284, 0.044]	6
E_2	[-0.205, 0.210]	5	[0.000, 0.035]	5	[0.525, 1.554]	5	[-0.191, 0.088]	5
E_3	[-0.187, 0.216]	4	[0.000, 0.023]	3	[0.549, 1.648]	2	[-0.075, 0.192]	3
E_4	[-0.200, 0.253]	2	[0.000, 0.010]	1	[0.603, 1.592]	3	[-0.061, 0.218]	2
E_5	[-0.184, 0.250]	1	[0.000, 0.013]	2	[0.592, 1.694]	1	[-0.067, 0.300]	1
E_6	[-0.203, 0.243]	3	[0.000, 0.023]	3	[0.579, 1.572]	4	[-0.114, 0.164]	4

¹I-MULTIMOORA-Borda: Interval MULTIMOORA-Borda.

Chapter IV

RF_S-MpMcDM_FRFP methodology: **Biomedical application**

In this chapter, we present the RF_S-MpMcDM_FRFP methodology. The methodology is based on fuzzy-rationality-based fuzzy prospect theory and trapezoidal fuzzy numbers and modeled by the BWM-MULTIMOORA approach.

In the real-world decision-making problems, the judgements of experts are often expressed as uncertain values. Such uncertain judgements may also be associated with some degrees of risk posed by irrationality of experts. Fuzzy prospect theory is a significant approach to tackle risky uncertain problems. Accordingly, we suggest a fuzzy-rationality-based fuzzy prospect theory based on a novel fuzzy distance measure and utilize the theory to develop a risky fuzzy multi-person best-worst method. In the proposed methodology, a director manages an expert panel. The risky fuzzy preferences of the experts are utilized in a fuzzy MULTIMOORA model with target-based normalization technique to generate the final results.

We discuss a practical case on biomedical problem regarding spinal prosthesis material selection. The proposed methodology is beneficial for the problem as multiple biomedical engineers are often involved in sensitive orthopedic treatments and their judgments are subjected to risk of irrationality entailed from work pressures.

1. Introduction

In real-life, rationality of human decisions is imperfect. Thus, considering the risk of the degree of rationality is an interesting topic in the decision-making process. Risky Fuzzy Multi-person Multi-criteria Decision-Making supported on Fuzzy Prospect theory (RF_MpMcDM_FP) deals with the risk of rationality degree of decision-makers' judgments in uncertain circumstances. The RF_MpMcDM_FP approaches have grabbed attention of researchers in the recent years[4], [132]–[134]. Liu and Zhang[132] introduced a hesitant fuzzy MABAC method on the basis of prospect theory. Liu et al.[133] developed a trapezoidal intuitionistic fuzzy Choquet integral operator to obtain the overall prospect value related to each alternatives. Huang et al.[134] put forward an extension of quality function deployment using hesitant fuzzy linguistic sets integrated with prospect theory. Dai et al.[4] presented a dynamic fuzzy MULTIMOORA based on prospect theory.

As philosophers of science and methodologists criticize, human knowledge is subjected to cognitive imperfections leads to "classical and bounded rationalities" evaluated in the "classical paradigm of decision-choice". However, according to "the nature of human cognitive process" which needs flexibility, a "paradigm shift" is inevitable. The new paradigm is regarded as the "fuzzy paradigm of decision-choice". "Fuzzy rationality" is evaluated in this new paradigm[34].

In classical paradigm of decision-choice theory, rationality of decision has been supposed as crisp values[34]. In reallife, decisions are associated with uncertainty and there is doubt on rationality of these decisions. Knowledge vagueness places us under conditions of fuzzy rationality[135]. Fuzzy rationality of decisions is an important issue from the perspective of realistic problems such as Political Economy[136].

Psychologically speaking, human decision always include some degrees of "irrationality" based on Sigmund Freud's opinion. Even if a seller seemingly has a perfect solution for a buyer that fits all their needs, buyers often make irrational decisions and choose alternatives[137]. In real life, the decisions of humans are often associated with uncertainty[138]. The risk entailed from the irrationality of decisions can also have uncertain nature. In healthcare service, biomedical engineers may make irrational decisions because of their critical work and job pressure and their irrational decisions can have some degrees of uncertainty because of the uncertain nature of biomedical problems.

In the context of RF-MpMcDM-FP, irrationality of fuzzy decisions is supposed to be a crisp value (irrationality is considered through fuzzy prospect theory). However, the question is "*how to derive fuzzy irrationality of fuzzy decisions* (*in the context of RF-MpMcDM-FP*)?" Fuzzy irrationality fits to real-life problems and also leads to a computation without degeneration of fuzzy data. Thus, a research gap exist in the context to tackle fuzzy irrationality needed for realistic risky fuzzy problems with critical situation.

To plug the gap related to "fuzzy rationality in the RF-MpMcDM-FP context", we introduce the fuzzy-rationalitybased fuzzy prospect theory in the context of RF-MpMcDM-FP. In the proposed theory, a fuzzy value function can represent the fuzzy rationality of experts' judgments. Thus, the fuzzy-distance-based prospect theory structure allows us to Risky Fuzzy Supervised Multi-person Multi-criteria Decision-Making methodology supported on Fuzzy-Rationality-based Fuzzy Prospect Theory (RF_S-MpMcDM_FRFP) applying the BWM-MULTIMOORA approach. The suggested methodology is derived by introducing the following decision-making models:

Risky fuzzy BWM-based model supported on the FRFP for weighting process of experts and criteria: We
introduce a fuzzy prospect theory supported on a novel fuzzy distance measure of trapezoidal fuzzy numbers
applicable for risky fuzzy decision-making models. In the proposed prospect theory, a fuzzy-distance-based
value function has been defined. As we have used the fuzzy distance, our development ensures fully uncertain

computation without degeneration of fuzzy information. We employ the proposed fuzzy-distance-based prospect theory to develop a risky fuzzy multi-person best–worst method. In this model, a director determines the relative preference (i.e., competence) of the members of an expert panel. The director and the expert panel both participate in obtaining preferences of problem criteria. A self-reliance coefficient is considered to combine the preferences of the director and expert panel. All preferences are in the form of risky fuzzy values based on the suggested fuzzy-distance-based prospect theory. The risky fuzzy multi-person best–worst method results in subjective weights of problem criteria. We combine the subjective weights with the objective weights obtained supported on a fuzzy TOPSIS-inspired method to reach the integrated weights of problem criteria. The proposed fuzzy distance measure is employed in the fuzzy TOPSIS-inspired method.

2) Fuzzy MULTIMOORA-based model for ranking process of alternatives: To obtain the subordinate rankings of alternatives in the risky fuzzy decision-making model, we utilize the fuzzy MULTIMOORA method in which the aforementioned integrated weights of problem criteria are exploited. In the formulation of the fuzzy MULTIMOORA, we use a target-based normalization technique employing the proposed fuzzy distance measure. Eventually, to generate the final rankings of alternatives, the subordinate outcomes of the fuzzy MULTIMOORA are aggregated based on a fuzzy distance matrix. The fuzzy distance matrix outweighs the dominance theory used for rank aggregation in the traditional MULTIMOORA. In the formulation of the fuzzy distance matrix, we have also employed the introduced fuzzy distance measure.

To perform a practical analysis of the proposed methodology, we assess a real-life problem in the area of biomedical engineering application. The purpose of problem is finding appropriate biomaterial for a fixator used in treatment of spinal disorders by considering risk of irrational decisions of biomedical engineers. Besides, we implement sensitivity analyses on the consistency of subjective weights of problem criteria and the variation of the final rankings. The results of the introduced model are compared with two other decision-making methods.

The remander of this chapter is devised as follows: The proposed RF_S-MpMcDM_FRFP methodology is introduced in Section 2. We evaluate a practical problem of surgical implants in Section 3.

2. Theory of RF_S-MpMcDM_FRFP methodology

In this section, we present the RF_S-MpMcDM_FRFP methodology as illustrated in Fig. IV.1. The methodology is derived in two phases: (1) we develop the risky fuzzy weighting process of criteria comprising subjective, objective, and integrated approaches. Subjective weights of problem criteria are obtained using the proposed risky fuzzy multi-person best–worst method. The objective weights are computed using a TOPSIS-inspired method which are then combined with the subjective weights (see Section 2.1); and (2) we utilize a fuzzy ranking process of alternatives including the three subordinate parts of the fuzzy MULTMOORA model and the ranking aggregation technique based on fuzzy distance matrix (see Section 2.2).

The risky fuzzy decision-making problem based on trapezoidal fuzzy numbers has a decision matrix represented as:

$$\tilde{\mathbf{X}} = \left[\tilde{x}_{ij}\right]_{m \times n}, \tag{IV.1}$$

where $\tilde{x}_{ij} = (x_{ij,1}, x_{ij,2}, x_{ij,3}, x_{ij,4})$ is the fuzzy rating of alternative A_i , i = 1, ..., m, on criterion C_j , j = 1, ..., n. For each criterion, the fuzzy target \tilde{t}_i is defined as:

$$\tilde{t}_{j} = \left\{ \max_{i} \tilde{x}_{ij}, \text{ if } j \in I; \quad \min_{i} \tilde{x}_{ij}, \text{ if } j \in J; \quad \tilde{g}_{j}, \text{ if } j \in K \right\},$$
(IV.2)

where $_{i}$ and J are related to beneficial and cost criteria, respectively. The maximum and minimum of fuzzy ratings on beneficial and cost criteria can be determined utilizing the preference matrix introduced in Section I.2.1. K represents the criteria for which a given value (i.e., \tilde{g}_{j}) is preferred. The value of \tilde{g}_{j} can be specified by decision-makers or obtained through experiments or extracted from handbooks based on the nature of practical problems.

Fuzzy decision matrix $\hat{\mathbf{X}}$ should be normalized before exploitation in the decision-making model as the ratings of alternatives on criteria are often not comparable due to various dimensions[16]. We propose a fuzzy target-based normalization technique, as follows:

$$\tilde{f}_{ij} = e^{-\tilde{\rho}_{ij}}, \qquad (\text{IV.3})$$

where the exponent value is defined as:

$$\tilde{\rho}_{ij} = \tilde{d}\left(\tilde{x}_{ij}, \tilde{t}_j\right) / \max_i \, \overline{d}\left(\tilde{x}_{ij}, \tilde{t}_j\right). \tag{IV.4}$$

In Eq. (IV.4), \tilde{x}_{ij} represents the fuzzy rating (i.e., each array of the fuzzy decision matrix $\tilde{\mathbf{X}}$) and \tilde{t}_j is the fuzzy target for each criterion defined in Eq. (IV.2). The fuzzy distance in the numerator of Eq. (IV.4) is obtained based on Eq. (I.17) and the crisp distance in the denominator is the defuzzified value of the fuzzy distance which could be computed by Eq. (I.13).

Risky Fuzzy Weighting Process of Criteria
 Subjective Risky Fuzzy Weights of Criteria: Step 1 (Prioritize members of the expert panel by the director – using the risky fuzzy multi-person best–worst method)
Step 2 (Prioritize criteria by the director and expert panel – using the risky fuzzy multi-person best–worst method)
2. Objective Fuzzy Weights of Criteria: Generate objective fuzzy weights of criteria employing the proposed fuzzy distance measure
3. Integrated Risky Fuzzy Weights of Criteria: Combine the subjective and objective weights to obtain the integrated risky fuzzy weights of criteria
Fuzzy Ranking Process of Alternatives
1. Ranking of Alternatives Based on Fuzzy Ratio System
2. Ranking of Alternatives Based on Fuzzy Reference Point Approach
3. Ranking of Alternatives Based on Fuzzy Full Multiplicative Form
4. Final ranking of Alternatives Based on Fuzzy Distance Matrix
Fig. IV.1. Flowchart of the RF S-MpMcDM FRFP methodology.

2.1. BWM-based model for weighting process

In this study, we use an integrative approach for weighting of criteria and experts' competence. First, the subjective weights of criteria are computed based on the comments of a director and an expert panel by utilizing the risky fuzzy multi-person best–worst method. Expertise level of the members of the panel are also determined in the proposed best–worst method by the director. Second, the objective weights are calculated exploiting the fuzzy TOPSIS-inspired method. Third, the aforementioned subjective and objective weights are consolidated to be further used in the fuzzy ranking process of alternatives.

Subjective risky fuzzy weights of criteria

We propose a risky fuzzy multi-person best-worst method to obtain subjective weights of problem criteria. The risk of decisions are supported on the FRFP theory introduced in Section I.3.2. Best-worst method has exclusive privileges for generating subjective weights in a decision-making problem: (i) it requires less pairwise evaluation in comparison to AHP to obtain criteria weights; (ii) best-worst method is more consistent than AHP method; (iii) the consistency percentage of BWM also denotes the confidence level; and (iv) in contrast with AHP, best-worst method only employs integers to consider preferences.

The supervised multi-person decision-making structure has two levels of decision-making as illustrated in Fig. IV.2. To obtain the subjective weights of criteria using the model, the following procedure is considered (the judgments are made in the form of linguistic terms then converted into trapezoidal fuzzy numbers):



Fig. IV.2. Supervised multi-person structure in the RF_S-MpMcDM_FRFP methodology (Ω denotes self-reliance coefficient considered in Model (IV.16)).

Consider a "director" and a "expert panel" {Expert₁, ..., Expert_k, ..., Expert_t}. In the risky fuzzy multi-person bestworst method, the director assesses the priority of the expert panel, and both the director and expert panel specify the priority of the criteria { C_1 , ..., C_j , ..., C_n }.

Step 1 (Prioritize members of the expert panel by the director – using the risky fuzzy multi-person best–worst method): The director ascertains the best and worst experts and evaluates the risky fuzzy preference degrees of the expert panel regarding the best and worst experts. The director determines the fuzzy preference degree $\tilde{h}_{Bk} = (h_{Bk,1}, h_{Bk,2}, h_{Bk,3}, h_{Bk,4})$ of the best expert *B* over expert *k* in the panel. \tilde{h}_{Bk} is a trapezoidal fuzzy value ranging from (1, 1, 1, 1) to (9, 9, 9, 9). Then, based on the value function of the FRFP theory proposed in Section I.3.2, i.e., Eq. (I.23), the risky fuzzy best-toothers experts' competence is calculated as:

$$\tilde{h}_{Bk}^{r} = \exp\left(-\delta\left[\tilde{d}^{*}\left(\tilde{h}_{Bk}, \tilde{h}_{BB}\right)\right]^{\beta}\right). \tag{IV.5}$$

In Eq. (IV.5), we use the normalized value of fuzzy distance, i.e., \tilde{d}^* , to keep the value of \tilde{h}_{Bk}^r in the same range of \tilde{h}_{Bk} , i.e., (1, 1, 1, 1) to (9, 9, 9, 9). Also, we employ the exponential function to avoid zero values. The normalized values of fuzzy distance are computed as $\tilde{d}^* = (d_1 / \kappa, d_2 / \kappa, d_3 / \kappa, d_4 / \kappa)$ where d_1 , d_2 , d_3 , and d_4 are the base values of the fuzzy distance measure, and $\kappa = \sqrt{\frac{1}{4}\sum_{j=1}^n (d_1^2 + d_2^2 + d_3^2 + d_4^2)}$. To generate results consistent with the empirical data, we set $\theta = 225$ and $\beta = 0.88$ [4]. $\tilde{H}_B^r = (\tilde{h}_{B1}^r, \dots, \tilde{h}_{Bk}^r, \dots, \tilde{h}_{Bt}^r)$ denotes the risky fuzzy best-to-others vector of the

we set $\theta = 223$ and $\beta = 0.88[4]$. $H_B = (h_{B1}, ..., h_{Bk}, ..., h_{Bl})$ denotes the risky fuzzy best-to-others vector of the preference degrees of the expert panel. The director determines the fuzzy preference degree $\tilde{h}_{kW} = (h_{kW,1}, h_{kW,2}, h_{kW,3}, h_{kW,4})$ of expert k over the worst expert W. Then, based on the value function of the fuzzy prospect theory introduced in Eq. (I.23), the risky fuzzy others-to-worst experts' competence is calculated as:

$$\tilde{h}_{kW}^{r} = \exp\left(\left[\tilde{d}^{*}\left(\tilde{h}_{kW}, \tilde{h}_{WW}\right)\right]^{\gamma}\right).$$
(IV.6)

The reason of employing the normalized values of fuzzy distance and exponential function is similar to the explanation aforementioned in Step 3.1 for obtaining \tilde{h}_{Bk}^r . To achieve consistent results with the empirical data, we set $\gamma = 0.88$ [4]. $\tilde{H}_W^r = (\tilde{h}_{1W}^r, \dots, \tilde{h}_{kW}^r, \dots, \tilde{h}_{rW}^r)^T$ indicates the risky fuzzy others-to-worst vector of the preference degrees of the members of the expert panel. The optimal weights vector $(\lambda_1^*, \lambda_2^*, \dots, \lambda_t^*)^T$ for the expert panel is computed utilizing a min-max model as follows:

$$\min \max_{k} \left\{ \left| \lambda_{B} - \tilde{h}_{Bk}^{r} \lambda_{k} \right|, \left| \lambda_{k} - \tilde{h}_{kW}^{r} \lambda_{W} \right| \right\},$$

s.t.:
$$\sum_{k=1}^{t} \lambda_{k} = 1, \quad \lambda_{k} \ge 0, \text{ for all } k.$$
 (IV.7)

Model (IV.7) could be linearized as follows:

min ξ ,

s.t.:
$$\left|\lambda_{B} - \tilde{h}_{Bk}^{r} \lambda_{k}\right| \leq \xi$$
, for all k ,
 $\left|\lambda_{k} - \tilde{h}_{kW}^{r} \lambda_{W}\right| \leq \xi$, for all k ,
 $\sum_{k=1}^{t} \lambda_{k} = 1, \ \lambda_{k} \geq 0$, for all k .
(IV.8)

In Model (IV.8), the symbol \lesssim shows 'almost lesser than' constraint. An equivalent form of Model (IV.8) could be

formulated taking account of the absolute values definitions of $\left|\lambda_{B} - \tilde{h}_{Bk}^{r}\lambda_{k}\right|$ and $\left|\lambda_{k} - \tilde{h}_{kW}^{r}\lambda_{W}\right|$, as follows:

min ξ ,

s.t.:
$$\lambda_B - \xi \leq \tilde{h}_{Bk}^r \lambda_k$$
, for all k ,
 $\lambda_B + \xi \geq \tilde{h}_{Bk}^r \lambda_k$, for all k ,
 $\lambda_k - \xi \leq \tilde{h}_{kW}^r \lambda_W$, for all k ,
 $\lambda_k + \xi \geq \tilde{h}_{kW}^r \lambda_W$, for all k ,
 $\sum_{k=1}^r \lambda_k = 1, \lambda_k \geq 0$, for all k .
(IV.9)

By considering the crisp equivalents of fuzzy constraints in Model (IV.9) besides the α -cut (with $\alpha = 0.5$) of \tilde{h}_{Bk}^{r} and \tilde{h}_{kW}^{r} , the model is transformed as follows:

min
$$\xi$$
,

s.t.:
$$\lambda_{B} - \xi \leq \left[\frac{\left(h_{Bk,1}^{r} + h_{Bk,2}^{r}\right)}{2} + \frac{\left(h_{Bk,3}^{r} + h_{Bk,4}^{r} - h_{Bk,1}^{r} - h_{Bk,2}^{r}\right)}{2} \sigma \right] \lambda_{k}, \text{ for all } k,$$

$$\lambda_{B} + \xi \geq \left[\frac{\left(h_{Bk,1}^{r} + h_{Bk,2}^{r}\right)}{2} - \frac{\left(h_{Bk,3}^{r} + h_{Bk,4}^{r} - h_{Bk,1}^{r} - h_{Bk,2}^{r}\right)}{2} \sigma \right] \lambda_{k}, \text{ for all } k,$$

$$\lambda_{k} - \xi \leq \left[\frac{\left(h_{kW,1}^{r} + h_{KW,2}^{r}\right)}{2} + \frac{\left(h_{kW,3}^{r} + h_{kW,4}^{r} - h_{kW,1}^{r} - h_{kW,2}^{r}\right)}{2} \sigma \right] \lambda_{W}, \text{ for all } k,$$

$$\lambda_{k} + \xi \geq \left[\frac{\left(h_{kW,1}^{r} + h_{kW,2}^{r}\right)}{2} - \frac{\left(h_{kW,3}^{r} + h_{kW,4}^{r} - h_{kW,1}^{r} - h_{kW,2}^{r}\right)}{2} \sigma \right] \lambda_{W}, \text{ for all } k,$$

$$\sum_{k=1}^{r} \lambda_{k} = 1, \ \lambda_{k} \geq 0, \qquad \text{ for all } k,$$

where σ ($0 \le \sigma \le 1$) is a possibility level assigned by the decision-makers. As Model (IV.10) has a linear form, it results in a unique optimal weight vector $(\lambda_1^*, \lambda_2^*, ..., \lambda_t^*)^T$ and an objective value ξ^* for any given possibility level σ . ξ^* represents the consistency level of the fuzzy risky preferences. If $\varepsilon^* = 0$ for all σ , we have $\lambda_B / \lambda_j = \tilde{h}_{Bk}^r$ and $\lambda_j / \lambda_W = \tilde{h}_{kW}^r$, and correspondingly $\tilde{h}_{Bk}^r \times \tilde{h}_{kW}^r = \tilde{h}_{BW}^r$ for all k. Hence, the fuzzy risky preferences vectors \tilde{H}_B^r and \tilde{H}_W^r are fully consistent. Step 2 (Prioritize criteria by the director and expert panel – using the risky fuzzy multi-person best-worst method): The director determines the best and worst criteria and considers the fuzzy preference degree $\tilde{p}_{Bj} = (p_{Bj,1}, p_{Bj,2}, p_{Bj,3}, p_{Bj,4})$ of the best criterion *B* over criterion j. Then, according to the fuzzy-distance-based prospect theory, the risky fuzzy form of \tilde{p}_{Bj} is obtained as:

$$\tilde{p}_{Bj}^{r} = \exp\left(-\theta \left[\tilde{d}^{*}\left(\tilde{p}_{Bj}, \tilde{p}_{BB}\right)\right]^{\beta}\right).$$
(IV.11)

The risky fuzzy best-to-others vector of the preference degrees of criteria determined by the director is $\tilde{P}_B^r = \left(\tilde{p}_{B1}^r, \dots, \tilde{p}_{Bj}^r, \dots, \tilde{p}_{Bn}^r\right)^T$. In analogous, the director allocates the fuzzy preference degree $\tilde{p}_{jW} = \left(p_{jW,1}, p_{jW,2}, p_{jW,3}, p_{jW,4}\right)$ of criterion C_j over the worst criterion W. The risky fuzzy form of \tilde{p}_{jW} is calculated as:

$$\tilde{p}_{jW}^{r} = \exp\left(\left[\tilde{d}^{*}\left(\tilde{p}_{jW}, \tilde{p}_{WW}\right)\right]^{r}\right).$$
(IV.12)

The risky fuzzy others-to-worst vector of the preference degrees of criteria determined by the director is obtained as $\tilde{P}_{W}^{r} = \left(\tilde{p}_{1W}^{r}, \dots, \tilde{p}_{jW}^{r}, \dots, \tilde{p}_{nW}^{r}\right)^{T}$ Each expert determines the best and worst criteria and considers the fuzzy preference degree $\tilde{q}_{Bj}^{[k]} = \left(q_{Bj,1}^{[k]}, q_{Bj,2}^{[k]}, q_{Bj,3}^{[k]}, q_{Bj,4}^{[k]}\right)$ of the best criterion *B* over each criterion *C*_j. Then, according to the fuzzy-distance-based
prospect theory, the risky fuzzy form of $\tilde{q}_{Bj}^{[k]}$ is computed as:

$$\tilde{q}_{Bj}^{[k],r} = \exp\left(-\theta \left[\tilde{d}^* \left(\tilde{q}_{Bj}^{[k]}, \tilde{q}_{BB}^{[k]}\right)\right]^\beta\right).$$
(IV.13)

The risky fuzzy best-to-others vector of the preference degrees of criteria evaluated by expert k is obtained as $\tilde{Q}_{B}^{k]r} = (\tilde{q}_{B}^{k]r}, \dots, \tilde{q}_{B}^{k]r}, \dots, \tilde{q}_{B}^{k}, \dots, \tilde{q}_{B}^{k},$

$$\tilde{q}_{jW}^{[k],r} = \exp\left(\left[\tilde{d}^*\left(\tilde{q}_{jW}^{[k]}, \tilde{q}_{WW}^{[k]}\right)\right]^{\gamma}\right).$$
(IV.14)

The risky fuzzy others-to-worst vector of the preference degrees of criteria specified by expert k is obtained as $\tilde{Q}_{W}^{[k],r} = (\tilde{q}_{1W}^{[k],r}, ..., \tilde{q}_{jW}^{[k],r}, ..., \tilde{q}_{nW}^{[k],r})^{T}$. The optimal weight vector of criteria $(w_{1}^{*}, w_{2}^{*}, ..., w_{n}^{*})^{T}$ is generated using the following programming model:

$$\min\left\{\Omega\max_{j}\left\{\left|w_{B}-\tilde{p}_{Bj}^{r}w_{j}\right|,\left|w_{j}-\tilde{p}_{jW}^{r}w_{W}\right|\right\}+\left(1-\Omega\right)\sum_{k=1}^{l}\lambda_{k}^{*}\max_{j}\left\{\left|w_{B}-\tilde{q}_{Bj}^{[k],r}w_{j}\right|,\left|w_{j}-\tilde{q}_{jW}^{[k],r}w_{W}\right|\right\}\right\},$$
s.t.: $\sum_{j=1}^{n}w_{j}=1, \ w_{j}\geq0, \ \text{for all } j,$
(IV.15)

where λ_k^* is the optimal weight of expert k calculated based on Model (IV.10). Model (IV.15) combines the individual and multi-person decisions of the director and expert panel. The director makes a tradeoff between his decision

and those of the expert panel by considering a self-reliance coefficient Ω ($0 \le \Omega \le 1$). $\Omega=0$ means a democracy situation where the director does not interfere in criteria weighting (i.e., s/he only judges the priorities of the expert panel), while $\Omega=1$ indicates an autocracy situation where the criteria weighting of the expert panel are neglected (*i.e.*, the director only evaluates the importance of criteria). Model (IV.15) can be reformulated as:

$$\min \Psi = \Omega \varepsilon + (1 - \Omega) \sum_{k=1}^{i} \lambda_{k}^{*} \varepsilon_{k},$$

s.t.: $|w_{B} - \tilde{p}_{Bj}^{r} w_{j}| \leq \varepsilon,$ for all $j,$
 $|w_{j} - \tilde{p}_{jW}^{r} w_{W}| \leq \varepsilon,$ for all $j,$
 $|w_{B} - \tilde{q}_{Bj}^{(k),r} w_{j}| \leq \varepsilon_{k},$ for all j
 $|w_{j} - \tilde{q}_{jW}^{(k),r} w_{W}| \leq \varepsilon_{k},$ for all j } for all $k,$
 $\sum_{j=1}^{n} w_{j} = 1, w_{j} \geq 0,$ for all $j.$
(IV.16)

Model (IV.16) could be transformed to an equivalent crisp model similar to the aforementioned procedure of Model (IV.10). As the resultant crisp formulation is a linear programming model, its solution, for any given possibility level σ ,

is in the form of an unique optimal weight vector $(w_1^*, w_2^*, ..., w_n^*)^T$, ε^* , and $(\varepsilon_1^*, \varepsilon_1^*, ..., \varepsilon_t^*)^T$. The objective function

value Ψ^* implies the consistency of the risky fuzzy preference degrees of the director and expert panel. $\Psi=0$ indicates the full consistency. Nevertheless, the preferences are subjected to variance because of the probable irrationality of the director and expert panel or other reasons. Thus, the full consistency may be not feasible.

• Objective weights of criteria

We employ an objective weighting model supported on the theory of TOPSIS method. The proposed approach exploits the fuzzy distance measure, i.e., Eq. (I.17), and also utilizes the normalization technique, i.e., Eq. (IV.3). To derive the fuzzy objective weights, the optimistic and pessimistic values of the fuzzy normalized ratings of each alternative are first computed as:

$$\tilde{f}_{i}^{+} = \max_{j} \tilde{f}_{ij}, \quad \tilde{f}_{i}^{-} = \min_{j} \tilde{f}_{ij}.$$
 (IV.17)

The overall distances between the fuzzy target-based normalized ratings and the optimistic/pessimistic values are obtained:

$$\tilde{D}_{j}^{+} = \sqrt{\sum_{i=1}^{m} \left(\tilde{d} \left(\tilde{f}_{i}^{+}, \tilde{f}_{ij} \right) \right)^{2}}, \quad \tilde{D}_{j}^{-} = \sqrt{\sum_{i=1}^{m} \left(\tilde{d} \left(\tilde{f}_{i}^{-}, \tilde{f}_{ij} \right) \right)^{2}}.$$
(IV.18)

The objective fuzzy weights are determined as follows:

$$\tilde{w}_{j}^{o} = \tilde{D}_{j}^{-} \oslash \left(\tilde{D}_{j}^{+} \oplus \tilde{D}_{j}^{-} \right).$$
(IV.19)

• Integrated risky fuzzy weights of criteria

The combination of the subjective weight w_j^* obtained by Model (IV.16) and the objective fuzzy weight \tilde{w}_j^o determined by Eq. (IV.19) generates the integrated fuzzy weight \tilde{s}_j :

$$\tilde{s}_{j} = \left(w_{j}^{*} \otimes \tilde{w}_{j}^{o}\right) \oslash \sum_{j=1}^{n} \left(w_{j}^{*} \otimes \tilde{w}_{j}^{o}\right)$$
(IV.20)

2.2. MULTIMOORA-based model for ranking process

In this part, we obtain the three ranking lists based on subordinate models of fuzzy MULTIMOORA, *i.e.*, the fuzzy ratio system, fuzzy reference point approach, and fuzzy full multiplicative form. The results of each method are then combined into the final ranking using a fuzzy distance matrix.

MULTIMOORA is an integrative decision-making approach based on three subordinate parts: ratio system, reference point approach, and full multiplicative form. Ratio system, as a fully compensatory approach, takes account of independent criteria, whereas full multiplicative form considers dependent criteria and has an incompletely compensatory algorithm. Reference point approach is a non-compensatory technique tends to provide a conservative solution. In reference point approach, imperfections of ratings of an alternative on one criterion is not compensated by the satisfactory performance on another criterion. Thus, by exploiting the advantages of three approaches, MULTIMOORA can be a key tool in the problems where there are independent and dependent criteria while a conservative result is also important[16].

• Ranking of alternatives based on fuzzy ratio system

Using arithmetic operations of trapezoidal fuzzy numbers presented in Eq. (I.14) and considering the integrated fuzzy weights obtained by Eq. (IV.20) as well as the normalized fuzzy ratings by Eq. (IV.3), the assessment value \tilde{Y}_i is generated as follows:

$$\tilde{Y}_i = \sum_{j=1}^{g} \left(\tilde{s}_j \otimes \tilde{f}_{ij} \right)$$
(IV.21)

The best alternative in the fuzzy ratio system has the maximum value of \tilde{Y}_i , obtained supported on the mathematics of trapezoidal fuzzy numbers given in Section I.2.2:

$$A_{\text{F-RS}}^* = \left\{ A_{i' \mid i' = \arg\max_{i} \tilde{Y}_i} \right\}$$
(IV.22)

The ranking of alternatives based on the fuzzy ratio system is obtained by ordering the assessment values descendingly.

• Ranking of alternatives based on fuzzy reference point approach

For this approach, first, the deviation between the normalized fuzzy rating \tilde{f}_{ij} and the reference point \tilde{f}_{j}^{+} is defined as:

$$\tilde{\Delta}_{ij} = \tilde{d} \left(\tilde{f}_j^+, \, \tilde{f}_{ij} \right), \tag{IV.23}$$

where \tilde{d} denotes the fuzzy distance obtained by Eq. (I.17) and $\tilde{f}_j^+ = \max_i \tilde{f}_{ij}$. To generate the assessment value of the fuzzy target-based reference point, we have:

$$\tilde{Z}_{i} = \max_{j} \left(\tilde{s}_{j} \otimes \tilde{\Delta}_{ij} \right)$$
(IV.24)

The best alternative obt(IV.21) ained by the fuzzy reference point approach has the minimum assessment value, where:

$$A_{\text{F-RP}}^* = \left\{ A_{i'|i'=\arg\min_i \tilde{Z}_i} \right\}$$
(IV.25)

The ranking of alternatives in the fuzzy reference point approach is generated by ordering the assessment values ascendingly.

• Ranking of alternatives based on fuzzy full multiplicative form

The assessment value of the fuzzy full multiplicative form is generated as:

$$\tilde{U}_i = \prod_{j=1}^n \left(\tilde{f}_{ij}\right)^{\tilde{s}_j} \tag{IV.26}$$

As all the elements of \tilde{f}_{ij} and \tilde{s}_j are between 0 and 1, it can be shown that:

$$\left(\tilde{f}_{ij}\right)^{\tilde{s}_{j}} = \left(\left(f_{ij,1}\right)^{s_{j,4}}, \left(f_{ij,2}\right)^{s_{j,3}}, \left(f_{ij,3}\right)^{s_{j,2}}, \left(f_{ij,4}\right)^{s_{j,1}}\right).$$
(IV.27)

The optimal alternative with the fuzzy full multiplicative form is calculated as follows:

$$A_{\text{F-MF}}^* = \left\{ A_{i'|i'=\arg\max_i \operatorname{ax} \tilde{U}_i} \right\}.$$
(IV.28)

The ranking of alternatives based on the fuzzy full multiplicative is determined by ordering the assessment values descendingly.

• Final ranking of alternatives based on fuzzy distance matrix

We employ a fuzzy distance matrix supported on the proposed fuzzy distance measure to obtain the final ranking of alternatives.

In the original MULTIMOORA method, the dominance theory was employed to consolidate the subordinate ranks of alternatives; however, the theory reveals some disadvantages, such as circular reasoning [18]. The ranking aggregation technique based on fuzzy distance matrix does not have the drawbacks of the dominance theory. The fuzzy distance matrix is defined by considering the assessment values of the fuzzy ratio system, fuzzy reference point approach, and fuzzy full multiplicative form as shown in Table IV.1. In Table IV.1, $\tilde{\pi}_{il}$ equals to:

$$\tilde{\pi}_{il} = \begin{cases} 0, & i = l, \ i, l = 1, \dots, m, \\ \tilde{I}_{il}^{y} \ominus \tilde{I}_{il}^{z} \oplus \tilde{I}_{il}^{u}, & i \neq l, \ i, l = 1, \dots, m, \end{cases}$$
(IV.29)

where indices \dot{l} and l represent the position of alternatives. \tilde{I}_{il}^{y} , \tilde{I}_{il}^{z} , and \tilde{I}_{il}^{u} are respectively defined as:

$$\begin{split} \tilde{I}_{il}^{y} &= \left\{ \tilde{d} \left(\tilde{Y}_{i}, \tilde{Y}_{l} \right), \quad \text{if } \overline{Y}_{i} \geq \overline{Y}_{l}; \quad -\tilde{d} \left(\tilde{Y}_{i}, \tilde{Y}_{l} \right), \quad \text{if } \overline{Y}_{i} < \overline{Y}_{l} \right\}, \\ \tilde{I}_{il}^{z} &= \left\{ \tilde{d} \left(\tilde{Z}_{i}, \tilde{Z}_{l} \right), \quad \text{if } \overline{Z}_{i} \geq \overline{Z}_{l}; \quad -\tilde{d} \left(\tilde{Z}_{i}, \tilde{Z}_{l} \right), \quad \text{if } \overline{Z}_{i} < \overline{Z}_{l} \right\}, \\ \tilde{I}_{il}^{u} &= \left\{ \tilde{d} \left(\tilde{U}_{i}, \tilde{U}_{l} \right), \quad \text{if } \overline{U}_{i} \geq \overline{U}_{l}; \quad -\tilde{d} \left(\tilde{U}_{i}, \tilde{U}_{l} \right), \quad \text{if } \overline{U}_{i} < \overline{U}_{l} \right\}, \end{split}$$
(IV.30)

where \tilde{Y}_i , \tilde{Z}_i , and \tilde{U}_i are the assessment values of the fuzzy ratio system, reference point approach, and fuzzy full multiplicative form obtained by Eqs. (IV.21), (IV.24), and (IV.26), respectively. \overline{Y}_i , \overline{Z}_i , and \overline{U}_i are the defuzzified values

(IV.31)

of their correspondent assessment indices calculated by Eq. (I.13). \tilde{d} denotes the fuzzy distance measure introduced in Eq. Eq. (I.17). In Eq. (IV.30), the negative terms are obtained based on the following formula (for a positive trapezoidal fuzzy number $\tilde{A} = (A_1, A_2, A_3, A_4)$ and a negative real number η):

$$\tilde{A} \odot \eta = (A_4 \eta, A_3 \eta, A_2 \eta, A_1 \eta)$$

			ě	ę			
Relative	fuzzy distances b	y considering sign	Aggregated distances				
	A_1		A_l		A_m		
$A_{\rm l}$	$ ilde{\pi}_{11}$		$ ilde{\pi}_{ll}$		$ ilde{\pi}_{1m}$	$\tilde{\Gamma}_1 = \sum_{l=1}^m \tilde{\pi}_{ll}$	
÷	÷		÷		÷	:	
A_i	$ ilde{\pi}_{i1}$		$ ilde{\pi}_{il}$		$ ilde{\pi}_{\scriptscriptstyle im}$	$\tilde{\Gamma}_i = \sum_{l=1}^n \tilde{\pi}_{ll}$	
÷	÷		÷		÷	:	
A_m	$ ilde{\pi}_{m1}$		$ ilde{\pi}_{ml}$		$ ilde{\pi}_{mm}$	$\tilde{\Gamma}_{ml} = \sum_{l=1}^{m} \pi_{ml}$	

Table IV.1. The fuzzy distance matrix for obtaining final ranking.

ſ

The best alternative of the fuzzy MULTIMOORA model supported on the fuzzy distance matrix is determined as:

$$A_{\text{F-MULTIMOORA-FDM}}^{*} = \left\{ A_{i' \mid i' = \arg\max_{i} \tilde{\Gamma}_{i}} \right\}$$
(IV.32)

3. Application of RF S-MpMcDM FRFP methodology in biomedical sector

In this part, we utilize the decision-making methodology introduced in Section 2 for material selection in a practical biomedical engineering problem. First, the motivation and specifications of the practical problem are depicted. Second, the solution of the biomaterial selection problem based on the RF_S-MpMcDM_FRFP methodology is provided

The biomedical decision-making problems are sensitive cases as they deal with the health of human-beings. The physicians must decide about the situation of their patients with precision and quickness (where applicable) to present satisfactory diagnose and treatment. The precision and quickness lead to prevent fatal or permanent effects on the health of patients. To reach a final decision, the cases should often be evaluated by multiple experts. Additionally, biomedical engineers may make irrational decisions because of their critical work and job pressure. Thus, a systemic multi-person decision-making approach considering risk and consistent with the requirements of biomedical problems can be an effective tool for the experts.

This practical case is related to biomaterial selection for a spinal fixator. A biomaterial is entitled to any substance engineered to be used in biological systems for the aim of diagnostic or therapeutic applications. A number of biomaterials can often be candidate solutions for orthopedic disorders. The proposed methodology helps in finding an appropriate alternative for biomaterial selection problems.

The data of this case study was gathered based on several references of biomaterials [139]–[142] and also by consulting with the orthopedic experts deal with designing implants and surgical treatments. To collect the comments, biomedical engineers have been provided with specific forms to present their preferences about the performance of the candidate biomaterials.

Biocompatibility is a significant feature needed for biomaterials utilized for implants and prostheses[143]. Biocompatibility means the capability of a biomaterial to act suitably without showing serious defects; however, this property is rather case-sensitive which means biocompatibility in a particular usage does not guarantee suitability for another application[144]. Usually, to ensure biocompatibility, the value of the elastic modulus related to biomaterial and that of the tissue should be close as possible. This issue leads to decreasing stress concentration of biomaterial[145]. Thus, elastic modulus is supposed as a target-based criterion for selection of biomaterials. That is, the target value of elastic modulus of a biomaterial should be very close to elastic modulus of the adjacent tissue. Biomaterial properties are ordinarily given as uncertain information in the handbooks of materials science.

Generally, spinal implants are produced to increase the stability of spine and lessen the pressure on the vertebral disc [146]. A sample of spine fixator is shown in Fig. IV.3 in the supplementary file.



Fig. IV.3. The case study of the RF_S-MpMcDM_FRFP methodology: A typical spinal fixator.

We follow the phases of the proposed methodology (shown in Fig. IV.1), i.e., (i) risky fuzzy weighting process of criteria, (ii) fuzzy ranking process of alternatives, to reach the outcome of the biomaterial selection problem.

Six criteria exist in the problem: elastic modulus, density, tensile strength, fatigue strength, yield strength, and elongation. Elastic modulus and density are target-based criteria and the rest are beneficial. Table IV.2 denotes units and the fuzzy target values for the criteria. Table IV.3 represents the decision matrix of the practical case study. The decision matrix provides the fuzzy performances (i.e., fuzzy values of biomaterials properties) of eight biomaterials preselected as candidates in response to the criteria of the problem. The fuzzy ratings of the biomaterials are normalized by Eq. (IV.3).

ls

Criterion name	Unit	Туре
Elastic modulus	GPa	Target-based $\{\tilde{g}_1 = (12.8, 13.1, 14.9, 15.2)\}$
Density	g/cm ³	Target-based $\{\tilde{g}_2 = (1.9, 2.0, 2.2, 2.3)\}$
Tensile strength	MPa	Beneficial
Fatigue strength	MPa	Beneficial
Yield strength	MPa	Beneficial
Elongation	%	Beneficial

Table IV.3. Decision matrix constructed form the performance of the candidate biomaterials on their criteria.

BioM. Biomaterial name Target-based criteria Beneficial criteria

ID^*							
	Elas	stic modulus D	Density	Tensile strength	Fatigue strength	Yield strength	Elongation
M_{I}	Stainless Steel 316L (cold-worked) (177	7, 180, 206, 209) (7.2, 7.3, 8.5, 8.6)	(826, 834, 886, 894)	(298, 301, 319, 322)	(662, 669, 711, 718)	(10.9, 11.2, 12.8, 13.1)
M_2	Stainless Steel 316L (177	7, 180, 206, 209) (7	7.2, 7.3, 8.5, 8.6)	(496, 501, 533, 538)	(252, 254, 270, 272)	(318, 321, 341, 344)	(36.4, 37.2, 42.8, 43.6)

M ₃	Wrought CoNiCrMo	(208, 212, 242, 246)	(8.4, 8.6, 9.8, 10.0)	(1721, 1739, 1847, 1865)	(480, 485, 515, 520)	(1523, 1538, 1634, 1649)	(7.3, 7.4, 8.6, 8.7)
M_4	Pure Titanium	(92, 94, 107, 109)	(4.1, 4.2, 4.8, 4.9)	(528, 534, 567, 572)	(230, 233, 247, 250)	(354, 358, 380, 384)	(13.7, 14.0, 16.1, 16.4)
M_5	Ti-6Al-4V	(97, 99, 113, 115)	(4.1, 4.2, 4.8, 4.9)	(826, 834, 886, 894)	(499, 504, 536, 541)	(763, 771, 819, 827)	(9.1, 9.3, 10.7, 10.9)
M_6	Ti-6Al-7Nb	(101, 103, 117, 119)	(4.1, 4.2, 4.8, 4.9)	(1008, 1019, 1082, 1092)	(431, 436, 462, 467)	(768, 776, 824, 832)	(11.8, 12.1, 13.9, 14.2)
M_7	Co–Cr alloy (castable)	(220, 224, 256, 260)	(7.6, 7.7, 8.9, 9.0)	(629, 635, 675, 681)	(408, 412, 438, 442)	(432, 437, 464, 468)	(18.2, 18.6, 21.4, 21.8)
M_8	Co–Cr alloy (wrought)	(220, 224, 256, 260)	(8.3, 8.5, 9.7, 9.9)	(860, 869, 923, 932)	(576, 582, 618, 624)	(624, 631, 670, 676)	(18.2, 18.6, 21.4, 21.8)

*BioM. ID: Biomaterial identification code.

As illustrated in Fig. IV.4, the decision-making structure comprises the manager of biomedical engineering department as the director and four biomedical engineers as the expert panel (the manager of biomedical engineering department is only referred as "manager" in the remainder of this study). To calculate subjective weights employing the proposed risky fuzzy multi-person best–worst method, first we need to obtain risky fuzzy preferences.



Fig. IV.4. Supervised multi-person structure for the case study of the RF_S-MpMcDM_FRFP methodology.

Table IV.4 lists the terms used for assigning linguistic preferences and the related trapezoidal fuzzy preferences. The manager estimates the relative preferences (i.e., the experts' competence) of the biomedical engineers using linguistic terms. The risky fuzzy preferences are then calculated based on the linguistic terms and Eqs. (IV.5) and (IV.6). Table IV.5 gives the best-to-others and others-to-worst vectors of the biomedical engineers' risky fuzzy preferences degrees from the viewpoint of the manager.

Table IV.4. Linguistic preferences and corresponding trapezoidal fuzzy values.

Linguistic preferences	Corresponding trapezoidal fuzzy preferences
Equal importance	(1, 1, 1, 1)
Minor importance	(2, 2.75, 3.5, 4.25)
Fair importance	(4.25, 4.75, 5.25, 5.75)
Major importance	(5.75, 6.5, 7.25, 8)
Extreme importance	(9,9,9,9)

Table IV.5. Expertise degrees in the form of risky fuzzy preferences.

DM*	Best & worst biomedical Risky fuzzy preferences based on the proposed fuzzy prospect theory					
	engineers		Biomedical	Biomedical	Biomedical	Biomedical
			engineer 1	engineer 2	engineer 3	engineer 4
Managan	<i>Best</i> : Biomedical engineer ₃	\tilde{H}_{B}^{r}	(0.20, 0.20, 0.20, 0.20)	(0.20, 0.48, 0.48, 0.48)	(0.2, 1.00, 1.00, 1.00)	(0.20, 0.36, 0.36, 0.36)
Manager	Worst: Biomedical engineer	\tilde{H}_W^r	(1.00, 1.00, 1.00, 1.00)	(1.72, 1.72, 1.72, 1.86)	(2.36, 2.36, 2.36, 2.36)	(1.15, 1.15, 1.15, 1.25)

*DM: Decision-maker.

The manager and all biomedical engineers identify the best and worst criteria and then measure the preferences of the biomaterials' criteria with reference to the pre-assigned best and worst criteria. The linguistic judgments of the manager and biomedical engineers are converted into fuzzy values based on Table IV.4 and then the resultant values are transformed into risky fuzzy preference degrees employing Eqs. (IV.11) and (IV.12) besides Eqs. (IV.13) and (IV.14), respectively. Table IV.6 lists the risky fuzzy preference degrees of the criteria from the viewpoints of the manager and biomedical engineers as best-to-others and others-to-worst vectors.

	-		-	-	
DMs	Best	Worst	Risky fuzzy prefer	rences based on the proposed fuzzy prospect theory	
	criterion	criterion	C.	C ₂ C ₄	

Table IV.6. Risky fuzzy preference degrees of the criteria of the biomaterial selection problem.

DIVIS	Best	worst	risky ruzzy preferences based on the proposed ruzzy prospect theory						
	criterion	criterion		C_1 (Elastic modulus)	C ₂ (Density)	C ₃ (Tensile strength)	C ₄ (Fatigue strength)	C ₅ (Yield strength)	C ₆ (Elongation)
Manager	<i>C</i> ₁	<i>C</i> ₆	$ ilde{P}^r_B \ ilde{P}^r_W$	(1.00, 1.00, 1.00, 1.00) (2.25, 2.25, 2.25, 2.25)	(0.31, 0.31, 0.31 0.31) (1.14, 1.14, 1.14 1.24)	, (0.74, 0.74, 0.74, 0.74) , (1.44, 1.44, 1.44, 1.52)	, (0.43, 0.43, 0.43, 0.43) , (1.14, 1.14, 1.14, 1.24)	, (0.74, 0.74, 0.74, 0.74) , (1.67, 1.67, 1.67, 1.79)	(0.16, 0.16, 0.16, 0.16) (1.00, 1.00, 1.00, 1.00)
Biomedical engineer 1	C_4	<i>C</i> ₃	$ ilde{Q}^{\mathrm{I}\mathrm{],r}}_{B}$ $ ilde{Q}^{\mathrm{I}\mathrm{],r}}_{W}$	(0.32, 0.32, 0.32, 0.32) (1.15, 1.15, 1.15, 1.15)	(0.45, 0.45, 0.45 0.45) (1.15, 1.15, 1.15 1.15)	, (0.17, 0.17, 0.17, 0.17) , (1.00, 1.00, 1.00, 1.00)	, (1.00, 1.00, 1.00, 1.00) , (2.37, 2.37, 2.37, 2.37)	, (0.75, 0.75, 0.75, 0.75) , (1.72, 1.72, 1.72, 1.72)	(0.45, 0.45, 0.45, 0.45) (1.15, 1.15, 1.15, 1.15)
Biomedical engineer 2	<i>C</i> ₅	<i>C</i> ₆	$ ilde{Q}^{[2],r}_B \\ ilde{Q}^{[2],r}_W \\ ilde{Q}^{[2],r}_W$	(0.73, 0.73, 0.73, 0.73) (1.64, 1.64, 1.64, 1.64)	(0.41, 0.41, 0.41 0.41) (1.13, 1.13, 1.13 1.13)	, (0.41, 0.41, 0.41, 0.41) , (1.13, 1.13, 1.13, 1.13)	, (0.73, 0.73, 0.73, 0.73) , (1.64, 1.64, 1.64, 1.64)	, (1.00, 1.00, 1.00, 1.00) , (2.19, 2.19, 2.19, 2.19)	(0.14, 0.14, 0.14, 0.14) (1.00, 1.00, 1.00, 1.00)
Biomedical engineer 3	<i>C</i> ₄	<i>C</i> ₃	$ ilde{Q}^{3],r}_B \ ilde{Q}^{3],r}_W$	(0.39, 0.39, 0.39, 0.39) (1.12, 1.12, 1.12, 1.21)	(0.72, 0.72, 0.72 0.72) (1.57, 1.57, 1.57 1.67)	, (0.13, 0.13, 0.13, 0.13) , (1.00, 1.00, 1.00, 1.00)	, (1.00, 1.00, 1.00, 1.00) , (2.04, 2.04, 2.04, 2.04)	, (0.72, 0.72, 0.72, 0.72) , (1.57, 1.57, 1.57, 1.67)	(0.72, 0.72, 0.72, 0.72) , (1.57, 1.57, 1.57, 1.67)
Biomedical engineer 4	C_1	C_6	$ ilde{Q}^{[4],r}_B \ ilde{Q}^{[4],r}_W \ ilde{Q}^{[4],r}_W$	(1.00, 1.00, 1.00, 1.00) (2.28, 2.28, 2.28, 2.28)	(0.43, 0.43, 0.43 0.43) (1.14, 1.14, 1.14 1.14)	, (0.74, 0.74, 0.74, 0.74) , (1.68, 1.68, 1.68, 1.68)	, (0.74, 0.74, 0.74, 0.74) , (1.45, 1.45, 1.45, 1.45)	, (0.31, 0.31, 0.31, 0.31) , (1.14, 1.14, 1.14, 1.14)	(0.16, 0.16, 0.16, 0.16) (1.00, 1.00, 1.00, 1.00)

The values of deviations of the normalized fuzzy ratings are obtained by Eq. (IV.23). Table IV.7 provides the values of assessment indices for the fuzzy ratio system, fuzzy reference point, and fuzzy full multiplicative form, generated utilizing Eqs. (IV.21), (IV.24), and (IV.26), respectively. Moreover, the subordinate and two aggregate rankings lists are provided in Table IV.7. The first aggregate ranking list is obtained based on the dominance theory. To generate the second aggregate ranking list, the fuzzy distance matix introduced in Section 2.2 is used. The related assessment values are computed based on Table IV.1.

BioM.	RF-RS ¹	RF-RP ²		RF-MF ³		RF-MULTIMOORA-	RF-MULTIMOORA	A-FDM ⁵
ID	\widetilde{Y}_i $r(\widetilde{Y}_i)$	\widetilde{Z}_i	$r(\tilde{Z}_i)$	Ũi	$r(\widetilde{\boldsymbol{U}}_{i})$	Dominance ⁴	\tilde{L}_i	$r(\tilde{L}_i)$
M_{I}	(0.419, 0.427, 0.437, ₈ 0.445)	(0.156, 0.157, 0.159, 0.161)	6	(0.486, 0.485, 0.486, 0.486)	7	6	(-1.705, -1.654, - 1.632, -1.583)	6
M_2	(0.444, 0.450, 0.460, 6) (0.467)	(0.175, 0.176, 0.178, 0.180)	7	(0.482, 0.481, 0.481, 0.479)	8	6	(-1.750, -1.717, - 1.688, -1.644)	7
M_3	(0.628, 0.638, 0.649, 1) (0.660)	(0.082, 0.082, 0.083, 0.084)	3	(0.647, 0.648, 0.649, 0.650)	2	2	(1.580, 1.642, 1.660, 1.717)	1
M_4	(0.438, 0.445, 0.456, ₇ 0.464)	(0.182, 0.184, 0.186, 0.188)	8	(0.486, 0.485, 0.486, 0.486)	6	6	(-1.793, -1.755, - 1.732, -1.681)	8
M_5	(0.607, 0.620, 0.632, ₂ 0.646)	(0.078, 0.078, 0.078, 0.079, 0.080)	2	(0.659, 0.662, 0.664, 0.667)	1	1	(1.577, 1.622, 1.658, 1.696)	2
M_6	(0.576, 0.587, 0.599, ₄ 0.611)	(0.082, 0.082, 0.083, 0.083, 0.084)	4	(0.642, 0.645, 0.646, 0.649)	3	4	(1.258, 1.309, 1.335, 1.389)	3
M_7	(0.463, 0.472, 0.483, 5) (0.493)	(0.097, 0.098, 0.099, 0.100)	5	(0.521, 0.521, 0.522, 0.523)	5	5	(-0.703, -0.648, - 0.627, -0.572)	5
M_8	(0.580, 0.588, 0.599, ₃ 0.608)	(0.073, 0.074, 0.074, 0.075)	1	(0.608, 0.608, 0.609, 0.608)	4	3	(1.056, 1.106, 1.122, 1.166)	4

Table IV.7. Outcomes of the fuzzy MULTIMOORA with risky fuzzy weights supported on the fuzzy decision matrix.

¹RF-RS: Fuzzy Ratio System with Risky fuzzy weights; ²RF-RP: Fuzzy Reference Point approach with Risky fuzzy weights; ³RF-MF: Fuzzy full Multiplicative Form with Risky fuzzy weights; ⁴RF-MULTIMOORA-Dominance: Fuzzy MULTIMOORA supported on Dominance theory with Risky fuzzy weights; ⁵RF-MULTIMOORA-FDM: Fuzzy MULTIMOORA supported on Fuzzy Distance Matrix with Risky fuzzy weights.

The best alternative based on the fuzzy ratio system, fuzzy reference point, fuzzy full multiplicative form, and fuzzy target-based MULTIMOORA model with the fuzzy distance matrix are computed by Eqs. (IV.22), (IV.25), (IV.28), and (IV.32), respectively (as risky weights are used to compute subordinate and final rankings, we consider a prefix "R" to indicate the weights):

$$A_{\rm RF-RS}^* = A_{\rm RF-MULTIMOORA-FDM}^* = M_3, A_{\rm RF-RP}^* = M_8, \text{ and } A_{\rm RF-MF}^* = A_{\rm RF-MULTIMOORA-Dominance}^* = M_5.$$

Chapter V

DI_S-MpMcDM_SoESs methodology: Energy application

In this chapter, we present the DI_S-MpMcDM_SoESs methodology. The methodology is based interval numbers and modeled by the BWM-MULTIMOORA approach. The supervised multi-person structure is with subject-oriented expert segments. It is applied in an case study on renewable energy investment project.

Most socio-economic and industrial decision-making problems are interdisciplinary, needing different expertise areas for handling. Considering Subject-oriented Expert Segments (SoESs) can streamline the decision-making process for such problems. The decisions of SoESs are dynamically improved by gaining experience. Accordingly, we propose a methodology to tackle SoESs sturcture. In this methodology, a director evaluates the relative preferences of experts and problem subjects. The methodology is derived through hybrid modeling based on BWM and MULTIMOORA method.

Finally, we evaluate a real-life decision-making problem regarding investment in renewable energy sources. According to the consistency of the BWM-based models and the robustness of the MULTIMOORA-based model, the proposed methodology can be efficient in solving practical DI_S-MpMcDM_SoESs problems.

1. Introduction

Dynamic Multi-person Multi-criteria Decision-Making (D_MpMcDM) deals with the significance of time in decisionmaking area [147]. In D_MpMcDM problems, multiple experts are responsible for evaluating alternatives according to multiple criteria in a changeable situation by passing time.

The situation is mainly affected by dynamic changes in experts' relative importance and judgments. Consequently, experts' judgments progressively become more reliable based on their gained experience [13], [148]. The D_MpMcDM approaches have been utilized in various practical applications, including safety management, construction industry, and reverse logistics [149]–[151].

In real-life, many socio-economic and industrial institutes tackle interdisciplinary dynamic decision-making problems. In this regard, multiple groups of experts with special competence in each subject are often required [1]. United Nations Economic and Social Council includes several groups of experts focused on information management, accounting, and transport dynamically directed by government officials. Industrially speaking, the Operations Engineering Department of Qatar Petroleum comprises several groups (e.g., related to inspection, process, systems, and planning) that enable the corporation to fulfill dynamic customer requirements [152], [153].

In the context of D_MpMcDM, experts have not been segmented according to their competence [154]–[156]. That is, all experts are supposed to be competent in all problem subjects. The question is: "*Do all experts have competence in each subject?*" In real-life, experts may not have all-inclusive competence [157]. Thus, a research gap exists in the context to tackle the segmentation of experts needed for realistic interdisciplinary problems.

To plug the gap related to "the segmentation of experts according to experts' competence," we introduce the Subjectoriented Expert Segments (SoESs) structure in the context of (D_MpMcDM). The SoESs structure is a multi-person decision-making framework in which experts are segmented based on their competence in the problem subjects and supervised by a director. The relative importance of subjects is also assessed by the director. SoESs provide the preferences of the problem alternatives based on their subject-oriented competence.

The introduced SoESs structure can enhance the decision-making process in (D_MpMcDM) problems as: (1) the interdisciplinary nature of practical decision-making problems can be efficiently handled by considering a SoES for each subject; (2) precise evaluation is reached because each subject of the problem is assessed by the experts competent in the subject; (3) the supervision of SoESs can improve the whole decision-making process as the competence of experts is appreciated.

The SoESs structure allows us to propose a decision-making methodology for D_MpMcDM with Subject-oriented Expert Segments (DI_S-MpMcDM_SoESs methodology) applying the BWM-MULTIMOORA approach with interval computation. The suggested methodology is derived by introducing the following decision-making models:

- BWM-Based Models Supported on the SoESs Structure for Obtaining Dynamic Weights of Subjects, Experts, and Alternatives: We develop three stepwise interval BWM-based models to calculate dynamic weights for the decisionmaking problem. The first model is an interval BWM to calculate the dynamic weights of problem subjects (according to the competence of each SoES). The second model is also an interval BWM to obtain the dynamic weights of SoESs' experts. The third model is an interval multi-person BWM based on the global criterion method to calculate the dynamic weights of alternatives (based on the preferences of SoESs' experts).
- 2) *MULTIMOORA-Based Model for Obtaining Final Ranking of Alternatives*: We introduce a MULTIMOORA-based model to integrate dynamic weights of alternatives obtained for each time period. First, the integrated significance

coefficients are calculated based on the consistency of the interval multi-person BWM and information entropy method. Second, the dynamic weights of alternatives are aggregated based on the MULTIMOORA-Borda approach to reach the final ranking of alternatives.

We discuss a case study on a renewable energy investment project utilizing the proposed DI_S-MpMcDM_SoESs methodology. Segmentation of experts according to experts' competences is essential in this case study as different competences are required to examine the problem's financial, technological, environmental, and social subjects.

The remainder of this chapter is arranged as follows. Section II presents the theory of the proposed DI_S-MpMcDM_SoESs methodology. A real-life decision-making problem regarding investment in renewable energy sources is assessed in Section 3.

2. Theory of DI_S-MpMcDM_SoES methodology

This section introduces the derivation of the DI_S-MpMcDM_SoESs methodology, allowing segmentation of experts according to experts' competence. Two phases exist for developing this methodology: (1) we compute the dynamic weights of subjects, experts, and alternatives using BWM-based models in which the preferences of director and experts of SoESs are considered (see Section 2.1); (2) we calculate the final ranking of alternatives using a MULTIMOORA-based model by considering the dynamic weights of alternatives and their integrated significance coefficients (see Section 2.2). The flowchart of the proposed decision-making is illustrated in Fig. V.1.



Fig. V.1. Flowchart of the DI_S-MpMcDM_SoES methodology.

2.1. BWM-based models for weighting process

In this section, we develop three BWM-based models supported on the SoESs structure to process experts' evaluations in each time period. The sequence of the three models and the notation (*i.e.*, preferences and weights) are illustrated in Fig. V.1. All preferences are in the form of interval numbers. The purpose of the modeling is to generate dynamic weights of alternatives (obtained by the third model). The first and second models lead to dynamic weights of subjects and experts, respectively. The results of the first and second models are employed in the third model.

The SoESs structure is shown with details in Fig. V.2. The subjects of the problem are represented as $\{S_1, ..., S_j, ..., S_n\}$, j = 1, 2..., n. For each subject, an SoES is allocated: $\{SoES_1, ..., SoES_j, ..., SoES_n\}$. Each SoES includes h_j experts, represented by $\{Expert_1^j, ..., Expert_k^j, ..., Expert_{h_j}^j\}$ in which $k = 1, 2..., h_j$. The director supervises SoESs and evaluates the relative importance of the subjects.



Fig. V.2. Supervised multi-person structure in the DI_S-MpMcDM_SoES methodology.

The criteria of the problem are categorized based on the subjects. For each subject, the set of criteria is represented as $\{C_1^j, ..., C_k^j, ..., C_{h_j}^j\}$. Each expert of SoES takes decision based on his/her competence in one criterion. Thus, Expert^j_k (*i.e.*, the k th expert in the j th SoES) evaluates all alternatives (i.e., $\{A_1, ..., A_i, ..., A_m\}$, i = 1, 2, ..., m) according to his/her competence in C_k^j (*i.e.*, the k th criterion in the category of the j th subject).

We consider a set of time periods for dynamic decision-making, i.e., $\{P^0, ..., P^t, ..., P^T\}$ where t=0, 1..., T as shown in Fig. V.2. P^0 denotes the present time and the set $\{P^1, ..., P^t, ..., P^T\}$ represents the historical records. For each time period, the same SoESs structure is assigned.

To obtain dynamic weights of subjects, experts, and alternatives, a step-by-step procedure is adopted as follows:

Step 1 (Dynamic Weights of Subjects Using the Interval BWM Model): For each time period, the director assesses the priority of the problem subjects according to the best and worst items by considering best-to-others and others-to-worst

preferences. After designation of the best and worst subjects, the director assesses the interval preferences of all subjects, considering the best and worst subjects.

 $\overline{a}_{Bj}^{t} = \begin{bmatrix} a_{Bj}^{t,L}, a_{Bj}^{t,U} \end{bmatrix} \text{ is the interval preference of } S_{B}(\text{the best subject}) \text{ over } S_{j} \text{ for time period } t \cdot \overline{a}_{jW}^{t} = \begin{bmatrix} a_{jW}^{t,L}, a_{jW}^{t,U} \end{bmatrix} \text{ is the interval preference of subject } j \text{ over } S_{W} \text{ (the worst subject) for time period } t \cdot \text{ The best-to-others and others-to-worst interval preference vectors are obtained for the subjects, respectively, as } \overline{A}_{B}^{t} = (\overline{a}_{B1}^{t}, \overline{a}_{B2}^{t}, \dots, \overline{a}_{Bn}^{t}) \text{ and } \overline{A}_{W}^{t} = (\overline{a}_{1W}^{t}, \overline{a}_{2W}^{t}, \dots, \overline{a}_{nW}^{t}).$

For time period t, the weights of subjects are determined by solving the following interval BWM model:

$$\min \varepsilon^{t}, \text{ s.t.} \left| \theta^{t}_{B} - \overline{a}^{t}_{jj} \theta^{t}_{j} \right| \leq \varepsilon^{t}, \quad \forall j, t, \left| \theta^{t}_{j} - \overline{a}^{t}_{jW} \theta^{t}_{W} \right| \leq \varepsilon^{t}, \quad \forall j, t, \sum_{j=1}^{n} \theta^{t}_{j} = 1, \quad \forall t; \quad \theta^{t}_{j} \geq 0, \; \forall j, t.$$

$$(V.1)$$

Model (V.1) can be converted into the following form by considering crisp equivalents for the constraints:

$$\min \varepsilon^{t}, \quad \text{s.t.}$$

$$\theta^{t}_{B} - \varepsilon^{t} \leq \left[a^{t,L}_{Bj} + \left(a^{t,U}_{Bj} - a^{t,L}_{Bj} \right) \alpha \right] \theta^{t}_{j}, \quad \forall j, t,$$

$$\theta^{t}_{B} + \varepsilon^{t} \geq \left[a^{t,L}_{Bj} - \left(a^{t,U}_{Bj} - a^{t,L}_{Bj} \right) \alpha \right] \theta^{t}_{j}, \quad \forall j, t,$$

$$\theta^{t}_{j} - \varepsilon^{t} \leq \left[a^{t,L}_{jW} + \left(a^{t,U}_{jW} - a^{t,L}_{jW} \right) \alpha \right] \theta^{t}_{W}, \quad \forall j, t,$$

$$\theta^{t}_{j} + \varepsilon^{t} \geq \left[a^{t,L}_{jW} - \left(a^{t,U}_{jW} - a^{t,L}_{jW} \right) \alpha \right] \theta^{t}_{W}, \quad \forall j, t,$$

$$\sum_{j=1}^{n} \theta^{t}_{j} = 1, \quad \forall t; \quad \theta^{t}_{j} \geq 0, \quad \forall j, t.$$

$$(V.2)$$

where α ($0 \le \alpha \le 1$) is the possibility level of preferences determined by the director. Model (V.2) is linear; thus, for each time period, solving the problem for any given possibility level α leads to a unique optimal weight vector $\left(\theta_1^{t,*}, \theta_2^{t,*}, \dots, \theta_n^{t,*}\right)$ and a corresponding objective value $\varepsilon^{t,*}$.

 $\mathcal{E}^{t,*}$ denotes the consistency level of the interval preference relation. If $\mathcal{E}^{t,*} = 0$ for all α , we have $\theta_B^t / \theta_j^t = \overline{a}_{Bj}^t$ and $\theta_j^t / \theta_W^t = \overline{a}_{jW}^t$, and correspondingly $\overline{a}_{Bj}^t \times \overline{a}_{jW}^t = \overline{a}_{BW}^t$ for all j and t. Thus, the interval preference relations \overline{A}_B^t and \overline{A}_W^t are entirely consistent.

Step 2 (Dynamic Weights of SoESs' Experts Using the Interval BWM Model): For each time period, the director distinguishes the best and worst experts in each SoES. Afterward, s/he provides the interval preferences (i.e., competence degrees) of all experts of the SoES according to the best and worst experts. $\overline{e}_{j,Bk}^t = \left[e_{j,Bk}^{t,L}, e_{j,Bk}^{t,U}\right]$ is the interval preference of Expert^{*j*}_{*B*} (the best expert in SoES *j*) over Expert^{*j*}_{*k*} (*k* = 1,2,..., *h*_{*j*}) for time period *t*. $\overline{e}_{j,kW}^t = \left[e_{j,kW}^{t,L}, e_{j,kW}^{t,U}\right]$ is the interval preference of Expert^{*j*}_{*k*} (*k* = 1,2,..., *h*_{*j*}) over Expert^{*j*}_{*W*} (the worst expert in SoES *j*) for time period *t*.

For SoES \vec{j} , two interval preference vectors are obtained according to the best and worst experts, respectively, as $\vec{E}_{j,B}^{t} = \left(\vec{e}_{j,B1}^{t}, \vec{e}_{j,B2}^{t}, ..., \vec{e}_{j,Bh}^{t}\right)$ and $\vec{E}_{j,W}^{t} = \left(\vec{e}_{j,1W}^{t}, \vec{e}_{j,2W}^{t}, ..., \vec{e}_{j,hW}^{t}\right)$. For time period t, we consider the following interval BWM model to obtain optimal weights of the experts of n SoESs:

$$\min \Delta^{t} = \sum_{j=1}^{n} \theta_{j}^{t,*} \varepsilon_{j}^{t},$$
s.t.
$$\left| \lambda_{j,B}^{t} - \overline{e}_{j,Bk}^{t} \lambda_{j,k}^{t} \right| \leq \varepsilon_{j}^{t}, \quad \forall k,$$

$$\left| \lambda_{j,k}^{t} - \overline{e}_{j,KW}^{t} \lambda_{j,W}^{t} \right| \leq \varepsilon_{j}^{t}, \quad \forall k,$$

$$\sum_{k=1}^{h_{j}} \lambda_{j,k}^{t} = 1, \quad \forall j,t; \quad \lambda_{j,k}^{t} \geq 0, \quad \forall k, j, t,$$

$$(V.3)$$

where $\theta_j^{t,*}$ represents the optimal weight of each subject obtained for each time period by solving Model (V.2) in Step 1. Similar to Model (V.2), a crisp equivalent can be formulated for Model (V.3).

The crisp equivalent of Model (V.3) has a linear form; thus, for each time period, its solution leads to a unique optimal weight vector $\left(\lambda_{j,1}^{t,*}, \lambda_{j,2}^{t,*}, \dots, \lambda_{j,h}^{t,*}\right)$ for each expert and the consistency levels $\left(\mathcal{E}_{1}^{t,*}, \mathcal{E}_{1}^{t,*}, \dots, \mathcal{E}_{h}^{t,*}\right)$ for any given possibility level α .

 $\Delta^{t,*} = 0$ means the complete consistency of interval competence degrees of all experts of the SoESs for time period t, i.e., $\overline{E}_{j,B}^{t}$ and $\overline{E}_{j,W}^{t}$. However, the interval preference relation may not be so precise to reach the complete consistency in practice. In $\overline{E}_{j,B}^{t}$ and $\overline{E}_{j,W}^{t}$, small deviations from the complete consistency situation lead to small positive values of $\Delta^{t,*}$. Thus, the magnitude of $\Delta^{t,*}$ represents the consistency level of multi-person decisions.

Step 3 (Dynamic Weights of Alternatives Using the Interval Multi-Person BWM Model): For time period t, Expert_k^j selects the best and worst alternatives according to his/her competence in C_k^j (i.e., the k th criterion in the category of the *j* th subject). Afterward, the expert considers interval preferences of all alternatives regarding the best and worst alternatives.

 $\overline{q}_{j,Bi}^{k,t} = \left[q_{j,Bi}^{k,t,U}, q_{j,Bi}^{k,t,U}\right] \text{ is the interval preference of } A_B \text{ (the best alternative) over } A_i \text{ considered by Expert}_k^j \text{ for time} \text{ period } t. \ \overline{q}_{j,iW}^{k,t} = \left[q_{j,iW}^{k,t,U}, q_{j,iW}^{k,t,U}\right] \text{ is the interval preference of } A_i \text{ over } A_W \text{ (the worst alternative) considered by Expert}_k^j \text{ for time} \text{ time period } t. \text{ For SoES } j \text{ and time period } t, \text{ two interval preference vectors are obtained according to the best and worst alternatives, respectively, as } \overline{Q}_{j,B}^{k,t} = (\overline{q}_{j,Bi}^{k,t}, \overline{q}_{j,Bi}^{k,t}), \dots, \overline{q}_{j,Bm}^{k,t}) \text{ and } \overline{Q}_{j,W}^{k,t} = (\overline{q}_{j,Wi}^{k,t}, \overline{q}_{j,2W}^{k,t}, \dots, \overline{q}_{j,nW}^{k,t}).$

For each time period, the following interval multi-person BWM model (as a multi-objective programming problem) is derived as follows:

$$\min \mathbf{F}^{t} = \begin{bmatrix} f_{1}^{t}, \dots, f_{j}^{t}, \dots, f_{n}^{t} \end{bmatrix} \\ = \begin{bmatrix} \sum_{k=1}^{h_{1}} \lambda_{1,k}^{t,*} \varepsilon_{1,k}^{t}, \dots, \sum_{k=1}^{h_{j}} \lambda_{j,k}^{t,*} \varepsilon_{j,k}^{t}, \dots, \\ \sum_{k=1}^{h_{n}} \lambda_{n,k}^{t,*} \varepsilon_{n,k}^{t} \end{bmatrix},$$
s.t.:
$$\begin{bmatrix} W_{B}^{t} - \overline{q}_{j,Bl}^{k,t} W_{i}^{t} \end{bmatrix} \leq \varepsilon_{j,k}^{t}, \quad \forall i, k, \\ |W_{i}^{t} - \overline{q}_{j,iW}^{k,t} W_{W}^{t} \end{bmatrix} \leq \varepsilon_{j,k}^{t}, \quad \forall i, k, \\ W_{i}^{t} - \overline{q}_{j,iW}^{k,t} W_{W}^{t} \end{bmatrix} \leq \varepsilon_{j,k}^{t}, \quad \forall i, k, \\ \sum_{i=1}^{m} W_{i}^{t} = 1, \quad \forall t; \quad w_{i}^{t} \ge 0, \quad \forall i, t, \\ \end{bmatrix}$$
where $\lambda_{j,k}^{t,*}$ is the optimal weight of each expert of SoESs in time period t computed based on Model (V.3) in Step 2.

 $\mathbf{F}^{t} = \begin{bmatrix} f_{1}^{t}, \dots, f_{j}^{t}, \dots, f_{n}^{t} \end{bmatrix}$ is the objective function vector including the set of objective functions of each SoES in time period *t*, i.e., $f_{j}^{t} = \sum_{k=1}^{h_{j}} \lambda_{j,k}^{t,*} \varepsilon_{j,k}^{t}$.

Model (V.4) can be solved using multi-objective programming methods such as Global Criterion Method, which is supported on L*p*-metric. Based on this method, an optimal vector is obtained that minimizes a global criterion [158]. The transformed form of Model (V.4) based on the global criterion method is as follows:

$$\min \widehat{F}^{t} = \sum_{j=1}^{n} \theta_{j}^{t,*} \left(\frac{\left| f_{j}^{t,*} - f_{j}^{t} \right|}{f_{j}^{t,*}} \right)^{p},$$
s.t.:

$$\left| w_{B}^{t} - \overline{q}_{j,Bi}^{k,t} w_{i}^{t} \right| \leq \varepsilon_{j,k}^{t}, \quad \forall i, k,$$

$$\left| w_{I}^{t} - \overline{q}_{j,iW}^{k,t} w_{W}^{t} \right| \leq \varepsilon_{j,k}^{t}, \quad \forall i, k,$$

$$\left| w_{i}^{t} - \overline{q}_{j,iW}^{k,t} w_{W}^{t} \right| \leq \varepsilon_{j,k}^{t}, \quad \forall i, k,$$

$$\sum_{i=1}^{m} w_{i}^{t} = 1, \quad \forall t; \quad w_{i}^{t} \geq 0, \quad \forall i, t,$$
where $\theta_{i}^{t,*}$ is the extinct which of each subject in time period t attained through solving Model (W 2) in Step 1...f^{t,*} is

where $\theta_j^{t,*}$ is the optimal weight of each subject in time period t attained through solving Model (V.2) in Step 1. $f_j^{t,*}$ is the ideal objective function of each SoES in time period t. The set of $f_j^{t,*}$ for different values of j represents as $\mathbf{F}^{t,*} = \left[f_1^{t,*}, \dots, f_j^{t,*}, \dots, f_n^{t,*}\right].$

Based on the concept of the global criterion method [159], the aim of Model (V.5) is how close we can get to ideal objective function vector $\mathbf{F}^{t,*}$. The exponent of the objective in Model (V.5), i.e., p, is an integer considered from the set $\{1, 2, ..., \infty\}$.

The value of p, considered by the director, denotes the type of distance. For p = 1, the deviation terms with lower values are considered more significant. As the value of p increases from 2 to ∞ , the greater deviation terms gain more importance [159]. For $p = \infty$, only the greatest deviation term is important and the summation operator converts to maximization of expression $\theta_j^{t,*}(|f_j^{t,*} - f_j^t|/f_j^{t,*})$ for different values of j.

Similar to Model (V.2), a crisp equivalent can be formulated for Model (V.5). The crisp equivalent of Model (V.5) has a linear form; thus, its solution leads to a unique optimal weight vector $(w_1^{t,*}, w_2^{t,*}, \dots, w_m^{t,*})$ and the consistency levels $(\varepsilon_{j,1}^{t,*}, \varepsilon_{j,2}^{t,*}, \dots, \varepsilon_{j,h_j}^{t,*})$ for each time period and subject j and any given possibility level α .

 $\hat{F}^{t,*} = 0$ means the complete consistency of interval preference relations of all SoESs for time period t, i.e., $\bar{Q}^{k,t}_{j,B}$ and $\bar{Q}^{k,t}_{j,W}$. However, the interval preference relation may not be so precise to reach the complete consistency in practice. In $\bar{Q}^{k,t}_{j,B}$ and $\bar{Q}^{k,t}_{j,W}$, slight deviations from the complete consistency situation lead to small positive values of $\hat{F}^{t,*}$. Thus, the magnitude of $\hat{F}^{t,*}$ is a measure of the consistency level of group decisions in each time period.

As illustrated in Fig. V.2, a dynamic matrix of overall assessments of all experts can be constructed from the results obtained for each time period, where each column is the output of Model (V.5) for the given time period. We show the dynamic matrix in Eq. (V.6). The decision-making process is the same in each time period; however, the judgment of experts may differ based on the variation of their judgments by passing time and changes in the conditions of the problem based on the essential varying dynamic feature of realistic problems.

$$\mathbf{D} = \begin{bmatrix} P^{0} & \cdots & P^{T} & \cdots & P^{T} \\ w_{1}^{0,*} & \cdots & w_{1}^{t,*} & \cdots & w_{1}^{T,*} \\ \vdots & \vdots & \vdots & \vdots \\ w_{i}^{0,*} & \cdots & w_{i}^{t,*} & \cdots & w_{i}^{T,*} \\ \vdots & \vdots & \vdots & \vdots \\ w_{m}^{0,*} & \cdots & w_{m}^{t,*} & \cdots & w_{m}^{T,*} \end{bmatrix} A_{n}$$

$$(V.6)$$

2.2. MULTIMOORA-based model for ranking process

In this section, we develop a MULTIMOORA-based model to aggregate the judgments in multiple time periods to reach an integrative outcome of experts' assessment. First, we must compute the integrated significance coefficients of the dynamic decisions to use in the MULTIMOORA-based model.

The integrated significance coefficients of each time period are obtained based on the concept of consistency of the interval multi-person BWM model and Information Entropy Method.

Step 1 (Subjective, Objective, and Integrated Significance Coefficients of the Dynamic Decisions): The optimal values of the objective function of Model (V.5) are considered to calculate subjective significance coefficients of each time period, i.e., $\hat{F}^{t,*}$. As $\hat{F}^{t,*}$ represents the consistency index of the interval multi-person BWM model, it can reflect the importance of the ranking of alternatives for each time period.

The normalized values of $\hat{F}^{t,*}$ are considered as subjective significance coefficients of each time period:

$$s^{t,c} = \widehat{F}^{t,*} / \sum_{t=0}^{T} \widehat{F}^{t,*} \,. \tag{V.7}$$

Information Entropy, also called Shannon Entropy, is a powerful tool to compute objective significance coefficients in D-MpMcDM problems. In the theory of Information Entropy, contrasts of data sets are applied to help distinguish more significant sets [160].

To compute the objective significance coefficients, first, the dynamic matrix, introduced in Eq. (V.6), has to be normalized as:
$$v_i^t = w_i^{t,*} / \sum_{i=1}^m w_i^{t,*}$$
 (V.8)

Second, Information Entropy Measure is computed using the normalized dynamic matrix as follows:

$$E' = -k \sum_{i=1}^{m} \left[v_i^i \ln\left(v_i^i\right) \right]$$
(V.9)

where $k = 1/\ln(m)$. Objective significance coefficients are computed based on Information Entropy as:

$$s^{t,e} = (1 - E^t) / \sum_{t=0}^{T} (1 - E^t).$$
 (V.10)

Subjective and objective significance coefficients, i.e., $s^{t,c}$ and $s^{t,e}$, respectively, are consolidated to form integrated significance coefficients as:

$$s^{t,*} = \left(s^{t,c} \cdot s^{t,e}\right) / \sum_{t=0}^{T} \left(s^{t,c} \cdot s^{t,e}\right).$$
(V.11)

Step 2 (Subordinate Utilities and Rankings of Alternatives Based on MULTIMOORA Approach): In this step, we present the derivation of MULTIMOORA subordinate parts, i.e., Ratio System, Reference Point Approach, and Full Multiplicative Form.

The utility of Ratio System is calculated as follows:

$$y_i = \sum_{t=0}^{T} \left(s^{t,*} w_i^{t,*} \right), \tag{V.12}$$

where $s^{t,*}$ and $w_i^{t,*}$ are the integrated significance coefficients of each time period and dynamic weights of alternatives based on Eqs. (V.11) and (V.12), respectively. The utilities of y_i are ordered descendingly to reach the ranking of alternatives based on Ratio System.

The utility of Reference Point Approach is obtained as follows:

$$z_{i} = \max_{t} \left| \left(s^{t,*} r^{t} \right) - \left(s^{t,*} w_{i}^{t,*} \right) \right|, \tag{V.13}$$

where the maximal objective reference point, i.e., r', equals to $\max_{i} W_{i}^{t,*}$. The utilities of z_{i} are ordered ascendingly to find the ranking of alternatives based on Reference Point Approach.

The utility of Full Multiplicative Form is generated as:

$$u_i = \prod_{t=0}^T \left(w_i^{t,*} \right)^{S^{t,*}}.$$
 (V.14)

The utilities of u_i are arranged descendingly to obtain the ranking of alternatives based on Full Multiplicative Form.

Step 3 (Final Utility and Ranking of Alternatives Based on MULTIMOORA-Borda Approach): We use the concept of Borda Rule to integrate the subordinate results of MULTIMOORA.

Borda Rule is a technique from the category of voting theory. From the viewpoint of computation and straightforwardness for decision-makers, Borda Rule outweighs the dominance theory traditionally used in MULTIMOORA method [18].

The formula for generating the final utility of MULTIMOORA-Borda approach is:

$$MB_{i} = y_{i}^{*} \frac{m - r(y_{i}) + 1}{m(m+1)/2} - z_{i}^{*} \frac{r(z_{i})}{m(m+1)/2} + u_{i}^{*} \frac{m - r(u_{i}) + 1}{m(m+1)/2},$$
(V.15)

where y_i^* , z_i^* , and u_i^* are the normalized subordinate utilities computed similar to Eq. (V.8). $r(y_i)$, $r(z_i)$, and $r(u_i)$ are the rankings obtained for Ratio System, Reference Point Approach, and Full Multiplicative Form, respectively. *m* is the number of alternatives.

The utilities of MB_i are arranged descendingly to obtain the final ranking of alternatives based on MULTIMOORA-Borda approach.

3. Application of DI_S-MpMcDM_SoES methodology in energy sector

This section discusses a practical problem regarding an energy investment project. The case study concerns investment in renewable energy sources (see Fig. V.3) tackled using the proposed D-MpMcDM-SoESs methodology.

As illustrated in Fig. V.4, a SoESs structure is considered because different competences are needed for the problem. The decision-making process is handled by four SoESs focused on financial, technological, environmental, and social subjects. As the director, a project manager supervises them. Each SoESs includes four experts. A project manager assesses the relative importance of the problem subjects and evaluates the relative competence degree of the experts of each SoES. Each expert of a SoES appraises the weights of the candidate renewable energy sources based on their specific competence in a single criterion.



Fig. V.3. The case study of the DI_S-MpMcDM_SoES methodology: Renewable energy investment project.



Fig. V.4. Supervised multi-person structure for the case study of the RF_S-MpMcDM_FRFP methodology.

The alternatives of renewable energy sources are six items: solar power, hydroelectric energy, wave energy, bioenergy, geothermal energy, and wind energy.

Solar power can be collected through photovoltaic cells, specially-designed power plants, or water heating systems operated based on sunshine. Hydroelectric energy is a traditional form of harnessing clean energy. The water flowing off the dam powers a turbine to produce electricity. Wave energy is captured utilizing different devices such as absorbers or inverted-pendulum systems. Bioenergy can be absorbed from any organic matter called biomass, which can be directly fired or processed into gas (biogas) or liquid (biofuel). Geothermal energy is obtained by pumping water down into the underground, specifically near the boundaries of Earth's tectonic plates. Wind energy is gathered by turbines established on windy lands [161].

First, we solve the decision-making problem for the present time (t = 0), i.e., the data of Oct-2021, based on Models (V.1)–(V.5). Linguistic preferences for all judgments made in Models (V.1)–(V.5) are the same values listed previously in Table III.4. The table also shows the equivalent interval values for the linguistic preferences.

Interval preference vectors considered for obtaining weights of subjects by the project manager are gathered in Table V.1. Besides, optimal weights of the subjects obtained from Model (V.2), i.e., $\theta_j^{0,*}$, for possibility level 0.5 are listed in the table. The value of the objective function of Model (V.2), i.e., $\varepsilon^{0,*}$, is equal to 0.085. Table V.2 gives the interval preference vectors considered by the project manager for comparing experts of SoESs.

Table V.1. Interval preferences of subject of SoESs assessed by the project manager (obtained for the present time (t=0)).

Interval preferences and optimal weights	Finance	Technology	Environment	Society
\overline{A}_B^0 (Best-to-Others vector) Best subject: Environment	[3, 5]	[7, 9]	[1, 1]	[3, 5]
\bar{A}_{W}^{0} (Others-to-Worst vector) <i>Worst subject</i> : Technology	[3, 5]	[1, 1]	[7, 9]	[3, 5]
$\theta_j^{0,*}$ (Optimal weights)	0.170	0.064	0.596	0.170

Table V.2. Competence degrees of the experts of SoESs assessed by project manager (obtained for the present time (t = 0)).

SoES	Best expert	Worst expert	Interval preferences transformed from linguistic terms				
				Expert 1	Expert 2	Expert 3	Expert 4
Financial segment Exp	Error e et 1	Error and 2	$\overline{E}^0_{l,B}$	[1, 1]	[1, 3]	[7, 9]	[3, 5]
	Expert 1	Expert 5	$\bar{E}^0_{\mathrm{I},W}$	[7, 9]	[5, 7]	[1, 1]	[1, 3]
Technological segment	Expert 2	Expert 1	$\overline{E}^0_{2,B}$	[7, 9]	[1, 1]	[1, 3]	[3, 5]
			$\overline{E}^0_{2,W}$	[1, 1]	[7, 9]	[5, 7]	[3, 5]
Environmental segment	Expert 3 Ex	Expert 4 $\overline{E}^0_{3,B}$ $\overline{E}^0_{3,W}$	$\overline{E}^0_{3,B}$	[3, 5]	[3, 5]	[1, 1]	[7, 9]
			$\overline{E}^0_{3,W}$	[3, 5]	[1, 3]	[7, 9]	[1, 1]
Social segment	Expert 4	Expert 2	$\overline{E}^0_{4,B}$	[1, 3]	[7, 9]	[1, 3]	[1, 1]
			$\overline{E}^0_{4,W}$	[5, 7]	[1, 1]	[5, 7]	[7, 9]

The six alternatives of renewable energy sources are assessed according to the related criterion based on the comments of each expert of the four SoESs. Table V.3 lists the relative interval preference degrees of the renewable energy sources provided by financial, technological, environmental, and social segments.

SoES	Expert of	Criterion	Best	Worst	Interval preferences transformed from linguistic terms						
	SoES	alternative alternative	alternative		Solar power	Hydroelectric energy	Wave energy	Bioenergy	Geothermal energy	Wind energy	
Expert 1 Investment cost		Geothermal	$\overline{\mathcal{Q}}_{1,B}^{1,0}$	[1, 1]	[1, 3]	[3, 5]	[3, 5]	[7, 9]	[5,7]		
	Solar power	energy	$\overline{\mathcal{Q}}_{1,W}^{1,0}$	[7, 9]	[5,7]	[3, 5]	[1,3]	[1, 1]	[1, 3]		
men	F (2	TT 1	D.	Geothermal energy	$\overline{\mathcal{Q}}_{1,B}^{2,0}$	[5, 7]	[3, 5]	[1, 3]	[1, 1]	[7, 9]	[5, 7]
al seg	Expert 2	Hardware cost	Bioenergy		$\overline{\mathcal{Q}}_{1,W}^{2,0}$	[1, 3]	[3, 5]	[3, 5]	[7, 9]	[1, 1]	[1, 3]
inci	Even out 2	Maintenance	Solar power	Geothermal energy	$\overline{\mathcal{Q}}_{1,B}^{3,0}$	[1, 1]	[1, 3]	[1, 3]	[1, 3]	[7, 9]	[5, 7]
Fina	Expert 5	cost			$\overline{\mathcal{Q}}_{1,W}^{3,0}$	[7, 9]	[5, 7]	[3, 5]	[5, 7]	[1, 1]	[1, 3]
	Expert 4	Payback period	Solar power	Wave energy	$\overline{\mathcal{Q}}{}^{4,0}_{1,B}$	[1, 1]	[3, 5]	[7, 9]	[1, 3]	[3, 5]	[3, 5]
	Блрон 4				$\overline{\mathcal{Q}}{}^{4,0}_{1,W}$	[7, 9]	[1,3]	[1, 1]	[5, 7]	[3, 5]	[1, 3]
	Export 1	Efficiency	Wind onergy	Hydroelectric	$\overline{\mathcal{Q}}{}^{1,0}_{2,B}$	[5, 7]	[7,9]	[1, 3]	[5, 7]	[3, 5]	[1, 1]
lent	Expert I	Efficiency	wind energy	energy	$\bar{\mathcal{Q}}_{2,W}^{1,0}$	[1, 3]	[1,1]	[3, 5]	[1, 3]	[3, 5]	[7,9]
segn	Export 2	Canacity factor	XV. 1	W	$\overline{Q}{}^{2,0}_{2,B}$	[3, 5]	[5, 7]	[7, 9]	[1, 3]	[5, 7]	[1, 1]
ical	Expert 2	Capacity factor	wind energy	wave energy	$\overline{Q}_{2,W}^{2,0}$	[1, 3]	[1,3]	[1, 1]	[5, 7]	[1, 3]	[7, 9]
olog	Export 3	Technical	W/in d an analy	Hydroelectric	$\overline{\mathcal{Q}}{}^{3,0}_{2,B}$	[3, 5]	[7,9]	[5, 7]	[1, 3]	[1, 3]	[1, 1]
in Expert 3 maturity	wind energy	energy	$\overline{\mathcal{Q}}{}^{3,0}_{2,W}$	[1, 3]	[1,1]	[1, 3]	[5, 7]	[3, 5]	[7,9]		
Envert 4. Esseibility	Solar power	Hydroelectric	$\bar{\mathcal{Q}}{}^{4,0}_{2,B}$	[1, 1]	[7,9]	[1, 3]	[5, 7]	[1, 3]	[5,7]		
		1 casionity		energy	$\overline{\mathcal{Q}}_{2,W}^{4,0}$	[7, 9]	[1, 1]	[3, 5]	[1, 3]	[3, 5]	[1,3]
Expert 1 Pollution emission	Wave energy	Bioanaray	$\overline{\mathcal{Q}}{}^{1,0}_{3,B}$	[3, 5]	[3, 5]	[1, 1]	[7, 9]	[5, 7]	[1,3]		
		Dioenergy	$\overline{\mathcal{Q}}{}^{1,0}_{3,W}$	[1, 3]	[1,3]	[7, 9]	[1, 1]	[1, 3]	[3, 5]		
		Geothermal	$\overline{\mathcal{Q}}{}^{2,0}_{3,B}$	[5, 7]	[3, 5]	[1, 3]	[5, 7]	[7, 9]	[1, 1]		
ntal s	Expert 2 Land use wind energy	wind energy	energy	$\overline{\mathcal{Q}}_{3,W}^{2,0}$	[1, 3]	[3, 5]	[5, 7]	[1, 3]	[1, 1]	[7,9]	
nmei	Expert 3	Water use	Wind energy	Bioenergy	$\overline{\mathcal{Q}}_{3,B}^{3,0}$	[3, 5]	[5, 7]	[1, 3]	[7, 9]	[5, 7]	[1, 1]
wiro		D D	Disenergy	$\overline{\mathcal{Q}}_{3,W}^{3,0}$	[1, 3]	[1, 3]	[3, 5]	[1, 1]	[1, 3]	[7,9]	
ц Б Expert 4 Noise emission	Noise emission	Wind energy	Bigenergy	$\overline{\mathcal{Q}}{}^{4,0}_{3,B}$	[1, 3]	[5, 7]	[1, 3]	[7, 9]	[3, 5]	[1, 1]	
			wind energy	Dioenergy	$\overline{\mathcal{Q}}{}^{4,0}_{3,W}$	[5, 7]	[1, 3]	[5, 7]	[1, 1]	[3, 5]	[7,9]
	Expert 1	Job creation	Solar power	Bioenergy	$\overline{\mathcal{Q}}{}^{1,0}_{4,B}$	[1, 1]	[5,7]	[3, 5]	[7, 9]	[1, 3]	[3, 5]
	Expert I	500 creation	Solai powei	Bioenergy	$\bar{\mathcal{Q}}{}^{1,0}_{4,W}$	[7, 9]	[1,3]	[3, 5]	[1, 1]	[5, 7]	[1,3]
Expert 2 Social acceptance	Hydroelectric	Bioenergy	$\overline{\mathcal{Q}}{}^{2,0}_{4,B}$	[1, 3]	[1,1]	[5, 7]	[7, 9]	[3, 5]	[3, 5]		
	energy	Dioenergy	$\overline{\mathcal{Q}}{}^{2,0}_{4,W}$	[3, 5]	[7,9]	[1, 3]	[1, 1]	[3, 5]	[1, 3]		
ocial	Expert 3	Market maturity	Solar power	Bioenergy	$\overline{\mathcal{Q}}{}^{3,0}_{4,B}$	[1, 1]	[1, 3]	[5, 7]	[7, 9]	[5, 7]	[1,3]
Ň	Experts	market maturity			$\overline{\mathcal{Q}}{}^{3,0}_{4,W}$	[7, 9]	[3, 5]	[1, 3]	[1, 1]	[1, 3]	[5,7]
	Expert 4	Fatal accidents	Solor norman	Geothermal	$\overline{\mathcal{Q}}{}^{4,0}_{4,B}$	[1, 1]	[1,3]	[3, 5]	[3, 5]	[7, 9]	[1,3]
Expert 4 Fatal accidents		energy	$\overline{\mathcal{Q}}{}^{4,0}_{4,W}$	[7, 9]	[5,7]	[1, 3]	[3, 5]	[1, 1]	[5, 7]		

Table V.3. Interval preference degrees of the alternatives of renewable energy sources assessed by SoESs (obtained for the present time (t = 0)).

For each time, the computation procedure is similar to the present time. For briefness, we ignore to present the preferences tables and only report the final results, i.e., the optimal weights obtained based on Model (V.5). Utilizing the outcomes of Oct-2020, Apr-2021, and Oct-2021, besides the present time results, i.e., Apr-2022, we can construct a dynamic matrix as presented in Table V.4 (mathematically defined previously in Eq. (V.6)). The data of Table V.4 is obtained for the possibility level 0.5 of interval preferences and the exponent value 2 of the global criterion method.

Renewable energy sources	Results of the interval multi-person BWM model for the four time periods					
	Oct-2020	Apr-2021	Oct-2021	Apr-2022		
	(t = -3)	(t = -2)	(t = -1)	(t = 0)		
Solar power	0.148	0.116	0.151	0.153		
Hydroelectric energy	0.103	0.082	0.093	0.102		
Wave energy	0.205	0.219	0.199	0.191		
Bioenergy	0.079	0.063	0.067	0.076		
Geothermal energy	0.103	0.082	0.088	0.096		
Wind energy	0.362	0.438	0.402	0.382		

Table V.4. Dynamic matrix of the results based on the data of the four time periods.

Table V.5 put forward utilities and rankings of MULTIMOORA-Borda for the case study on renewable energy investment. For Reference Point Approach, the rankings are calculated by arranging the related utilities ascendingly, while for the two other subordinate methods, rankings are obtained based on descending order.

Final utilities are obtained based on Borda Rule The final rankings are calculated by arranging the related utilities descendingly. The best alternative based on subordinate and final results of MULTIMOORA-Borda is similar, *i.e.*, Alternative 6 (wind energy).

The ranking lists of Ratio System and Full Multiplicative Form precisely match. The ranks of Reference Point Approach are very close to those of the two other techniques, i.e., they only differ in one rank (related to geothermal energy).

Renewable energy sources	Ratio system	Reference point approach	Full multiplicative form	MULTIMOORA- Borda
Solar power	0.142 (3)	0.081 (3)	0.141 (3)	0.028 (3)
Hydroelectric energy	0.096 (4)	0.098 (4)	0.096 (4)	-0.015 (4)
Wave energy	0.204 (2)	0.059 (2)	0.204 (2)	0.085 (2)
Bioenergy	0.073 (6)	0.107 (6)	0.073 (6)	-0.062 (6)
Geothermal energy	0.094 (5)	0.098 (4)	0.094 (5)	-0.024 (5)
Wind energy	0.391 (1)	0(1)	0.389 (1)	0.223 (1)

Table V.5. Utilities and rankings of MULTIMOORA-Borda for the renewable energy investment project.

Chapter VI

Conclusions, publications, and future works

In this chapter, we present the conclusions of the overview and methodologies as well as the published research works based on the thesis and future works in Sections 1, 2, and I.2, respectively.

1. Conclusions

In this section, we present the concluding remarks regarding the three proposed methodologies (i.e., I_S-MpMcDM, RF_S-MpMcDM_FRFP, and DI_S-MpMcDM_SoESs) as well as the points realized by analyzing the previous studies based on the overview on MULTIMOORA.

The following points can be concluded from the MULTIMOORA overview:

- Among uncertain sets, triangular fuzzy number, as a simple fuzzy number, is mostly applied to develop extensions.
- some uncertain developments need more mathematical concepts for generating the models; however, there are several developments which do not require uncertain arithmetic because they simply use score functions which only need crisp arithmetic

The following points can be concluded from the I_S-MpMcDM methodology:

- The formulation of the proposed methodology is based on total interval computation. That is, the proposed theory of interval numbers lead to interval assessment values for subordinate parts of the interval MULTIMOORA-based model. Interval data is usually the outcome of laboratory experiments performed in many industries. Thus, in contrast to the most fuzzy/interval MpMcDM techniques existed in the literature, the model based on I_S-MpMcDM methodology does not need any degradation of the interval numbers. As a result, no data is missed in the modeling and the results are more reliable.
- The integrated weights of criteria used in this methodology combine the comments of experts, based the interval BWM-based model, and the disorder of interval data of decision matrix, based on interval entropy. Thus, the consolidation of the two concepts results in more robust weights of criteria.

The following points can be concluded from the RF_S-MpMcDM_FRFP methodology:

• The risky fuzzy BWM-based model without considering director's decision theoretically leads to more consistent criteria weights, but considering decisions of the expert panel is important in practice.

- Different values of possibility level of the risky fuzzy preferences nearly do not influence the consistency index of the risky fuzzy BWM-based model.
- This risky fuzzy decision-making methodology fits well to the biomaterial selection problem because the work stresses of biomedical engineers may cause irrational decisions.

The following points can be concluded from the DI_S-MpMcDM_SoESs methodology:

- The triple interval BWM-based models are efficient for obtaining dynamic weights in the DI_S-MpMcDM SoESs problem as the obtained values for consistency levels are high.
- The MULIMOORA-based model is effective for the problem as the values of the resultant subordinate and final rankings closely coincide.
- Considering various values for possibility levels of interval experts' preferences has a negligible impact on final ranking and does not change the best alternative.

2. Publications

The two publications based on the thesis are as follows:

MULTIMOORA overview was published in Information Fusion (2019). The details of the publication is:

 Arian Hafezalkotob, Ashkan Hafezalkotob, Huchang Liao, and Francisco Herrera, "An overview of MULTIMOORA for multi-criteria decision-making: Theory, developments, applications, and challenges," Information Fusion, vol. 51, pp. 145–177, 2019.

The I_S-MpMcDM methodology was published in *IEEE Transactions on Cybernetics* (2020). The details of the publication is:

 Arian Hafezalkotob, Ashkan Hafezalkotob, Huchang Liao, and Francisco Herrera. Interval MULTIMOORA method integrating interval Borda rule and interval best-worst-method-based weighting model: Case study on hybrid vehicle engine selection. IEEE Transactions on Cybernetics, vol. 50 (3), pp. 1157–1169, 2020.

3. Future works

In this section, we give several guidelines for future research in the regard of the theory and application of the three proposed methodologies as well as the challenges for development of the MULTIMOORA.

The following directions are suggested for future works in the area of the MULTIMOORA:

- Prospective researchers may focus on cooperative and non-cooperative multi-person MULTIMOORA models. The comparative analysis of the cooperative and non-cooperative multi-person decision-making models can also be interesting.
- Risk management and data mining approaches are less worked in the studies on MULTIMOORA. Measuring multifarious risks and specifying the acceptability degree of each risk and analysis of costs and advantages of considering risks are the challenges at the heart of real-world decision-making problems.

The following directions are suggested for future works following the I_S-MpMcDM methodology:

• We considered that director only assesses the relative importance of the experts. The methodology can be enhanced to a more general form when director is responsible for evaluating the relative importance of the criteria of the problem.

• The interval BWM-based model can be extended by calculating the lower and upper limits for the subjective optimal weights. That is, the subjective crisp optimal weights obtained in this study can be converted to interval numbers by finding extreme values.

The following directions are suggested for future works following the RF_S-MpMcDM_FRFP methodology:

- The risks of irrational decisions of director and the expert panel can be regarded with different attitudes. Such assumption can be reasonable as director probably has more experience and knowledge about the problem than expert panel.
- The RF_S-MpMcDM_FRFP methodology can be employed in many real-world problems in which risky fuzzy decisions are significant such as practical cases on healthcare management and socioeconomic policy-making.

The following directions are suggested for future works following the DI_S-MpMcDM_SoESs methodology:

- The proposed DI_S-MpMcDM_SoESs methodology may have defects in tackling problems with interdependent subjects. For such problems, an extension of the DI_S-MpMcDM_SoESs methodology considering multiple segments with different competence of experts oriented on each subject can be a feasible solution.
- The behavior assessment of SoESs regarding the risk aversion of the experts' judgments can be appealing future research for which the anticipated regret theory is beneficial.
- Application of realistic fuzzy and linguistic representations to compute uncertain preferences of experts can be another extension of the proposed methodology [162], [163].

References

- A. Burns, M. A. Rathwell, and R. C. Thomas, "A distributed decision-making system," *Decis. Support Syst.*, vol. 3, no. 2, pp. 121–131, 1987.
- [2] A. Hafezalkotob and A. Hafezalkotob, "A novel approach for combination of individual and group decisions based on fuzzy best-worst method," *Appl. Soft Comput.*, vol. 59, pp. 316–325, 2017.
- [3] A. Hafezalkotob and A. Hafezalkotob, "Risk-based material selection process supported on information theory: A case study on industrial gas turbine," *Appl. Soft Comput.*, vol. 52, pp. 1116–1129, 2017.
- [4] W. Dai, Q. Zhong, and C. Qi, "Multi-stage multi-attribute decision-making method based on the prospect theory and triangular fuzzy MULTIMOORA," *Soft Comput.*, 2018.
- [5] H.-C. Liu, X.-J. Fan, P. Li, and Y.-Z. Chen, "Evaluating the risk of failure modes with extended MULTIMOORA method under fuzzy environment," *Eng. Appl. Artif. Intell.*, vol. 34, pp. 168–177, 2014.
- [6] A. Awasthi and T. Baležentis, "A hybrid approach based on BOCR and fuzzy MULTIMOORA for logistics service provider selection," *Int. J. Logist. Syst. Manag.*, vol. 27, no. 3, pp. 261–282, 2017.
- [7] H. Bustince *et al.*, "A historical account of types of fuzzy sets and their relationships," *IEEE Trans. Fuzzy Syst.*, vol. 24, no. 1, pp. 179–194, 2016.
- [8] A. Ijadi Maghsoodi, A. Hafezalkotob, I. Azizi Ari, S. Ijadi Maghsoodi, and A. Hafezalkotob, "Selection of Waste Lubricant Oil Regenerative Technology Using Entropy-Weighted Risk-Based Fuzzy Axiomatic Design Approach," *Informatica*, vol. 29, no. 1, pp. 41–74, 2018.
- [9] O. Kulak, H. G. Goren, and A. A. Supciller, "A new multi criteria decision making approach for medical imaging systems considering risk factors," *Appl. Soft Comput.*, vol. 35, pp. 931–941, 2015.
- [10] H. G. Gören and O. Kulak, "A New Fuzzy Multi-criteria Decision Making Approach: Extended Hierarchical Fuzzy Axiomatic Design Approach with Risk Factors," *Lect. Notes Bus. Inf. Process.*, vol. 184, pp. 22–33, 2014.
- [11] A. Hafezalkotob, A. Hafezalkotob, and M. K. Sayadi, "Extension of MULTIMOORA method with interval numbers: An application in materials selection," *Appl. Math. Model.*, vol. 40, no. 2, pp. 1372–1386, 2016.
- [12] S. Rahimi, A. Hafezalkotob, S. M. Monavari, A. Hafezalkotob, and R. Rahimi, "Sustainable landfill site selection for municipal solid waste based on a hybrid decision-making approach: Fuzzy group BWM-MULTIMOORA-GIS," J. Clean. Prod., vol. 248, p. 119186, 2020.
- [13] Y. Wang, L. Tian, and Z. Wu, "Trust modeling based on probabilistic linguistic term sets and the MULTIMOORA method," *Expert Syst. Appl.*, vol. 165, p. 113817, 2021.
- [14] J. S. Bunderson, "Recognizing and utilizing expertise in work groups: A status characteristics perspective," *Adm. Sci. Q.*, vol. 48, no. 4, pp. 557–591, 2003.
- [15] X. Deng, X. Cheng, J. Gu, and Z. Xu, "An Innovative Indicator System and Group Decision Framework for Assessing Sustainable Development Enterprises," Gr. Decis. Negot., pp. 1–38, 2019.
- [16] A. Hafezalkotob, A. Hafezalkotob, H. Liao, and F. Herrera, "An overview of MULTIMOORA for multi-criteria decision-making: Theory, developments, applications, and challenges," *Inf. Fusion*, vol. 51, pp. 145–177, 2019.
- [17] X. Mi, M. Tang, H. Liao, W. Shen, and B. Lev, "The state-of-the-art survey on integrations and applications of the best worst method in decision making: Why, what, what for and what's next?," *Omega*, vol. 87, pp. 205–225, 2019.
- [18] A. Hafezalkotob, A. Hafezalkotob, H. Liao, and F. Herrera, "Interval MULTIMOORA Method Integrating Interval Borda Rule and Interval Best–Worst-Method-Based Weighting Model: Case Study on Hybrid Vehicle Engine Selection," *IEEE Trans. Cybern.*, vol. 50, no. 3, pp. 1157–1169, 2020.
- [19] M. Yazdi, A. Nedjati, E. Zarei, and R. Abbassi, "A reliable risk analysis approach using an extension of best-worst method based on democratic-autocratic decision-making style," J. Clean. Prod., p. 120418, 2020.
- [20] H. Omrani, M. Amini, and A. Alizadeh, "An integrated group best-worst method Data envelopment analysis approach for evaluating road safety: A case of Iran," *Meas. J. Int. Meas. Confed.*, vol. 152, p. 107330, 2020.
- [21] R. M. P. Trindade, B. R. C. Bedregal, A. D. D. Neto, and B. M. Acioly, "An interval metric," in *New Advanced Technologies*, InTech, 2010, pp. 323–341.
- [22] R. E. Moore, *Methods and Applications of Interval Analysis*. Philadelphia, PA: Society for Industrial and Applied Mathematics, 1979.
- [23] A. Hafezalkotob and A. Hafezalkotob, "Interval target-based VIKOR method supported on interval distance and

preference degree for machine selection," Eng. Appl. Artif. Intell., vol. 57, pp. 184–196, 2017.

- [24] P. Sevastianov, "Numerical methods for interval and fuzzy number comparison based on the probabilistic approach and Dempster-Shafer theory," *Inf. Sci. (Ny).*, vol. 177, no. 21, pp. 4645–4661, 2007.
- [25] Y. M. Wang, J. B. Yang, and D. L. Xu, "A preference aggregation method through the estimation of utility intervals," *Comput. Oper. Res.*, vol. 32, no. 8, pp. 2027–2049, 2005.
- [26] A. Hafezalkotob and A. Hafezalkotob, "Fuzzy entropy-weighted MULTIMOORA method for materials selection," J. Intell. Fuzzy Syst., vol. 31, no. 3, pp. 1211–1226, 2016.
- [27] L. Tran and L. Duckstein, "Comparison of fuzzy numbers using a fuzzy distance measure," *Fuzzy Sets Syst.*, vol. 130, no. 3, pp. 331–341, 2002.
- [28] C. Aguilar-Peña, A.-F. Roldán-López de Hierro, C. Roldán-López de Hierro, and J. Martínez-Moreno, "A family of fuzzy distance measures of fuzzy numbers," *Soft Comput.*, vol. 20, no. 1, pp. 237–250, 2016.
- [29] M. A. Jahantigh, S. Hajighasemi, and T. Allahviranloo, "A fuzzy distance measure for generalized fuzzy numbers," *Data Envel. Anal. Decis. Sci.*, vol. 2014, pp. 1–11, 2014.
- [30] S. Abbasbandy and T. Hajjari, "A new approach for ranking of trapezoidal fuzzy numbers," *Comput. Math. with Appl.*, vol. 57, no. 3, pp. 413–419, 2009.
- [31] D. Kahneman and A. Tversky, "Prospect theory: An analysis of decision under risk," *Econometrica*, vol. 47, no. 2, p. 263, 1979.
- [32] X. Sang, X. Liu, and L. Chen, "Parametric Weighting Function for WOWA Operator and Its Application in Decision Making," Int. J. Intell. Syst., vol. 29, no. 2, pp. 119–136, 2014.
- [33] H. Liao, G. Si, Z. Xu, and H. Fujita, "Hesitant Fuzzy Linguistic Preference Utility Set and Its Application in Selection of Fire Rescue Plans," *Int. J. Environ. Res. Public Health*, vol. 15, no. 4, p. 664, 2018.
- [34] K. K. Dompere, Fuzzy Rationality, vol. 235 (SFSC). 2009.
- [35] J. Qin, X. Liu, and W. Pedrycz, "An extended VIKOR method based on prospect theory for multiple attribute decision making under interval type-2 fuzzy environment," *Knowledge-Based Syst.*, vol. 86, pp. 116–130, 2015.
- [36] T. Baležentis and A. Baležentis, "A survey on development and applications of the multi-criteria decision making method MULTIMOORA," J. Multi-Criteria Decis. Anal., vol. 21, no. 3–4, pp. 209–222, 2014.
- [37] P. Wang, Z. Zhu, and Y. Wang, "A novel hybrid MCDM model combining the SAW, TOPSIS and GRA methods based on experimental design," *Inf. Sci. (Ny).*, vol. 345, pp. 27–45, 2016.
- [38] G. Stojić, Ž. Stević, J. Antuchevičienė, D. Pamučar, and M. Vasiljević, "A novel rough WASPAS approach for supplier selection in a company manufacturing PVC carpentry products," *Information*, vol. 9, no. 5, p. 121, 2018.
- [39] Z. Pei, J. Liu, F. Hao, and B. Zhou, "FLM-TOPSIS: The fuzzy linguistic multiset TOPSIS method and its application in linguistic decision making," *Inf. Fusion*, vol. 45, pp. 266–281, 2019.
- [40] T.-Y. Chen, "Remoteness index-based Pythagorean fuzzy VIKOR methods with a generalized distance measure for multiple criteria decision analysis," *Inf. Fusion*, vol. 41, pp. 129–150, 2018.
- [41] T.-Y. Chen, "An interval type-2 fuzzy PROMETHEE method using a likelihood-based outranking comparison approach," *Inf. Fusion*, vol. 25, pp. 105–120, 2015.
- [42] H. C. Liao, L. Y. Yang, and Z. S. Xu, "Two new approaches based on ELECTRE II to solve the multiple criteria decision making problems with hesitant fuzzy linguistic term sets," *Appl. Soft Comput.*, vol. 63, pp. 223–234, 2018.
- [43] X. Wu and H. Liao, "An approach to quality function deployment based on probabilistic linguistic term sets and ORESTE method for multi-expert multi-criteria decision making," *Inf. Fusion*, vol. 43, pp. 13–26, 2018.
- [44] X. Wu and H. Liao, "A consensus-based probabilistic linguistic gained and lost dominance score method," *Eur. J. Oper. Res.*, vol. 272, no. 3, pp. 1017–1027, 2019.
- [45] W. K. M. Brauers and E. K. Zavadskas, "The MOORA method and its application to privatization in a transition economy," *Control Cybern.*, vol. 35, no. 2, pp. 445–469, 2006.
- [46] W. K. M. Brauers and E. K. Zavadskas, "Project management by multimoora as an instrument for transition economies," *Technol. Econ. Dev. Econ.*, vol. 16, no. 1, pp. 5–24, 2010.
- [47] A. Hafezalkotob and A. Hafezalkotob, "Interval MULTIMOORA method with target values of attributes based on interval distance and preference degree: biomaterials selection," J. Ind. Eng. Int., vol. 13, no. 2, pp. 181–198, 2017.
- [48] W. Gardziejczyk and P. Zabicki, "Normalization and variant assessment methods in selection of road alignment variants case study," J. Civ. Eng. Manag., vol. 23, no. 4, pp. 510–523, 2017.
- [49] H. Liao, X. Wu, and F. Herrera, "DNBMA: A Double Normalization-Based Multi-Aggregation Method," in Information Processing and Management of Uncertainty in Knowledge-Based Systems. Applications. IPMU 2018, vol. 855, Y. R. Medina J., Ojeda-Aciego M., Verdegay J., Perfilieva I., Bouchon-Meunier B., Ed. Springer, Cham, 2018, pp. 63–73.
- [50] W. K. M. Brauers, E. K. Zavadskas, F. Peldschus, and Z. Turskis, "Multi-objective decision-making for road design," *Transport*, vol. 23, no. 3, pp. 183–193, 2008.
- [51] W. K. M. Brauers and E. K. Zavadskas, "Robustness of MULTIMOORA: a method for multi-objective optimization," *Informatica*, vol. 23, no. 1, pp. 1–25, 2012.

- [52] H. Liao, R. Qin, C. Gao, X. Wu, A. Hafezalkotob, and F. Herrera, "Score-HeDLiSF: A score function of hesitant fuzzy linguistic term set based on hesitant degrees and linguistic scale functions: An application to unbalanced hesitant fuzzy linguistic MULTIMOORA," *Inf. Fusion*, vol. 48, pp. 39–54, 2019.
- [53] A. Hafezalkotob and A. Hafezalkotob, "Extended MULTIMOORA method based on Shannon entropy weight for materials selection," *J. Ind. Eng. Int.*, vol. 12, no. 1, pp. 1–13, 2016.
- [54] A. Hafezalkotob and A. Hafezalkotob, "Comprehensive MULTIMOORA method with target-based attributes and integrated significant coefficients for materials selection in biomedical applications," *Mater. Des.*, vol. 87, pp. 949– 959, 2015.
- [55] M. Eghbali-Zarch, R. Tavakkoli-Moghaddam, F. Esfahanian, M. M. Sepehri, and A. Azaron, "Pharmacological therapy selection of type 2 diabetes based on the SWARA and modified MULTIMOORA methods under a fuzzy environment," *Artif. Intell. Med.*, vol. 87, pp. 20–33, 2018.
- [56] C. Karaca and A. Ulutaş, "Supplier performance evaluation by using SWARA and MULTIMOORA," in *Economics, Management, & Econometrics*, A. M. Çilingirtürk, A. Albrychiewicz-Słocińska, and B. B. Bali, Eds. London: IJOPEC Publication Limited, 2017, pp. 137–151.
- [57] S. Altuntas, T. Dereli, and M. K. Yilmaz, "Evaluation of excavator technologies: Application of data fusion based MULTIMOORA methods," J. Civ. Eng. Manag., vol. 21, no. 8, pp. 977–997, 2015.
- [58] X. Wu, H. Liao, Z. Xu, A. Hafezalkotob, and F. Herrera, "Probabilistic Linguistic MULTIMOORA: A Multicriteria Decision Making Method Based on the Probabilistic Linguistic Expectation Function and the Improved Borda Rule," *IEEE Trans. Fuzzy Syst.*, vol. 26, no. 6, pp. 3688–3702, 2018.
- [59] W. Pedrycz, P. Ekel, and R. Parreiras, *Fuzzy Multi-Criteria Decision Making*, vol. 16. Boston, MA: Springer US, 2008.
- [60] M. Kracka and E. K. Zavadskas, "Panel building refurbishment elements effective selection by applying multiplecriteria methods," *Int. J. Strateg. Prop. Manag.*, vol. 17, no. 2, pp. 210–219, 2013.
- [61] L. A. Zadeh, "Fuzzy sets," Inf. Control, vol. 8, no. 3, pp. 338–353, 1965.
- [62] Z. Tian, J. Wang, J. Wang, and H. Zhang, "A multi-phase QFD-based hybrid fuzzy MCDM approach for performance evaluation: A case of smart bike-sharing programs in Changsha," J. Clean. Prod., vol. 171, pp. 1068– 1083, 2018.
- [63] H.-C. Liu, J.-X. You, C. Lu, and Y.-Z. Chen, "Evaluating health-care waste treatment technologies using a hybrid multi-criteria decision making model," *Renew. Sustain. Energy Rev.*, vol. 41, pp. 932–942, 2015.
- [64] W. K. M. Brauers, A. Baležentis, and T. Baležentis, "Multimoora for the EU member states updated with fuzzy number theory," *Technol. Econ. Dev. Econ.*, vol. 17, no. 2, pp. 259–290, 2011.
- [65] B. Ceballos, M. T. Lamata, and D. A. Pelta, "Fuzzy multicriteria decision-making methods: A comparative analysis," *Int. J. Intell. Syst.*, vol. 32, no. 7, pp. 722–738, 2017.
- [66] F. Sezer, Ö. Bali, and P. Gürol, "Hazardous materials warehouse selection as a multiple criteria decision making problem," *J. Econ. Bibliogr.*, vol. 3, no. 1S, pp. 63–73, 2016.
- [67] A. Baležentis, T. Baležentis, and W. K. M. Brauers, "MULTIMOORA-FG: A multi-objective decision making method for linguistic reasoning with an application to personnel selection," *Informatica*, vol. 23, no. 2, pp. 173–190, 2012.
- [68] T. Baležentis, "A farming efficiency estimation model based on fuzzy MULTIMOORA," *Manag. Theory Stud. Rural Bus. Infrastruct. Dev.*, vol. 29, no. 5, pp. 43–52, 2011.
- [69] R. Fattahi and M. Khalilzadeh, "Risk evaluation using a novel hybrid method based on FMEA, extended MULTIMOORA, and AHP methods under fuzzy environment," *Saf. Sci.*, vol. 102, no. October 2017, pp. 290–300, 2018.
- [70] D. Deliktas and O. Ustun, "Student selection and assignment methodology based on fuzzy MULTIMOORA and multichoice goal programming," *Int. Trans. Oper. Res.*, vol. 24, no. 5, pp. 1173–1195, 2017.
- [71] D. K. Sen, S. Datta, S. K. Patel, and S. S. Mahapatra, "Green supplier selection in fuzzy context: a decision-making scenario on application of fuzzy-MULTIMOORA," Int. J. Serv. Oper. Manag., vol. 28, no. 1, p. 98, 2017.
- [72] A. Baležentis, T. Baležentis, and W. K. M. Brauers, "Personnel selection based on computing with words and fuzzy MULTIMOORA," *Expert Syst. Appl.*, vol. 39, no. 9, pp. 7961–7967, 2012.
- [73] L. Balezentiene, D. Streimikiene, and T. Baležentis, "Fuzzy decision support methodology for sustainable energy crop selection," *Renew. Sustain. Energy Rev.*, vol. 17, pp. 83–93, 2013.
- [74] F. Çebi and I. Otay, "A two-stage fuzzy approach for supplier evaluation and order allocation problem with quantity discounts and lead time," *Inf. Sci. (Ny).*, vol. 339, pp. 143–157, 2016.
- [75] A. Liu *et al.*, "A novel two-stage integrated model for supplier selection of green fresh product," *Sustainability*, vol. 10, no. 7, p. 2371, 2018.
- [76] D. Stanujkic, E. K. Zavadskas, W. K. M. Brauers, and D. Karabasevic, "An extension of the MULTIMOORA method for solving complex decision-making problems based on the use of interval-valued triangular fuzzy numbers," *Transform. Bus. Econ.*, vol. 14, no. 2B, pp. 355–377, 2015.

- [77] Y. Dorfeshan, S. M. Mousavi, V. Mohagheghi, and B. Vahdani, "Selecting project-critical path by a new interval type-2 fuzzy decision methodology based on MULTIMOORA, MOOSRA and TPOP methods," *Comput. Ind. Eng.*, vol. 120, pp. 160–178, 2018.
- [78] S. Mishra, A. K. Sahu, S. Datta, and S. S. Mahapatra, "Application of fuzzy integrated MULTIMOORA method towards supplier/partner selection in agile supply chain," *Int. J. Oper. Res.*, vol. 22, no. 4, p. 466, 2015.
- [79] A. K. Sahu, S. Datta, and S. S. Mahapatra, "Use of IVFNs and MULTIMOORA method for supply chain performance measurement, benchmarking and decision-making: an empirical study," *Int. J. Bus. Excell.*, vol. 7, no. 2, pp. 237–280, 2014.
- [80] A. K. Sahu, N. K. Sahu, and A. K. Sahu, "Application of modified MULTI-MOORA for CNC machine tool evaluation in IVGTFNS environment: an empirical study," *Int. J. Comput. Aided Eng. Technol.*, vol. 8, no. 3, pp. 234–259, 2016.
- [81] T. Baležentis and S. Zeng, "Group multi-criteria decision making based upon interval-valued fuzzy numbers: An extension of the MULTIMOORA method," *Expert Syst. Appl.*, vol. 40, no. 2, pp. 543–550, 2013.
- [82] T. Baležentis and A. Baležentis, "Group decision making procedure based on trapezoidal intuitionistic fuzzy numbers: MULTIMOORA methodology," *Econ. Comput. Econ. Cybern. Stud. Res.*, vol. 50, no. 1, pp. 103–122, 2016.
- [83] T. Baležentis, S. Zeng, and A. Baležentis, "MULTIMOORA-IFN: A MCDM method based on intuitionistic fuzzy number for performance management," *Econ. Comput. Econ. Cybern. Stud. Res.*, vol. 48, no. 4, 2014.
- [84] E. K. Zavadskas, J. Antuchevičienė, S. H. Razavi Hajiagha, and S. S. Hashemi, "The interval-valued intuitionistic fuzzy MULTIMOORA method for group decision making in engineering," *Math. Probl. Eng.*, vol. 2015, no. Article ID 560690, p. 13 pages, 2015.
- [85] H. Zhao, J.-X. You, and H.-C. Liu, "Failure mode and effect analysis using MULTIMOORA method with continuous weighted entropy under interval-valued intuitionistic fuzzy environment," *Soft Comput.*, vol. 21, no. 18, pp. 5355–5367, 2017.
- [86] Z.-H. Li, "An extension of the MULTIMOORA method for multiple criteria group decision making based upon hesitant fuzzy sets," J. Appl. Math., vol. 2014, no. Article ID 527836, p. 16 pages, 2014.
- [87] S. Zeng, A. Baležentis, and W. Su, "The multi-criteria hesitant fuzzy group decision making with MULTIMOORA method," *Econ. Comput. Econ. Cybern. Stud. Res.*, vol. 47, no. 3, pp. 171–184, 2013.
- [88] L. Dong, X. Gu, X. Wu, and H. Liao, "An improved MULTIMOORA method with combined weights and its application in assessing the innovative ability of universities," *Expert Syst.*
- [89] M. Lazauskas, V. Kutut, and E. K. Zavadskas, "Multicriteria assessment of unfinished construction projects," J. Croat. Assoc. Civ. Eng., vol. 67, no. 3, pp. 319–328, 2015.
- [90] W. K. M. Brauers, S. Kildienė, E. K. Zavadskas, and A. Kaklauskas, "The construction sector in twenty European countries during the recession 2008–2009 – country ranking by MULTIMOORA," *Int. J. Strateg. Prop. Manag.*, vol. 17, no. 1, pp. 58–78, 2013.
- [91] W. K. M. Brauers, M. Kracka, and E. K. Zavadskas, "Lithuanian case study of masonry buildings from the Soviet period," J. Civ. Eng. Manag., vol. 18, no. 3, pp. 444–456, 2012.
- [92] E. K. Zavadskas, R. Bausys, B. Juodagalviene, and I. Garnyte-Sapranaviciene, "Model for residential house element and material selection by neutrosophic MULTIMOORA method," *Eng. Appl. Artif. Intell.*, vol. 64, no. May, pp. 315–324, 2017.
- [93] M. Kracka, W. K. M. Brauers, and E. K. Zavadskas, "Ranking heating losses in a building by applying the MULTIMOORA," *Eng. Econ.*, vol. 21, no. 4, pp. 352–359, 2010.
- [94] E. K. Zavadskas, J. Antuchevičienė, J. Šaparauskas, and Z. Turskis, "MCDM methods WASPAS and MULTIMOORA: Verification of robustness of methods when assessing alternative solutions," *Econ. Comput. Econ. Cybern. Stud. Res.*, vol. 47, no. 2, 2013.
- [95] Ž. Stević, D. Pamučar, M. Vasiljević, G. Stojić, and S. Korica, "Novel integrated multi-criteria model for supplier selection: Case study construction company," *Symmetry (Basel).*, vol. 9, no. 11, p. 279, 2017.
- [96] E. K. Zavadskas, Z. Turskis, R. Volvačiovas, and S. Kildienė, "Multi-criteria assessment model of technologies," *Stud. Informatics Control*, vol. 22, no. 4, pp. 249–258, 2013.
- [97] X. Ding and J. Zhong, "Power Battery Recycling Mode Selection Using an Extended MULTIMOORA Method," Sci. Program., vol. 2018, no. Article ID 7675094, p. 14 pages, 2018.
- [98] H.-C. Liu, M. Yang, M. Zhou, and G. Tian, "An integrated multi-criteria decision making approach to location planning of electric vehicle charging stations," *IEEE Trans. Intell. Transp. Syst.*, 2018.
- [99] H. Liu, H. Zhao, X.-Y. You, and W.-Y. Zhou, "Robot Evaluation and Selection Using the Hesitant Fuzzy Linguistic MULTIMOORA Method," J. Test. Eval., vol. 47, no. 2, pp. 1405–1426, 2019.
- [100] S.-M. Wu, X.-Y. You, H.-C. Liu, and L.-E. Wang, "Improving quality function deployment analysis with the cloud MULTIMOORA method," *Int. Trans. Oper. Res.*, vol. 00, pp. 1–22, 2017.
- [101] N. Kundakci, "Combined multi-criteria decision making approach based on MACBETH and MULTI-MOORA

methods," Alphanumeric J., vol. 4, no. 1, pp. 17–26, 2016.

- [102] A. Hafezalkotob, A. Hami-Dindar, N. Rabie, and A. Hafezalkotob, "A decision support system for agricultural machines and equipment selection: A case study on olive harvester machines," *Comput. Electron. Agric.*, vol. 148, no. March, pp. 207–216, 2018.
- [103] D. Stanujkic, E. K. Zavadskas, F. Smarandache, W. K. M. Brauers, and D. Karabasevic, "A neutrosophic extension of the MULTIMOORA method," *Informatica*, vol. 28, no. 1, pp. 181–192, 2017.
- [104] W. Liang, G. Zhao, and C. Hong, "Selecting the optimal mining method with extended multi-objective optimization by ratio analysis plus the full multiplicative form (MULTIMOORA) approach," *Neural Comput. Appl.*, 2018.
- [105] D. Karabasevic, D. Stanujkic, S. Urosevic, and M. Maksimovic, "Selection of candidates in the mining industry based on the application of the SWARA and the MULTIMOORA methods," *Acta Montan. Slovaca*, vol. 20, no. 2, pp. 116–124, 2015.
- [106] S. Aydın, "Augmented reality goggles selection by using neutrosophic MULTIMOORA method," J. Enterp. Inf. Manag., 2018.
- [107] A. Baležentis and T. Baležentis, "Assessing the efficiency of Lithuanian transport sector by applying the methods of multimoora and data envelopment analysis," *Transport*, vol. 26, no. 3, pp. 263–270, 2011.
- [108] Z. Tian, J. Wang, J. Wang, and H. Zhang, "An improved MULTIMOORA approach for multi-criteria decisionmaking based on interdependent inputs of simplified neutrosophic linguistic information," *Neural Comput. Appl.*, vol. 28, no. S1, pp. 585–597, 2017.
- [109] H. Souzangarzadeh, M. J. Rezvani, and A. Jahan, "Selection of optimum design for conical segmented aluminum tubes as energy absorbers: Application of MULTIMOORA method," *Appl. Math. Model.*, vol. 51, pp. 546–560, 2017.
- [110] J. Qin and X. Liu, "2-tuple linguistic Muirhead mean operators for multiple attribute group decision making and its application to supplier selection," *Kybernetes*, vol. 45, no. 1, pp. 2–29, 2016.
- [111] W. K. M. Brauers and E. K. Zavadskas, "A multi-objective decision support system for project selection with an application for the Tunisian textile industry," *E a M Ekon. a Manag.*, vol. 15, no. 1, pp. 28–43, 2012.
- [112] H.-C. C. Liu, J.-X. X. You, C. Lu, and M.-M. M. Shan, "Application of interval 2-tuple linguistic MULTIMOORA method for health-care waste treatment technology evaluation and selection," *Waste Manag.*, vol. 34, no. 11, pp. 2355–2364, 2014.
- [113] P. Chiranjeevi and S. Sengupta, "Interval-Valued Model Level Fuzzy Aggregation-Based Background Subtraction," IEEE Trans. Cybern., 2017.
- [114] V. Balali, B. Zahraie, and A. Roozbahani, "Integration of ELECTRE III and PROMETHEE II Decision-Making Methods with an Interval Approach: Application in Selection of Appropriate Structural Systems," J. Comput. Civ. Eng., vol. 28, no. 2, pp. 297–314, 2014.
- [115] F. Liu and W.-G. Zhang, "TOPSIS-based consensus model for group decision-making with incomplete interval fuzzy preference relations," *IEEE Trans. Cybern.*, vol. 44, no. 8, pp. 1283–1294, 2014.
- [116] M. K. Sayadi, M. Heydari, and K. Shahanaghi, "Extension of VIKOR method for decision making problem with interval numbers," *Appl. Math. Model.*, vol. 33, no. 5, pp. 2257–2262, 2009.
- [117] X. Gou, H. Liao, Z. Xu, and F. Herrera, "Double hierarchy hesitant fuzzy linguistic term set and MULTIMOORA method: A case of study to evaluate the implementation status of haze controlling measures," *Inf. Fusion*, vol. 38, pp. 22–34, 2017.
- [118] A. Ijadi Maghsoodi, G. Abouhamzeh, M. Khalilzadeh, and E. K. Zavadskas, "Ranking and selecting the best performance appraisal method using the MULTIMOORA approach integrated Shannon's entropy," *Front. Bus. Res. China*, vol. 12, no. 1, p. 2, 2018.
- [119] J. Rezaei, "Best-worst multi-criteria decision-making method," Omega, vol. 53, pp. 49–57, 2015.
- [120] Q. Mou, Z. Xu, and H. Liao, "A graph based group decision making approach with intuitionistic fuzzy preference relations," *Comput. Ind. Eng.*, vol. 110, pp. 138–150, 2017.
- [121] Q. Mou, Z. Xu, and H. Liao, "An intuitionistic fuzzy multiplicative best-worst method for multi-criteria group decision making," Inf. Sci. (Ny)., vol. 374, pp. 224–239, 2016.
- [122] H. Aboutorab, M. Saberi, M. R. Asadabadi, O. Hussain, and E. Chang, "ZBWM: The Z-number extension of Best Worst Method and its application for supplier development," *Expert Syst. Appl.*, vol. 107, pp. 115–125, 2018.
- [123] Y. J. S. Benshan, "Joining methods in group appraising," Syst. Eng. Pract., vol. 1, p. 010, 1992.
- [124] G. R. Jahanshahloo, F. Hosseinzadeh Lotfi, and A. R. Davoodi, "Extension of TOPSIS for decision-making problems with interval data: Interval efficiency," *Math. Comput. Model.*, vol. 49, no. 5–6, pp. 1137–1142, 2009.
- [125] F. H. Lotfi and R. Fallahnejad, "Imprecise Shannon's entropy and multi attribute decision making," *Entropy*, vol. 12, no. 1, pp. 53–62, 2010.
- [126] D. Stanujkic, N. Magdalinovic, R. Jovanovic, and S. Stojanovic, "An objective multi-criteria approach to optimization using MOORA method and interval grey numbers.," *Technol. Econ. Dev. Econ.*, vol. 18, no. 2, pp. 331–363, 2012.

- [127] M. Ehsani, Y. Gao, and A. Emadi, *Modern Electric, Hybrid Electric, and Fuel Cell Vehicles: Fundamentals, Theory, and Design*, Second Edi. CRC Press, 2017.
- [128] C. Lv, Y. Liu, X. Hu, H. Guo, D. Cao, and F. Y. Wang, "Simultaneous observation of hybrid states for cyberphysical systems: A case study of electric vehicle powertrain," *IEEE Trans. Cybern.*, 2018.
- [129] M. Nikowitz, Advanced Hybrid and Electric Vehicles: System Optimization and Vehicle Integration. Switzerland: Springer International Publishing, 2016.
- [130] C. C. Chan, "The state of the art of electric and hybrid vehicles," Proc. IEEE, vol. 90, no. 2, pp. 247-275, 2002.
- [131] C. Mi and M. A. Masrur, Hybrid Electric Vehicles: Principles and Applications with Practical Perspectives. New Jersey: John Wiley & Sons, 2017.
- [132] P. Liu and P. Zhang, "A normal wiggly hesitant fuzzy MABAC method based on CCSD and prospect theory for multiple attribute decision making," *Int. J. Intell. Syst.*, vol. 36, no. 1, pp. 447–477, 2020.
- [133] Y. Liu, Y. Wang, M. Xu, and G. Xu, "Emergency Alternative Evaluation Using Extended Trapezoidal Intuitionistic Fuzzy Thermodynamic Approach with Prospect Theory," Int. J. Fuzzy Syst., vol. 21, no. 6, pp. 1801–1817, 2019.
- [134] J. Huang, X. Y. You, H. C. Liu, and S. L. Si, "New approach for quality function deployment based on proportional hesitant fuzzy linguistic term sets and prospect theory," *Int. J. Prod. Res.*, no. 3, pp. 1–17, 2018.
- [135] K. K. Dompere, Fuzzy Rationality: A Critique and Methodological Unity of Classical, Bounded and Other Rationalities, vol. 235 (SFSC). Springer Berlin Heidelberg, 2009.
- [136] K. K. Dompere, Social Goal-Objective Formation, Democracy and National Interest, vol. 4. 2014.
- [137] A. Grünbaum, "The role of psychological explanations of the rejection or acceptance of scientific theories," Sci. Pseudo-Science Soc., pp. 29–53, 2006.
- [138] A. Alaszewski and K. Coxon, "Uncertainty in everyday life: Risk, worry and trust," *Health, Risk and Society*, vol. 11, no. 3. Routledge, pp. 201–207, 2009.
- [139] E. O. Martz, V. K. Goel, M. H. Pope, and J. B. Park, "Materials and design of spinal implants—A review," J. Biomed. Mater. Res., vol. 38, no. 3, pp. 267–288, 1997.
- [140] M. Bahraminasab et al., "On the influence of shape and material used for the femoral component pegs in knee prostheses for reducing the problem of aseptic loosening," *Mater. Des.*, vol. 55, pp. 416–428, 2014.
- [141] A. Jahan and K. L. Edwards, "VIKOR method for material selection problems with interval numbers and targetbased criteria," *Mater. Des.*, vol. 47, pp. 759–765, 2013.
- [142] A. Jahan and K. L. Edwards, "Weighting of dependent and target-based criteria for optimal decision-making in materials selection process: Biomedical applications," *Mater. Des.*, vol. 49, pp. 1000–1008, 2013.
- [143] M. M. Farag, Materials and Process Selection for Engineering Design. Boca Raton, FL: CRC Press, 2014.
- [144] D. F. Williams, "There is no such thing as a biocompatible material," *Biomaterials*, vol. 35, no. 38, pp. 10009–10014, 2014.
- [145] A. Jahan, M. Bahraminasab, and K. L. Edwards, "A target-based normalization technique for materials selection," *Mater. Des.*, vol. 35, pp. 647–654, 2012.
- [146] H.-M. Lin et al., "Biomechanical analysis and design of a dynamic spinal fixator using topology optimization: a finite element analysis," Med. Biol. Eng. Comput., vol. 52, no. 5, pp. 499–508, 2014.
- [147] J. J. Jassbi, R. A. Ribeiro, and L. R. Varela, "Dynamic MCDM with future knowledge for supplier selection," J. Decis. Syst., vol. 23, no. 3, pp. 232–248, 2014.
- [148] I. J. Pérez, F. J. Cabrerizo, S. Alonso, Y. C. Dong, F. Chiclana, and E. Herrera-Viedma, "On dynamic consensus processes in group decision making problems," *Inf. Sci. (Ny).*, vol. 459, pp. 20–35, 2018.
- [149] M. Xue, C. Fu, and S. L. Yang, "Group consensus reaching based on a combination of expert weight and expert reliability," *Appl. Math. Comput.*, vol. 369, p. 124902, 2020.
- [150] M. Keshavarz-Ghorabaee, M. Amiri, E. K. Zavadskas, Z. Turskis, and J. Antucheviciene, "A dynamic fuzzy approach based on the EDAS method for multi-criteria subcontractor evaluation," *Inf.*, vol. 9, no. 3, p. 68, 2018.
- [151] Z. X. Su, M. Y. Chen, G. P. Xia, and L. Wang, "An interactive method for dynamic intuitionistic fuzzy multiattribute group decision making," *Expert Syst. Appl.*, vol. 38, no. 12, pp. 15286–15295, 2011.
- [152] W. Spröte, "ECOSOC Economic And Social Council," in A Concise Encyclopedia of the United Nations, Brill Nijhoff, 2010, pp. 147–152.
- [153] M. Al-Bidaiwi and K. V Sivakumar, "Deal With Start-Up and Commissioning Threats and Challeges at an Early Stage of the Project for a Sucessful Handover and Project Completion," in SPE International Production and Operations Conference & Exhibition, 14-16 May, Doha, Qatar, 2012.
- [154] B. Liu, S. Jiao, Y. Shen, Y. Chen, G. Wu, and S. Chen, "A dynamic hybrid trust network-based dual-path feedback consensus model for multi-attribute group decision-making in intuitionistic fuzzy environment," *Inf. Fusion*, vol. 80, pp. 266–281, 2022.
- [155] R. Tao, Z. Liu, R. Cai, and K. H. Cheong, "A dynamic group MCDM model with intuitionistic fuzzy set: Perspective of alternative queuing method," *Inf. Sci. (Ny).*, vol. 555, pp. 85–103, 2021.
- [156] M. Tang, H. Liao, E. Herrera-Viedma, C. L. P. Chen, and W. Pedrycz, "A Dynamic Adaptive Subgroup-to-Subgroup

Compatibility-Based Conflict Detection and Resolution Model for Multicriteria Large-Scale Group Decision Making," *IEEE Trans. Cybern.*, vol. 51, no. 10, pp. 4784–4795, 2021.

- [157] J. Shanteau, "Competence in experts: The role of task characteristics," Organ. Behav. Hum. Decis. Process., vol. 53, no. 2, pp. 252–266, 1992.
- [158] C.-L. Hwang and A. S. M. Masud, Multiple Objective Decision Making: Methods and Applications, A State-of-the-Art Survey. Springer-Verlag, 1979.
- [159] C. A. C. Coello, G. B. Lamont, D. A. Van Veldhuizen, D. E. Goldberg, and J. R. Koza, Evolutionary Algorithms for Solving Multi-Objective Problems. 2007.
- [160] J. A. Morente-Molinera, J. Mezei, C. Carlsson, and E. Herrera-Viedma, "Improving Supervised Learning Classification Methods Using Multigranular Linguistic Modeling and Fuzzy Entropy," *IEEE Trans. Fuzzy Syst.*, vol. 25, no. 5, pp. 1078–1089, 2017.
- [161] S. S. Goswami, S. K. Mohanty, and D. K. Behera, "Selection of a green renewable energy source in India with the help of MEREC integrated PIV MCDM tool," *Mater. Today Proc.*, vol. 52, pp. 1153–1160, 2022.
- [162] X. Gou, Z. Xu, H. Liao, and F. Herrera, "Consensus Model Handling Minority Opinions and Noncooperative Behaviors in Large-Scale Group Decision-Making Under Double Hierarchy Linguistic Preference Relations," *IEEE Trans. Cybern.*, vol. 51, no. 1, pp. 283–296, 2021.
- [163] E. Herrera-viedma et al., "Revisiting Fuzzy and Linguistic Decision Making : Scenarios and Challenges for Making Wiser Decisions in a Better Way," pp. 1–18, 2020.